# **ECONSTOR** Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Dobbelaere, Sabien; Kiyota, Kozo

# Working Paper Labor market imperfections, markups and productivity in multinationals and exporters

Tinbergen Institute Discussion Paper, No. 17-113/V

**Provided in Cooperation with:** Tinbergen Institute, Amsterdam and Rotterdam

*Suggested Citation:* Dobbelaere, Sabien; Kiyota, Kozo (2017) : Labor market imperfections, markups and productivity in multinationals and exporters, Tinbergen Institute Discussion Paper, No. 17-113/ V, Tinbergen Institute, Amsterdam and Rotterdam

This Version is available at: https://hdl.handle.net/10419/177681

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU





# Labor market imperfections, markups and productivity in multinationals and exporters

Sabien (S.) Dobbelaere<sup>1</sup> Kozo Kiyota<sup>2</sup>

- 1: Vrije Universiteit Amsterdam, The Netherlands; Tinbergen Institute, The Netherlands; Institute of Labor Economics (IZA), Germany; Tinbergen Institute, The Netherlands
- 2: Keio University, Japan; Research Institute of Economy, Trade and Industry (RIETI), Japan

Tinbergen Institute is the graduate school and research institute in economics of Erasmus University Rotterdam, the University of Amsterdam and VU University Amsterdam.

Contact: discussionpapers@tinbergen.nl

More TI discussion papers can be downloaded at the Tinbergen Site

Tinbergen Institute has two locations:

Tinbergen Institute Amsterdam Gustav Mahlerplein 117 1082 MS Amsterdam The Netherlands Tel.: +31(0)20 598 4580

Tinbergen Institute Rotterdam Burg. Oudlaan 50 3062 PA Rotterdam The Netherlands Tel.: +31(0)10 408 8900

# Labor market imperfections, markups and productivity in multinationals and exporters<sup>\*</sup>

Sabien DOBBELAERE<sup>†</sup> Kozo KIYOTA<sup>‡</sup>

December, 2017

#### Abstract

This paper examines the links between the internationalization mode of firms and market imperfections in product and labor markets. We develop a framework for modelling heterogeneity across firms in terms of (i) product market power (price-cost markups), (ii) labor market imperfections (workers' bargaining power during worker-firm negotiations or firm's degree of wage-setting power) and (iii) revenue productivity. We apply this framework to analyze whether the pricing behavior of firms in product and labor markets differs across firms that engage in different forms of internationalization. Engagement in international activities is found to matter for determining not only the type of imperfections in product and labor markets but also the degree of imperfections. Clear differences in behavior between firms that serve the foreign market either through exporting or through FDI are observed. Being an exporter introduces allocative inefficiencies in product as well as labor markets as we find export status to be positively correlated with both product market power (markups) and market power consolidated on the labor supply side (workers' bargaining power). But exporting firms where search frictions are inducing wages to vary with revenue are less able to exploit wage-setting power. Firms with foreign subsidiaries, on the other hand, seem to reduce price distortions in product and labor markets. In addition, we observe heterogeneous returns to being an exporter/MNE within an industry and also discern cross-industry differences.

JEL classification : C23, D24, F14, F16, J50, L13.

*Keywords* : Rent sharing, monopsony, price-cost mark-ups, productivity, exporting, multinational firms, panel data.

<sup>\*</sup>We benefited from useful discussions with and comments by Antonio Accetturo, Boris Hirsch, Fukunari Kimura, Jacques Mairesse, Lionel Nesta, Toshihiro Okubo, Stefano Schiavo and other participants at various conferences and seminars. This study was conducted as part of the project "Microeconometric analysis of firm and industry growth" undertaken at the Research Institute of Economy, Trade and Industry (RIETI). The empirical analysis uses micro data from the questionnaire in the Basic Survey of Japanese Business Structure and Activities, which is conducted by the Ministry of Economy, Trade and Industry (METI). Kiyota gratefully acknowledges financial support from the Japan Society for the Promotion of Science (JSPS), Grant-in-Aid (JP16H02018) and the MEXT-Supported Program for the Strategic Research Foundation at Private Universities.

<sup>&</sup>lt;sup>†</sup>Vrije Universiteit Amsterdam, Tinbergen Institute and Institute of Labor Economics (IZA). Corresponding author: *sabien.dobbelaere@vu.nl.* 

<sup>&</sup>lt;sup>‡</sup>Keio Economic Observatory, Keio University and RIETI.

#### 1 Introduction

During the past decades, the relationship between globalization and wages has been at the center of debate in industrialized countries. A growing theoretical literature emphasizes trade-induced variation in firm-specific wages as one of the main drivers of increased wage inequality. This literature takes the seminal contribution of Melitz (2003) as a point of departure but abstains from the assumption that all workers are employed for a common wage and considers rent sharing to be the key mechanism through which trade-induced variation in rents is transmitted to variation in wages.

Building on Hopenhayn (1992) and Krugman (1980), the Melitz-model is characterized by firm heterogeneity in productivity and fixed export costs, and monopolistic competition and generates trade-induced shifts in the productivity distribution through selection of efficient firms into exporting and inefficient firms into exit. This model does not provide a model of income distribution as workers are symmetrically affected by trade liberalization because the labor market is frictionless and all workers are identical.

A recent theoretical literature on heterogeneous firms and trade has paid attention to the interaction between firms' selection and labor markets. There exist various heterogeneous-firms approaches to trade and wage inequality which all draw on imperfect factor markets but differ in terms of the rent-sharing mechanism between workers and firms that generate inter-firm wage dispersion even with ex ante identical workers. A first approach considers fair wages (Egger and Kreickemeier, 2009; Amiti and Davis, 2011) or efficiency wages (Davis and Harrigan, 2011) as a source of labor market imperfections, with productivity-specific wages resulting from a fair-wage effort mechanism in the former and different monitoring technologies in the latter. A second approach focuses on search and matching frictions such that ex-post bargaining over the surplus of production can potentially induce wages to vary with revenue across firms (Davidson *et al.*, 2008; Helpman *et al.*, 2010; Felbermayr *et al.*, 2011; Fajgelbaum, 2013; Coçsar *et al.*, 2016). A third approach considers firm-level unionization as a source of labor market imperfections, with decentralized collective bargaining producing interfirm wage disparities (Montagna and Nocco, 2013).

On the empirical side, microeconometric studies testing some of the predictions of these (or extended) models can be classified in several groups. A first set of papers has established empirical support for Melitz's selection effect, i.e. the positive relationship between a firm's export or multinational enterprise (MNE) status and its productivity level.<sup>1</sup> A second set of papers has provided evidence of the theoretical conjecture that reductions in trade costs lead to a positive correlation between exports/FDI and wages.<sup>2</sup> A third, small set of papers has tested the prediction that ex-

<sup>&</sup>lt;sup>1</sup>See e.g. the surveys of Helpman (2006) and Bernard *et al.* (2007, 2012) for evidence on the positive exporter productivity premium and Temouri *et al.* (2008) for evidence on the positive MNE productivity premium.

<sup>&</sup>lt;sup>2</sup>See e.g. Harrison *et al.* (2011) for references on evidence of exporter wage premia and Malchow-Møller *et al.* (2013) and Konings *et al.* (2016) for references on evidence of MNE wage premia.

porters charge higher price-cost markups.<sup>3</sup> Imposing a particular rent-sharing mechanism on the data, a fourth set of papers has investigated the impact of openness on labor market imperfections.<sup>4</sup>

In spite of the aforementioned growing importance of labor market imperfections in theoretical trade models, no empirical study has so far investigated how product *and* labor market imperfections vary across firms that differ in terms of mode of internationalization. This paper serves the purpose of examining heterogeneity in product and labor market imperfections across exporters, non-exporters, MNEs and non-MNEs. We contribute to the empirical international trade literature and the econometric literature on identifying firm-specific market imperfections along various dimensions.

First, we develop an econometric framework that allows for three-dimensional firm heterogeneity: product market power (price-cost markups), labor market imperfections (workers' bargaining power during worker-firm negotiations or firm's degree of wage-setting power) and revenue total factor productivity (TFP). Rather than imposing a particular imperfect labor market model on the data, we let the data determine the type of competition prevailing in product and labor markets. We accomplish this by building on the econometric reduced-form productivity model with imperfect product and labor markets which has been developed in Dobbelaere and Mairesse (2013). As such, we derive product and labor market imperfection parameters and regression-based TFP measures from estimating firm production functions. The theoretical structural productivity model behind the econometric reduced-form productivity model nests two polar models of wage determination in imperfect labor markets in the seminal productivity model of Hall (1988) which allows to estimate price-cost markups: the strongly efficient bargaining model (one of the two canonical collective bargaining models; McDonald and Solow, 1981) allocates market power to employees through costs of firing, hiring and training while the static partial equilibrium monopsony model (Manning, 2003) allocates market power to employers through allowing workers to have heterogeneous preferences over workplace environments of different potential employers, which generates upward-sloping labor supply curves to individual firms.

The second contribution is to apply this framework to analyze the type and the degree of product and labor market imperfections in firms that differ in terms of internationalization, while accounting for

<sup>&</sup>lt;sup>3</sup>This prediction can either be generated by heterogeneity on the supply side (productivity) as in e.g. Melitz and Ottaviano (2008) or by heterogeneity on the demand side (quality differences). See e.g. Loecker and Warzynski (2012), Kato (2014) and references in De Loecker *et al.* (2016) for empirical support based on the former and e.g. Hallak and Sivadasan (2009) and Kugler and Verhoogen (2012) for empirical support based on the latter. Forlani *et al.* (2016) explicitly account for heterogeneity on both the supply and the demand side.

<sup>&</sup>lt;sup>4</sup>Relying on a collective bargaining framework, several studies have shown evidence of a relationship between international trade and workers' bargaining power using either firm panel data (e.g. Brock and Dobbelaere, 2006; Dumont *et al.*, 2006; Abraham *et al.*, 2009, Boulhol *et al.*, 2011 and Ahsan and Mitra, 2014) or matched employer-employee data (e.g. Felbermayr *et al.*, 2014). Relying on a search-and-matching framework, Davidson *et al.* (2014) examine the impact of openness on the degree of matching between workers and firms using matched employer-employee data, while Lu *et al.* (2017) investigate how liberalization of inward foreign direct investment affects firm's monopsony power using firm panel data.

differences in revenue productivity. Using an unbalanced panel of 7,458 manufacturing firms covering the period 1994-2012 in Japan, we consider exporters, non-exporters, MNEs and non-MNEs. As such, our analysis aims at improving our understanding of the wage determination process in firms that engage differently in international activities through discerning whether either market power on the supply side of labor or market power on the demand side of labor is predominantly responsible for introducing allocative inefficiencies through distorting factor prices. We might expect the precise form of firm-worker rent sharing to be different across, e.g., exporters and MNEs. Exporting firms, which are relatively more productive, might charge higher markups and realize higher rents, and might be willing to share part of these rents with their workers according to a surplus-sharing rule, thereby increasing market power on the labor supply side. Intra-firm competition in multinationals, triggered by the threat to transfer production, R&D or some other tasks to a competing subsidiary, is likely to increase intra-firm labor replacement. As such, MNEs could have considerable monopsony power in the labor market, implying that market power could be consolidated on the labor demand side. To examine the link between the internationalization mode of firms and the type of competition prevailing in product and labor markets, we estimate (two-equation) probit models. To obtain a detailed picture of the relationship between export/FDI behavior and the degree of product and labor market imperfections, we apply quantile regression techniques which allow to investigate how the impact of export/MNE status varies along the conditional distribution of either product or labor market imperfections.

Our main findings are summarized as follows. First, we find that engagement in international activities not only matters for determining the type of imperfections in product and labor markets but also for determining the degree of imperfections. Second, we observe clear differences in behavior between firms that serve the foreign market either through exporting or through FDI. Third, we show that differences in product and labor market imperfections between exporters (MNEs) and non-exporters (non-MNEs) vary across firms. In addition, we reveal heterogeneous returns to being an exporter/MNE within an industry and also discern cross-industry differences.

Focusing on differential impacts on the type of imperfections, we find that being an exporter increases the likelihood of being characterized by imperfect competition in the product market. Exporting firms are more likely to share rents based on the bargaining power of workers, but less likely to share rents based on the elasticity of the labor supply curve facing an individual employer. As such, workers' bargaining power rather than search frictions seems to be important in generating wage dispersion across exporting firms.

Focusing on differential impacts on the degree of imperfections, we find that being an exporter introduces allocative inefficiencies in product as well as labor markets as export status appears to be positively correlated with both product market power (markups) and market power consolidated on the labor supply side (workers' bargaining power). Interestingly, export status is positively correlated with the wage elasticity of a firm's labor supply curve. This indicates that exporting firms where search frictions are inducing wages to vary with revenue, are less able to exploit wage-setting power. Firms with foreign subsidiaries seem to reduce price distortions in product and labor markets. A negative correlation is observed between MNE status and markups. A potential explanation is that MNEs also perform service activities where price competition might be fierce, thereby lowering markups. Likewise, MNE status and workers' bargaining power is negatively correlated. This could be explained by the fact that offshoring could increase substitution between domestic and foreign workers. This might in turn flatten the labor demand curve and shift bargaining power over rent distribution from labor towards capital in MNEs that enjoy extra-normal profits.

The plan of the article is as follows. Section 2 presents the main ingredients of the theoretical structural productivity model with imperfect product and labor markets. Section 3 discusses our econometric model and the estimation procedure. Section 4 presents the Japanese firm panel data. Section 5 examines how the type of competition prevailing in product and labor markets varies across firms that differ in terms of engagement in international activities. Section 6 investigates potential links between internationalization and firms' degree of product and labor market imperfections. Section 7 concludes.

## 2 Theoretical structural productivity model with imperfect product and labor markets

A firm i at time t produces output using the following production technology:

$$Q_{it} = Q_{it}(N_{it}, M_{it}, K_{it}) \tag{1}$$

with  $(N_{it}, M_{it})$  a vector of static inputs in production free of adjustment costs (labor and intermediate inputs) and  $K_{it}$  capital treated as a dynamic input in production (predetermined in the short run).

We assume that (i)  $Q_{it}(\cdot)$  is continuous and twice differentiable with respect to its arguments, (ii) a firm takes the input price of materials as given, (iii) firms produce in a homogeneous good industry and compete in quantities (play Cournot)<sup>5</sup> and (iv) producers active in the market are maximizing short-run profits.

Let us turn to the oligopolistic firm's short-run profit maximization problem. Firm *i*'s short-run profits,  $\Pi_{it}$ , are given by:

$$\Pi_{it} = R_{it} - W_{it}N_{it} - J_{it}M_{it} \tag{2}$$

 $<sup>{}^{5}</sup>$  This assumption is consistent with only observing a domestic industry-wide output price index and not firm-specific output prices (see infra).

with  $R_{it} = P_t Q_{it}$  an increasing and concave revenue function,  $P_t$  the price of the homogenous good at time t, and  $W_{it}$  and  $J_{it}$  the firm's input prices for N and M, respectively, at time t.

Firm *i* must choose the optimal quantity of output and the optimal demand for intermediate inputs and labor. The optimal output choice  $Q_{it}$  satisfies the following first-order condition:

$$\frac{P_t}{(C_Q)_{it}} = \left(1 + \frac{s_{it}}{\eta_t}\right)^{-1} = \mu_{it} \tag{3}$$

with  $(C_Q)_{it} = \frac{\partial C_{it}}{\partial Q_{it}}$  the marginal cost of production,  $s_{it} = \frac{Q_{it}}{Q_t}$  the market share of firm i,  $\eta_t = \frac{\partial Q_t}{\partial P_t} \frac{P_t}{Q_t}$  the own-price elasticity of industry demand and  $\mu_{it}$  firm i's price-cost markup. Under Cournot competition, differences in price-cost mark-ups across firms are generated by differences in productivity and market structure  $(s_{it}, \eta_t)$ .

The first-order condition for the optimal choice of intermediate inputs is given by setting the marginal revenue product of intermediate inputs equal to the price of intermediate inputs:

$$(Q_M)_{it} = \frac{J_{it}}{P_t} \left(1 + \frac{s_{it}}{\eta_t}\right)^{-1} \tag{4}$$

Inserting Eq. (3) in Eq. (4) and multiplying both sides by  $\frac{M_{it}}{Q_{it}}$  yields:

$$(\varepsilon_M^Q)_{it} = \mu_{it} s_{Mit} \tag{5}$$

From Eq. (5), it follows that profit maximization implies that optimal demand for intermediate inputs is satisfied when a firm equalizes the output elasticity with respect to intermediate inputs, denoted by  $(\varepsilon_M^Q)_{it} = \frac{\partial Q_{it}}{\partial M_{it}} \frac{M_{it}}{Q_{it}}$ , to the price-cost mark-up  $\mu_{it}$  multiplied by the share of intermediate input expenditure in total sales, denoted by  $s_{Mit} = \frac{J_{it}M_{it}}{P_tQ_{it}}$ .

Firm *i*'s optimal demand for labor depends on the characteristics of its labor market. We distinguish three labor market settings (LMS): perfect competition or right-to-manage bargaining (PR), strongly efficient bargaining (EB) and static partial equilibrium monopsony (MO).

Under PR, labor is unilaterally determined by firm i from short-run profit maximization, which implies the following first-order condition:

$$(\varepsilon_N^Q)_{it} = \mu_{it} s_{Nit} \tag{6}$$

with  $(\varepsilon_N^Q)_{it} = \frac{\partial Q_{it}}{\partial N_{it}} \frac{N_{it}}{Q_{it}}$  the output elasticity with respect to labor and  $s_{Nit} = \frac{W_{it}N_{it}}{P_tQ_{it}}$  the share of labor expenditure in total sales. In the perfectly competitive labor market model, a firm takes the exogenously-determined market wage as given. A profit-maximizing firm always chooses employment such that the marginal revenue product of labor equals the wage (Eq. (6)). In the right-to-manage bargaining model, the firm and its workers bargain over any surplus in order to determine

the wage (Nickell and Andrews, 1983). The firm continues to choose the number of workers it wishes to employ once wages have been determined by the bargaining process, which implies the same static first-order condition for labor as in the perfectly competitive labor market model.

Under EB, the risk-neutral firm and its risk-neutral workers negotiate simultaneously over wages and employment in order to maximize the joint surplus of their economic activity (McDonald and Solow, 1981). An efficient wage-employment pair is obtained by maximizing a generalized Nash product<sup>6</sup> with respect to the wage rate and labor. The following first-order condition with respect to wages must hold at an interior optimum:

$$W_{it} = \overline{W}_{it} + \gamma_{it} \left[ \frac{R_{it} - W_{it}N_{it} - J_{it}M_{it}}{N_{it}} \right]$$
(7)

where  $\gamma_{it} = \frac{\phi_{it}}{1-\phi_{it}}$  is the relative extent of rent sharing and  $\phi_{it} \in [0, 1]$  the part of economic rents going to the workers.

The first-order condition for labor is given by:

$$W_{it} = (R_N)_{it} + \phi_{it} \left[ \frac{R_{it} - (R_N)_{it} N_{it} - J_{it} M_{it}}{N_{it}} \right]$$
(8)

with  $(R_N)_{it} = \frac{\partial R_{it}}{\partial N_{it}}$  the marginal revenue product of labor.

An efficient wage-employment pair is given by solving simultaneously the first-order conditions with respect to the wage rate and labor. As such, the equilibrium condition is given by:

$$(R_N)_{it} = W_{it} \tag{9}$$

Eq. (9) traces out the locus of efficient wage-employment pairs, known as the contract curve. Given that  $\mu_{it} = \frac{P_t}{(R_Q)_{it}}$  in equilibrium, with  $(R_Q)_{it} = \frac{\partial R_{it}}{\partial Q_{it}}$  the marginal revenue, we obtain the following expression for the output elasticity with respect to labor by combining Eqs. (7) and (9):

$$(\varepsilon_N^Q)_{it} = \mu_{it} s_{Nit} - \mu_{it} \gamma_{it} (1 - s_{Nit} - s_{Mit})$$

$$\tag{10}$$

So far, we have assumed that there is a potentially infinite supply of employees wanting a job in the firm. A small wage cut by the employer will result in the immediate resignation of all existing workers. However, under MO, the labor supply facing an individual employer might be less than perfectly elastic because workers might have heterogeneous preferences over workplace environments of different potential employers (Manning, 2003). Such heterogeneity in e.g. firm location or job characteristics (corporate culture, starting times of work) makes workers to view employers as imperfect substitutes. This in turn gives employers non-negligible market power over their workers.

<sup>&</sup>lt;sup>6</sup> The generalized Nash product is written as:  $\Omega_{EB} = \left\{ N_{it}W_{it} + (\overline{N}_{it} - N_{it})\overline{W}_{it} - \overline{N}_{it}\overline{W}_{it} \right\}^{\phi} \left\{ R_{it} - W_{it}N_{it} - J_{it}M_{it} \right\}^{1-\phi_{it}}$ with  $\overline{N}_{it}$  the competitive employment level,  $\overline{W}_{it}$  the workers' alternative wage and  $\phi_{it} \in [0, 1]$  the part of economic rents going to the workers or the degree of workers' bargaining power during worker-firm negotiations.

Let us assume that the monopsonist firm is constrained to set a single wage for all his workers and faces labor supply  $N_{it}(W_{it})$ , which is an increasing function of the wage W. Both  $N_{it}(W_{it})$  and the inverse of this relationship  $W_{it}(N_{it})$  are referred to as the labor supply curve of this firm. The monopsonist firm's objective is to maximize its short-run profit function  $\Pi_{it} = R_{it} - W_{it}(N_{it}) N_{it} - J_{it}M_{it}$ , taking the labor supply curve as given. Maximizing this profit function with respect to labor gives the following first-order condition:<sup>7</sup>

$$(R_N)_{it} = (W_N)_{it} N_{it} + W_{it} (N_{it})$$
(11)

Rewriting Eq. (11) gives:

$$W_{it} = \beta_{it} (R_N)_{it} \tag{12}$$

with  $\beta_{it} = \frac{W_{it}}{(R_N)_{it}} = \frac{(\varepsilon_W^N)_{it}}{1+(\varepsilon_W^N)_{it}}$ .  $\beta_{it} \leq 1$  represents the wage markdown and  $(\varepsilon_W^N)_{it} = \frac{\partial N_{it}(W_{it})}{\partial W_{it}} \frac{W_{it}}{N_{it}} \in \Re_+$  the wage elasticity of the labor supply curve that firm *i* faces, measuring the degree of wagesetting power that firm *i* possesses. Perfect competition corresponds to the case where  $(\varepsilon_W^N)_{it} = \infty$ , hence  $(R_N)_{it} = W_{it}$ . Under monopsony,  $(\varepsilon_W^N)_{it}$  is finite and the labor supply curve that firm *i* faces is upward sloping, hence, the firms sets  $W_{it} < (R_N)_{it}$ . As such, the degree of firm *i*'s wage-setting power decreases in the wage elasticity of its labor supply curve.

Rewriting Eq. (12) and using that  $(R_N)_{it} = \frac{P_t(Q_N)_{it}}{\mu_{it}}$  with  $(Q_N)_{it}$  the marginal product of labor, gives the following expression for the elasticity of output with respect to labor:

$$(\varepsilon_N^Q)_{it} = \mu_{it} s_{Nit} \left( 1 + \frac{1}{(\varepsilon_W^N)_{it}} \right)$$
(13)

Using the first-order condition for intermediate inputs, we obtain an expression for firm *i*'s price-cost markup  $(\mu_{it})$  and using the first-order conditions for intermediate inputs and labor, we define firm *i*'s parameter of product and labor market imperfections  $(\psi_{it})$ , which we label firm *i*'s joint market imperfections parameter, as follows:

$$\mu_{it} = \frac{(\varepsilon_M^Q)_{it}}{s_{Mit}} \tag{14}$$

$$\psi_{it} = \frac{(\varepsilon_M^Q)_{it}}{s_{Mit}} - \frac{(\varepsilon_N^Q)_{it}}{s_{Nit}}$$
(15)

$$= 0 \quad \text{if LMS} = PR \tag{16}$$

$$= \mu_{it}\gamma_{it} \left[\frac{1 - s_{Nit} - s_{Mit}}{s_{Nit}}\right] > 0 \quad \text{if LMS=EB}$$
(17)

$$= -\mu_{it} \frac{1}{(\varepsilon_W^N)_{it}} < 0 \quad \text{if LMS=MO}$$
(18)

<sup>&</sup>lt;sup>7</sup>From Eq. (11), it follows that profit maximization implies that the optimal demand for labor is satisfied when a firm equalizes the marginal revenue product of labor to the marginal cost of labor. The latter is higher than the wage paid to the new worker  $W_{it}(N_{it})$  by the amount  $(W_N)_{it}N_{it}$  because the firm has to increase the wage paid to all workers it already employs whenever it hires an extra worker.

#### **3** Econometric model

In order to obtain consistent estimates of the output elasticities  $(\varepsilon_N^Q)_{it}$  and  $(\varepsilon_M^Q)_{it}$ , we only consider production functions with (i) a scalar Hicks-neutral productivity term which is observed by the firm but unobserved by the econometrician (denoted by  $\omega_{it}$ ) and (ii) common technology parameters, governing the transformation of inputs to units of output, across a set of producers (denoted by the vector  $\boldsymbol{\beta}$ ). These two assumptions imply the following expression for the production function:

$$Q_{it} = F(N_{it}, M_{it}, K_{it}; \boldsymbol{\beta}) \exp(\omega_{it})$$
(19)

In order to obtain consistent estimates of the production function coefficients ( $\beta$ ) for each of the 15 two-digit industries that we consider (see infra), we need to control for unobserved productivity shocks  $\omega_{it}$ , which are potentially correlated with the firm's input choices. We apply the estimation procedure proposed by Ackerberg *et al.* (2015) using the insight that optimal input choices hold information about unobserved productivity. We denote the logarithms of  $Q_{it}$ ,  $N_{it}$ ,  $M_{it}$  and  $K_{it}$  by  $q_{it}$ ,  $n_{it}$ ,  $m_{it}$  and  $k_{it}$ , respectively.

We impose the following timing assumptions. Capital  $k_{it}$  is assumed to be decided a period ahead (at t-1) because of planning and installation lags. Labor is "less variable" than material. More precisely,  $n_{it}$  is chosen by firm i at time t-b (0 < b < 1), after  $k_{it}$  being chosen at t-1 but prior to  $m_{it}$  being chosen at t. This assumption is consistent with firms needing time to train new workers, with firms facing significant hiring or firing costs for labor, or with labor contracts being long term as e.g. in unionized firms/industries.

We assume that unobservable productivity  $(\omega_{it})$  evolves according to an endogenous first-order Markov process. In particular, we allow a firm's decision to import to endogenously affect future productivity. This allows us to decompose  $\omega_{it}$  into its conditional expectation given the information known by the firm in t - 1 (denoted  $I_{it-1}$ ) and a random innovation to productivity (denoted  $\xi_{it}$ ):

$$\omega_{it} = E[\omega_{it}|I_{it-1}] + \xi_{it}$$

$$= E[\omega_{it}|\omega_{it-1}, IMP_{it-1}] + \xi_{it}$$

$$= g(\omega_{it-1}, IMP_{it-1}) + \xi_{it}$$
(20)

with  $IMP_{it-1}$  the import status of firm *i* at period t-1 and  $g(\cdot)$  a general function.  $\xi_{it}$  is assumed to be mean independent of the firm's information set at t-1.

Given these timing assumptions, firm *i*'s intermediate input demand at *t* depends directly on  $n_{it}$  chosen prior to  $m_{it}$ , i.e. the input demand function for  $m_{it}$  is conditional on  $n_{it}$ :<sup>8</sup>

$$m_{it} = m_t(n_{it}, k_{it}, IMP_{it}, \omega_{it}) \tag{21}$$

<sup>&</sup>lt;sup>8</sup>By allowing for observed shifters (here  $IMP_{it}$ ) that enter the optimal demand function for  $m_{it}$ , but are excluded from the production function, we solve the non-identification problem of the output elasticity with respect to materials and, hence, are in a position to apply the control function approach for the estimation of a gross output production function (see Ghandi

Eq. (21) shows that firm *i*'s intermediate input demand decision is a function of the state variables  $n_{it}, k_{it}, IMP_{it}$  and  $\omega_{it}$ . It is crucial that  $\omega_{it}$  is the only unobservable entering the intermediate input demand function. This scalar unobservable assumption together with the assumption that  $m_t(\cdot)$  is strictly increasing in  $\omega_{it}$  conditional on  $n_{it}, k_{it}$  and  $IMP_{it}$  (strict monotonicity assumption)<sup>9</sup>, allow to invert  $\omega_{it}$  as a function of observables:

$$\omega_{it} = m_t^{-1}(m_{it}, n_{it}, k_{it}, IMP_{it})$$
(22)

Considering the logarithmic version of Eq. (19) and allowing for an idiosyncratic error term including non-predictable output shocks and potential measurement error in output and inputs ( $\epsilon_{it}$ ) gives:

$$y_{it} = f(n_{it}, m_{it}, k_{it}; \boldsymbol{\beta}) + \omega_{it} + \epsilon_{it}$$
(23)

where  $y_{it} = q_{it} + \epsilon_{it}$  with  $\epsilon_{it}$  assumed to be mean independent of current and past input choices.<sup>10</sup>

We approximate  $f(\cdot)$  by a second-order polynomial where all logged inputs, logged inputs squared and interaction terms between logged inputs are included (translog production function):

$$y_{it} = \beta_0 + \beta_n n_{it} + \beta_m m_{it} + \beta_k k_{it} + \beta_{nn} n_{it}^2 + \beta_{mm} m_{it}^2 + \beta_{kk} k_{it}^2 + \beta_{nm} n_{it} m_{it} + \beta_{nk} n_{it} k_{it} + \beta_{mk} m_{it} k_{it} + \omega_{it} + \epsilon_{it}$$

$$(24)$$

where  $\beta_0$  has to be interpreted as the mean efficiency level across firms.

Substituting Eq. (22) in Eq. (24) results in a first-stage equation of the form:

$$y_{it} = f_{it} + m_t^{-1}(m_{it}, n_{it}, k_{it}, IMP_{it}) + \epsilon_{it} = \varphi_t(n_{it}, k_{it}, m_{it}, IMP_{it}) + \epsilon_{it}$$
(25)

which has the purpose of separating  $\omega_{it}$  from  $\epsilon_{it}$ , i.e. eliminating the portion of output  $y_{it}$  determined by unanticipated shocks at time t, measurement error or any other random noise ( $\epsilon_{it}$ ).

Hence, the first stage involves using Eq. (25) and the moment condition  $E[\epsilon_{it}|I_{it}] = 0$ , with  $I_{it}$  the firm's information set at t, to obtain an estimate  $\hat{\varphi}_{it}$  of the composite term  $\varphi_t(n_{it}, k_{it}, m_{it}, IMP_{it}) = f_{it} + m_t^{-1}(m_{it}, n_{it}, k_{it}, IMP_{it})$ , which represents output net of  $\epsilon_{it}$ . In our application, estimation of Eq. (25) is implemented by regressing output on a second-order polynomial series expansion where all logged inputs, logged inputs squared and interaction terms between logged inputs are included. To allow for time variation in  $\varphi_t$ , these polynomial terms are interacted with a time trend.

et al., 2017). Intuitively, the non-identification problem would arise under  $m_{it} = m_t(n_{it}, k_{it}, \omega_{it})$ , because in that case, the only intermediate input demand shifter aside from the other inputs in the production functions would be  $\omega_{it}$ . As the elasticity of output with respect to intermediate inputs is identified with how output varies with  $m_{it}$ , holding fixed  $(n_{it}, k_{it})$ , the only source of variation in  $m_{it}$  (namely  $\omega_{it}$ ) would also simulteaneously shift output, causing the elasticity of output with respect to materials to be unidentified.

<sup>&</sup>lt;sup>9</sup>Melitz and Levinsohn (2006) show that this strict monotonicity assumption holds as long as more productive firms do not set inordinately higher markups than less productive firms. Under Cournot competition, lower marginal costs (higher  $\omega_{it}$ ) lead to an increase in a firm's usage of intermediate inputs at any level of residual demand.

to an increase in a firm's usage of intermediate inputs at any level of residual demand. <sup>10</sup>Note that  $(\varepsilon_N^Q)_{it} = \frac{\partial \ln F(\cdot)}{\partial \ln N_{it}}$  and  $(\varepsilon_M^Q)_{it} = \frac{\partial \ln F(\cdot)}{\partial \ln M_{it}}$ . These output elasticities are by definition independent of a firm's productivity shock.

Given a particular set of parameters  $\beta$ , we can compute (up to a scalar constant) an estimate of  $\omega_{it}$ :

$$\widehat{\omega}_{it}(\boldsymbol{\beta}) = \widehat{m}_{t}^{-1}(m_{it}, n_{it}, k_{it}, IMP_{it})$$

$$= \widehat{\varphi}_{it} - \beta_{0} - \beta_{n}n_{it} - \beta_{m}m_{it} - \beta_{k}k_{it} - \beta_{nn}n_{it}^{2} - \beta_{mm}m_{it}^{2} - \beta_{kk}k_{it}^{2}$$

$$-\beta_{nm}n_{it}m_{it} - \beta_{nk}n_{it}k_{it} - \beta_{mk}m_{it}k_{it}$$

$$(26)$$

In order to implement the second stage and to identify the production function coefficients, we need to recover the innovation to productivity  $\xi_{it}$  to form moments on. Using Eq. (26), a consistent (nonparametric) approximation to  $E[\omega_{it}|\omega_{it-1}, IMP_{it-1}]$  is given by the predicted values from regressing nonparametrically  $\hat{\omega}_{it}(\boldsymbol{\beta})$  on  $\hat{\omega}_{it-1}(\boldsymbol{\beta})$  and  $IMP_{it-1}$ . The residual from this regression provides us with an estimate of  $\xi_{it}$ .

Given the timing assumptions on input use, the following population moment conditions can be defined:  $E[\xi_{it}(\beta)\mathbf{d}] = 0$  where the set of instruments is:

$$\mathbf{d}_{it} = \left\{ n_{it-1}, m_{it-1}, k_{it}, n_{it-1}^2, m_{it-1}^2, k_{it}^2, n_{it-1}m_{it-1}, n_{it-1}k_{it}, m_{it-1}k_{it} \right\}$$
(27)

Exploiting these moment conditions, we can now estimate the production function coefficients  $\beta$ using standard GMM and rely on block bootstrapping for the standard errors. The estimated production function coefficients  $\hat{\beta}$  are then used together with data on inputs to compute the output elasticities at the firm-year level. In particular, we calculate the elasticity of output with respect to labor at the firm-year level as:

$$(\widehat{\varepsilon}_{N}^{Q})_{it} = \widehat{\beta}_{n} + 2\widehat{\beta}_{nn}n_{it} + \widehat{\beta}_{nm}m_{it} + \widehat{\beta}_{nk}k_{it}$$
(28)

Similarly, we calculate the elasticity of output with respect to material at the firm-year level as:<sup>11</sup>

$$(\widehat{\varepsilon}_{M}^{Q})_{it} = \widehat{\beta}_{m} + 2\widehat{\beta}_{mm}m_{it} + \widehat{\beta}_{mn}n_{it} + \widehat{\beta}_{mk}k_{it}$$
<sup>(29)</sup>

Using the shares of labor and intermediate input expenditure in total sales,  $s_{Nit}$  and  $s_{Mit}$ , respectively, and our estimates of the output elasticities,  $(\hat{\varepsilon}_N^Q)_{it}$  and  $(\hat{\varepsilon}_M^Q)_{it}$ , we are able to compute  $\hat{\mu}_{it}$ and  $\hat{\psi}_{it}$ . Since we only observe  $Y_{it} = Q_{it} \exp(\epsilon_{it})$ , we do not observe the correct expenditure shares for  $N_{it}$  and  $M_{it}$ . We can recover an estimate of  $\epsilon_{it}$  from the first stage to adjust the expenditure shares as follows:<sup>12</sup>

$$\widehat{s}_{Nit} = \frac{W_{it}N_{it}}{P_t \frac{Y_{it}}{\exp(\epsilon_{it})}} \tag{30}$$

$$\widehat{s}_{Mit} = \frac{J_{it}M_{it}}{P_t \frac{Y_{it}}{\exp(\epsilon_{it})}} \tag{31}$$

<sup>&</sup>lt;sup>11</sup>Under a Cobb-Douglas production function,  $(\varepsilon_N^Q)_{it}$  and  $(\varepsilon_M^Q)_{it}$  would be equal to  $\hat{\beta}_n$  and  $\hat{\beta}_m$ , respectively. <sup>12</sup>This correction is important as it eliminates any variation in expenditure shares that comes from variation in output not correlated with  $\varphi_t(\cdot)$ .

Using Eqs. (28), (29), (30) and (31), we compute  $\hat{\mu}_{it}$  and  $\hat{\psi}_{it}$  as follows:

$$\widehat{\mu}_{it} = \frac{(\widehat{\varepsilon}_M^Q)_{it}}{\widehat{s}_{Mit}} \tag{32}$$

$$\widehat{\psi}_{it} = \frac{(\widehat{\varepsilon}_M^Q)_{it}}{\widehat{s}_{Mit}} - \frac{(\widehat{\varepsilon}_N^Q)_{it}}{\widehat{s}_{Nit}}$$
(33)

Based on the estimates  $\hat{\mu}_{it}$  and  $\hat{\psi}_{it}$ , we are able to determine the product market setting PMS  $\in$  {PC,IC} and the labor market setting LMS  $\in$  {PR,EB,MO} of firm *i* at time *t* and hence, firm *i*'s regime of competitiveness  $\mathbf{R} \in \Re =$  {PC-PR,IC-PR,PC-EB,IC-EB,PC-MO,IC-MO} at time *t* as follows. We first compute the 95% two-sided confidence intervals (CI) for  $\mu_{it}$  and  $gap_{Nit} = \frac{(\varepsilon_N^Q)_{it}}{\hat{s}_{Nit}}$ .

95% confidence interval for  $\mu_{it}$ :

$$\left[\widehat{\mu}_{it} - 1.96 \times \widehat{\sigma}_{\widehat{\mu}_{it}}, \widehat{\mu}_{it} + 1.96 \times \widehat{\sigma}_{\widehat{\mu}_{it}}\right] = \left[A_{\widehat{\mu}_{it}}, B_{\widehat{\mu}_{it}}\right]$$
(34)

with  $\hat{\sigma}_{\hat{\mu}_{it}}$  the standard error of  $\hat{\mu}_{it}$ , which is an estimator of the standard deviation of the sampling distribution of  $\hat{\mu}_{it}$ .

95% confidence interval for  $gap_{Nit}$ :

$$\left[\widehat{gap}_{Nit} - 1.96 \times \widehat{\sigma}_{\widehat{gap}_{Nit}}, \widehat{gap}_{Nit} + 1.96 \times \widehat{\sigma}_{\widehat{gap}_{Nit}}\right] = \left[A_{\widehat{gap}_{Nit}}, B_{\widehat{gap}_{Nit}}\right]$$
(35)

with  $\widehat{\sigma}_{\widehat{gap}_{Nit}}$  the standard error of  $\widehat{gap}_{Nit}$ .

To determine firm *i*'s PMS at time *t*, we use the 95% CI for  $\mu_{it}$ . If the lower bound of the 95% CI  $(A_{\hat{\mu}_{it}})$  is lower than or equal to unity, firm *i* is characterized to be perfectly competitive (PC) at time *t*. If  $A_{\hat{\mu}_{it}}$  exceeds unity, firm *i* is characterized by imperfect competition (IC) at time *t*.

To determine firm i's LMS at time t, we compare the 95% CIs for  $gap_{Nit}$  and  $\mu_{it}$ . In particular, firm i

- is characterized by perfect competition/right-to-manage bargaining (PR) at time t if the 95% CIs for  $gap_{Nit}$  and  $\mu_{it}$  overlap which implies that  $\hat{\mu}_{it}$  is not significantly different from  $\widehat{gap}_{Nit}$ , hence  $\hat{\psi}_{it} = 0$  at the 5% significance level.
- is characterized by efficient bargaining (EB) at time t if  $A_{\hat{\mu}_{it}} > B_{\widehat{gap}_{Nit}}$ , hence  $\hat{\psi}_{it} > 0$  at the 5% significance level.
- is characterized by monopsony (MO) at time t if  $A_{\widehat{gap}_{Nit}} > B_{\widehat{\mu}_{it}}$ , hence  $\widehat{\psi}_{it} < 0$  at the 5% significance level.

Once firm *i*'s regime at time *t* is determined, we are able to quantify market power in product and labor markets. As explained in Section 2, the product and labor market imperfection parameters are derived from the estimated joint market imperfections parameter  $\hat{\psi}_{it}$  and their respective standard errors are computed using the Delta method (Wooldridge, 2002).

#### 4 Data

Our data come from the Basic Survey of Japanese Business Structure and Activities (BSJBSA) compiled by the Ministry of Economy, Trade, and Industry (METI) in Japan. The purpose of this survey is to capture an overall picture of Japanese corporate activities, including globalization and diversification, as well as basic corporate characteristics, including sales, costs, profits, employment, assets and debt. The survey is compulsory for firms with more than 50 employees and with capital of more than 30 million yen in both manufacturing and some service industries such as wholesale trade, retail trade, and information and communication. In this study, we focus on manufacturing firms only.

In the BSJBSA, an industry classification code is assigned to each firm based on their main activities. For example, let us assume that a firm engages in both manufacturing and wholesale trade activities. If its largest revenue comes from manufacturing activities, the firm is classified as a manufacturing firm. This implies that manufacturing firms do not necessarily engage in manufacturing activities only. Some firms switch from one industry to another during the sample period. Although switching behavior of firms is an important issue, we assign each firm to the industry to which it belongs most frequently during our sample period.

The variables involved in our regression analyses are defined and measured in the following way. Output (Q) is defined as real gross output measured by nominal sales divided by an industry-wide gross output price index. Labor (N) refers to the average number of permanent workers. Material input is defined as intermediate consumption deflated by an industry-wide intermediate consumption price index. The capital stock (K) is measured by the real capital stock computed from tangible assets and investment based on the perpetual inventory method. The price deflators are obtained from the Japan Industrial Productivity (JIP) 2014 database, which was compiled by RIETI and Hitotsubashi University.<sup>13</sup> The shares of labor  $(s_N)$  and material input  $(s_M)$  are constructed by dividing respectively the firm total labor cost and undeflated intermediate consumption by the firm undeflated production. The cost of capital is defined as the user cost of capital times the real capital stock. The user cost of capital is computed from the investment goods price deflator times the sum of the interest rate and the depreciation rate minus changes in the investment goods price. In addition, we use the firm's age and its share of non-production workers as controls in the regression models, where the latter is defined as the ratio of non-production workers to total employees. We calculate the Herfindahl-Hirschman index (HHI) at the industry-year level to obtain a measure of market concentration.

We first deleted firm-year observations with cost shares greater than or equal to one and smaller than or equal to zero. In order to remove outliers, we also disregarded firm-year observations with cost

 $<sup>^{13}</sup>$ For more details on the JIP database, see Fukao *et al.* (2007).

shares in the bottom 1% and top 1% of the respective industry-year distributions. We selected firms that survive at least two consecutive years because lagged inputs are needed to construct moment conditions in our estimation framework. We obtain an unbalanced estimation sample consisting of 64,481 observations for 7,458 firms over the years 1994-2012, which we decompose into 15 two-digit industries. Table A.1 in Appendix reports the panel structure of the estimation sample. Table A.2 reports the number of observations and firms by industry.

In addition to standard firm accounting information and the control variables mentioned above, the BSJBSA also provides information on firms' export and import behavior and foreign direct investment (FDI). A firm reporting positive exports is classified as an exporter. Multinational enterprises (MNEs) consist of two types of firms: foreign-owned firms and Japanese firms that engage in FDI. A foreign-owned firm is defined as a firm with a foreign capital share greater than 50% and with headquarters located outside of Japan. A firm that has at least one foreign affiliate is regarded as a firm engaging in FDI.<sup>14</sup> From Table A.2 in Appendix, it follows that a minority of firms within an industry export and/or have networks of foreign affiliates: the overall share of manufacturing firms that export is 25% and 16% of firms are identified as MNEs. 46% of exporters are MNEs and 73% of MNEs are exporting. There is considerable variation in export market participation rates and in the importance of FDI as a mode of serving the foreign market across manufacturing industries. In particular, the share of exporters ranges from only 6% in wood, wooden products and furniture to 50% in chemicals. Likewise, the share of MNEs ranges from only 4% in pulp, paper and paper products to 27% in chemicals. The shares of exporting and importing firms are significantly positively correlated across industries. Approximately 71% of exporters and 53%of MNEs also import. These findings are consistent with evidence in a wide range of other countries (see e.g. Mayer and Ottaviano, 2008; World Trade Organization, 2008; Bernard et al., 2012).

Table 1 reports the means, standard deviations and quartile values of our variables for the total estimation sample and split according to international activity. In the total estimation sample, real firm output, labor, materials and the Solow residual or conventional TFP measure have been stable over the considered period while capital has decreased at an average annual growth rate of 4.3%. On average, firm age equals 45 years, 35% of total employees are non-production workers and the price-cost margin amounts to 22%. Consistent with previous studies, our data reveal that exporters are systematically different from non-exporters. Among manufacturing firms, exporters pay higher wages, are larger, older, more capital-intensive, employ more non-production workers and are more productive. Table 1 also reveals that MNEs show the same performance differences as exporters.

#### < Table 1 about here>

Table A.3 in Appendix confirms these observations by summarizing the average percent difference for a particular characteristic between either exporters and non-exporters, or between MNEs and

<sup>&</sup>lt;sup>14</sup>If foreign-owned firms also have foreign affiliates outside Japan, they are not classified as FDI firms but as foreign-owned firms. In the BSJBSA, a Japanese foreign affiliate is defined as an affiliate with a capital share of more than 20%.

non-MNEs. The set of characteristics include the logarithms of firm size (employment), value added per worker, TFP, average wages, capital per worker, share of non-production workers and price-cost markups. The firm-year varying TFP and markup estimates are obtained by estimating translog production functions separately for each of our 15 industries. In order to ensure that the strict monotonicity assumption between productivity and intermediate inputs holds, we follow e.g. De Loecker and Warzynski (2012) and Ahsan and Mitra (2014) by ruling out inordinately high markups in the remainder of our empirical analysis.<sup>15</sup>

All results in column (1) of Table A.3 are from bivariate OLS regressions of a firm characteristic on a dummy variable indicating either a firm's export status or a firm's MNE status. Column (2) includes industry fixed effects and the logarithm of firm size as additional controls. Column (1) shows that there are substantial mean differences between exporters and non-exporters, and between MNEs and non-MNEs. As export/MNE participation is correlated with industry characteristics and firm size. the inclusion of industry fixed effects and firm size in column (2) reduces the magnitude of these coefficients. Exporters remain different from non-exporters even within the same disaggregated industry: Exporters are more productive by 13% for value added per worker and by 2.0% for total factor productivity, they pay higher wages by approximately 9% and are relatively more capitaland skill-intensive than non-exporters by approximately 7% and 30%, respectively. MNEs exhibit similar performance differences as exporters. One difference, though, is that there does not seem to be significant markup differences between exporters and non-exporters within the same industry whereas markups appear to be approximately 6% lower in MNEs. One tentative explanation is that MNEs might also engage in service activities where price competition is more likely to be the dominant form of competition. This could generate the negative MNE markup premium, despite the observed positive productivity differential between MNEs and non-MNEs.

Table A.4 in Appendix illustrates large variation in mean TFP and markup differences across industries. Column (1) again reports these differences without conditioning on additional covariates while column (2) includes the logarithm of firm size as an additional covariate. Controlling for firm size, we observe a significantly positive exporter productivity premium in about two thirds of the industries with this premium varying between 0.5% (transport equipment) and 5.1% (electrical machinery), whereas we do not find significant mean differences in markups between exporters and non-exporters (except for 2 industries). Similarly, the significantly positive MNE productivity premium holds for two thirds of the industries and ranges between 0.4% (food products and beverages) and 9.5% (electrical machinery). Consistent with the finding for all manufacturing firms (see Table A.3), average markups in non-MNEs appear to be higher than in MNEs in two thirds of the industries.

<sup>&</sup>lt;sup>15</sup>In particular, we trimmed the parameter estimates of  $\mu_{it}$  and  $gap_{Nit}$  at the 1<sup>st</sup> and 99<sup>th</sup> percentiles of the respective industry-year distributions to remove outliers.

To examine firm-level persistence in the mode of serving the foreign market, we looked at one-year transition probability rates from period t to period t + 1 of internationalization types across states over the considered period. The states are either defined as exporters and non-exporters, or as MNEs and non-MNEs. We find strong persistence in internationalization types among all manufacturing firms as we observe the highest values on the diagonal for each state: 96.7% of exporters and 94.1% of non-exporters, and 98.2% of MNEs and 97% of non-MNEs remain in their initial state.

#### 5 Firm heterogeneity in regimes of competitiveness

Based on the estimates of  $\mu_{it}$  and  $\psi_{it}$ , we obtain firm-year varying product market settings, labor market settings and regimes. We first examined firm-level persistence in the type of competition prevailing in product and labor markets and, hence, in the regime of competitiveness by investigating one-year transition probability rates across respective states over the period, where the states are defined as {PC,IC} in the case of PMS, {PR,EB,MO} in the case of LMS and {PC-PR,IC-PR,PC-EB,IC-EB,PC-MO,IC-MO} in the case of R. At the overall level, we find rather strong persistence in types of competitiveness as we observe the highest values on the diagonal for each regime. In particular, the fraction of firms remaining in their initial state ranges between 70% (PC-PR) and 93% (PC-MO). However, firm-year transitions appear to be important and the degree of persistence in regimes varies considerably across industries.<sup>16</sup>

We then determined the firm-specific PMS, LMS and regime by retaining the relevant type (PMS/LMS/R) that occurs most frequently in order to examine the link between the internationalization mode of firms and the type of product and labor market competition in a descriptive way. Table A.5 in Appendix presents the percentage of firms belonging to each of the six regimes of competitiveness for different subsets of firms. Among all manufacturing firms, about 25% are characterized by perfect competition and 75% by imperfect competition in the product market. The dominant labor market setting is efficient bargaining (EB; 42% of the firms), followed by perfect competition/right-to-manage bargaining (PR; 30% of the firms) and monopsony (MO; 28% of the firms). As such, the predominant regimes are IC-EB (42% of the firms), IC-PR (18% of the firms) and IC-MO (15% of the firms).

Let us now focus on the prevalence of regimes across firms that differ in terms of international activities. Comparing firms that differ according to export status reveals that a larger fraction of exporters are characterized by PMS=IC (79% of exporters compared to 67% of non-exporters). Exporters are dominantly characterized by efficient bargaining (46% of exporting firms) and far less so by monopsony (only 22% of exporting firms) whereas the three labor market settings are more evenly distributed among non-exporters. As such, market power on the supply side seems to be

<sup>&</sup>lt;sup>16</sup>For example, in transport equipment, only 14% of firms characterized by PC-PR stay in their initial state while in iron and steel, this holds for as much as 96% of firms typified by PC-MO.

predominantly responsible for introducing allocative inefficiencies through distorting factor prices among exporters. The distribution of product and labor market settings and regimes across MNEs (non-MNEs) is very similar to the one across exporters (non-exporters).

Table A.6 in Appendix shows considerable heterogeneity in regimes across and within manufacturing industries. On the product market side, the fraction of firms being characterized by perfect competition ranges between 0.2% (transport equipment) and 85% (iron and steel). On the labor market side, only 1.8% of firms within machinery are characterized by LMS=PR while this is true for 77% of firms in non-ferrous metals. The fraction of firms where market power is consolidated on the labor supply side (LMS=EB) varies between 1.6% (iron and steel) and 81% (transport equipment). Firms in industries manufacturing either wood, paper products or electrical machinery do not appear to have wage-setting power whereas the fraction of firms characterized by monopsony is 95% in machinery. Within industries, we observe large variation in regimes. For example, 37% of the firms in non-metallic mineral products are characterized by R=IC-PR, 29% by R=IC-EB, 22% by R=IC-MO and 8% by R=PC-MO whereas 83% of the firms in transport equipment belong to R=IC-EB.

The descriptive analysis presented above does not give a detailed picture on potential differences in firms' regimes across modes of internationalization for two main reasons. First, it does not exploit time variation in a firm's product and labor market setting. Second, it does not take into account correlations between firm observables and a firm's export/MNE status which could partially account for differences between exporters and non-exporters and/or between MNEs and non-MNEs. Indeed, firm *i*'s product market setting at time *t* might depend on its engagement in international activities, other observable characteristics as well as unobservable factors  $\epsilon$  such as managerial ability. To allow the marginal effect of being an exporter (MNE) to depend on MNE (EXP) status, we include an interaction term which is the product of the binary variables export status and MNE status. Suppressing firm and time subscripts (*i* and *t*, respectively) for simplicity, we thus have:

$$PMS^* = \beta_0 + \beta_1 EXP + \beta_2 MNE + \beta_3 (EXP \times MNE) + \beta_4 \widehat{\omega} + \mathbf{z}\beta_z + \epsilon$$
(36)

with EXP export status, MNE MNE status and  $\hat{\omega}$  estimated TFP.<sup>17</sup> The vector **z** comprises firmyear varying variables such as a firm's size (number of workers), age, the share of non-production workers, the Herfindahl-Hirschman index, a set of time dummies and industry fixed effects. Table A.7 in Appendix reports cross-correlations between these variables. In order to investigate the link between the internationalization mode of firms and the likelihood of being characterized by imperfect competition in the product market, we specify the following probit model:

$$Pr(PMS=IC|\mathbf{x}) = \Phi(\mathbf{x}\beta)$$
(37)

 $<sup>^{17}</sup>$ Since productivity is inherently a relative concept, we normalize the firm-year productivity-level estimates in the remaining regression models.

The baseline category is PMS=PC and the vector **x** includes the regressors specified in Eq. (36).

Whether market power in firm i in period t is consolidated on either the supply side or the demand side of labor might be influenced by common observable as well as unobservable factors such as a firm's corporate culture. To take into account the full covariance structure and to investigate the link between the internationalization mode of firms and the likelihood of being characterized by either efficient bargaining or monopsony, we specify the following two-equation multivariate probit model:

$$LMS_{m}^{*} = \mathbf{x}_{m}\beta_{m} + \epsilon_{m}, \quad m = 1, 2$$
  

$$LMS_{m} = I(LMS_{m}^{*} > 0), \quad m = 1, 2$$
  

$$\boldsymbol{\epsilon} = (\epsilon_{1}, \epsilon_{2})\prime \sim N(0, \Sigma)$$
(38)

where  $LMS_1 = Pr(LMS=EB|\mathbf{x})$  and  $LMS_2 = Pr(LMS=MO|\mathbf{x})$ . The baseline category is LMS=PR. We include the same regressors as in the univariate probit model (Eq. (37)).

Table 2 presents the marginal effect of the regressors in the univariate and the multivariate probit models. As such, columns 1 and 4 report how much the (conditional) probability of being characterized by PMS=IC changes when the value of a regressor changes, holding all other regressors constant whereas columns 2-3 and 5-6 show how much the likelihood of being characterized by either LMS=EB or LMS=MO changes. Accounting for the use of a generated regressor, we employ block bootstrapping for statistical inference.

We consider two specifications. The baseline specification (specification 1) permits testing the hypothesis that the effect of serving the foreign market through exporting (FDI) is the same for MNEs and non-MNEs (exporters and non-exporters) whereas specification 2 does not include the interacted regressor. The parameters of interest are  $\beta_1$ ,  $\beta_2$  and  $\beta_3$ .

The estimates of the baseline specification indicate that the coefficient on the interaction term (EXP×MNE) is not significantly different from zero in both the univariate and bivariate probit models. Therefore, we rely on the estimates of specification 2 and focus the discussion on our variables of interest. Being an exporter increases the likelihood of being characterized by imperfect competition in the product market by 1.4 percentage points. This finding is consistent with empirical support of the prediction that exporters charge higher price-cost markups relative to non-exporters. Being an exporter increases the likelihood of being characterized by LMS=EB by 3.7 percentage points whereas it decreases the likelihood of being characterized by LMS=MO by 2.4 percentage points. Put differently, exporting firms are more likely to share rents based on the bargaining power of workers, but less likely to share rents based on the elasticity of the labor supply curve facing an individual employer. This former rent-sharing mechanism could be explained by the fact that

exporters, which are relatively more productive, charge higher markups and realize higher rents, are willing to share part of these rents with their workers according to a surplus-sharing rule.

When focusing on correlations between being an MNE and the likelihood of being characterized by PMS=IC, LMS=EB or LMS=MO, respectively, we get a completely different picture. We interpret this as evidence of clear differences in behavior between firms that serve the foreign market either through exporting or through FDI. More precisely, being an MNE decreases the probability of being characterized by imperfect competition in the product market by 2.3 percentage points. Being a firm that serves the foreign market through FDI decreases the likelihood of being characterized by LMS=EB by 5.8 percentage points whereas it increases the likelihood of being characterized by LMS=MO by 2.0 percentage points. The latter finding is compatible with MNEs having considerable monopsony power in the labor market due to e.g. high intra-firm labor replacement in such firms.

#### < Table 2 about here>

Table A.8 in Appendix reports variation in correlations between the probability of being characterized by PMS=IC, PMS=EB or PMS=MO and firm observables across industries. Using our baseline specification, we present the marginal effects of export status, MNE status and the interacted regressor EXP×MNE in the univariate and multivariate probit models.<sup>18</sup> Three findings stand out.

First, we find that engagement in international activities matters for determining the type of competition in product and labor markets within quite a few industries. More precisely, we find a significantly positive correlation between being an exporter and the probability of being characterized by imperfect competition in the product in six out of fifteen industries, in particular industries manufacturing textiles and wearing apparel, wood, wooden products and furniture, publishing and printing, petroleum and coal products, transport equipment, and other manufacturing (industries  $J = \{2, 3, 5, 7, 14, 15\}$ ). We observe a significantly positive correlation between being an exporter and being characterized by LMS=EB in three industries (food products and beverages, chemicals, and petroleum and coal products). A significantly negative correlation is observed between being an exporter and the probability of being characterized by LMS=MO in four industries (food products and beverages, textiles and wearing apparel, chemicals, and non-ferrous metals). In non-metallic mineral products, non-ferrous metals and machinery, there is a significantly negative correlation between being an MNE and being characterized by PMS=IC whereas a significantly positive correlation is found in publishing and printing, and transport equipment. A significantly negative correlation is found between being an MNE and having either market power on the labor supply side (in textiles and wearing apparel, non-ferrous metals, and electrical machinery) or market power on the labor demand side (in other manufacturing).

<sup>&</sup>lt;sup>18</sup>The marginal effects of the other covariates are available upon request.

Second, the effect of engagement in international activities on the type of competition in product and labor markets within industries clearly differs between firms that serve the foreign market either through exporting or through FDI in the majority of industries. This finding is based on testing the equality of marginal effects of export status and MNE status.

Third, the hypothesis that the effect of being an exporter is the same for MNEs and non-MNEs, or put differently, the effect of being an MNE is the same for exporters and non-exporters on the probability of being characterized by PMS=IC (LMS=EB or LMS=MO, respectively) is rejected at conventional significance levels in 33% of the industries (21% or 33% of the industries, respectively). For example, we find a large, significantly negative marginal effect of the interacted regressor on the probability of being characterized by PMS=IC in industries manufacturing textiles, wooden products, and publishing and printing (industries  $J = \{2, 3, 5\}$ ). As such, within these industries, we observe a large significantly positive correlation between being an exporter that does not engage in FDI and the probability of being characterized by PMS=IC while the correlation is close to zero or even negative for exporters that engage in FDI. Likewise, the correlation between a non-exporting MNE and the likelihood of being characterized by PMS=IC is zero or slightly positive whereas a large significantly negative correlation is observed for exporting MNEs within these industries.

#### 6 Market imperfections and export/MNE status

To get a first insight into the link between the internationalization type of firms and the degree of product and labor market imperfections, Table A.9 in Appendix reports median values of estimated parameters – markups ( $\mu$ ), labor market imperfections (workers' bargaining power  $\phi$  or the wage elasticity of a firm's labor supply curve  $\varepsilon_w^N$ ) and productivity ( $\omega$ )– for subsets of firms within a particular regime. We define subsets of firms based on their engagement in international activities.

Let us focus the discussion on the regimes characterized by imperfect competition in product as well as labor markets. Conditional on being characterized by R=IC-EB, we find that the median value of markup estimates is lower for exporters relative to non-exporters (1.42 versus 1.55). When comparing MNEs to non-MNEs, this discrepancy is larger (1.35 versus 1.56). In addition, workers in MNEs seem to have a slightly lower bargaining power than in non-MNEs (median value of 0.27 for the former and 0.30 for the latter). Offshoring could increase substitution between domestic and foreign workers, thereby flattening the labor demand curve and, therefore, shifting bargaining power over rent distribution from labor towards capital in MNEs that enjoy extra-normal profits (Rodrick, 1997; Slaughter, 2001; Senses, 2010).

Interestingly, the opposite picture appears when comparing subsets of firms, conditional on being characterized by R=IC-MO. Irrespective of whether firms serve the foreign market either through exporting or through FDI, we find that firms that engage in international activities seem to have

larger market power in both product and labor markets. More specifically, the median value of markups is 1.21 for exporters compared to 1.16 for exporters and the median value of an individual firm's labor supply elasticity is 1.48 for exporters compared to 1.72 for non-exporters, implying that exporters have larger wage-setting power. On the labor side, the discrepancy is even larger when comparing MNEs to non-MNEs (median value of  $\varepsilon_w^N$  is 1.32 for the former and 1.74 for the latter).

The observed differences in the degree of market imperfection parameters discussed so far could, however, partly been driven by correlations between firm observables and a firm's export/MNE status. To address this concern, we examine the links between the internationalization mode of firms and the degree of market imperfections within a regression framework. We first estimate the average effect of export/MNE status (and other independent variables) on the degree of product and labor market imperfections in a 'representative enterprise'. As such, we define the following regression models:<sup>19</sup>

$$\ln \hat{\mu}_{it+1} = \alpha_0 + \alpha_1 \text{EXP}_{it} + \alpha_2 \text{MNE}_{it} + \alpha_3 (\text{EXP} \times \text{MNE})_{it} + \alpha_4 \hat{\omega}_{it} + \alpha_5 \text{IMR}_{it} + \mathbf{z}_{it} \alpha_z + \zeta_{it} \quad (39)$$

$$\ln\left(\frac{\widehat{\phi}_{it+1}}{1-\widehat{\phi}_{it+1}}\right) = \alpha_0 + \alpha_1 \text{EXP}_{it} + \alpha_2 \text{MNE}_{it} + \alpha_3 \left(\text{EXP} \times \text{MNE}\right)_{it} + \alpha_4 \widehat{\omega}_{it} + \alpha_5 \text{IMR}_{it} + \mathbf{z}_{it} \alpha_z + \zeta_{it} \quad (40)$$

$$\ln(\widehat{\varepsilon}_{w}^{N})_{it+1} = \alpha_{0} + \alpha_{1} \text{EXP}_{it} + \alpha_{2} \text{MNE}_{it} + \alpha_{3} \left(\text{EXP} \times \text{MNE}\right)_{it} + \alpha_{4} \widehat{\omega}_{it} + \alpha_{5} \text{IMR}_{it} + \mathbf{z}_{it} \alpha_{z} + \zeta_{it} \quad (41)$$

with IMR the inverse Mills ratio from the respective probit model, which we include to account for selection bias, and the vector  $\mathbf{z}$  comprising the same regressors as in Section 5. Because the effect of our regressors of interest might not be instantaneous, we use the one-year lead of the dependent variables. To deal with generated regressands and regressors, we use block bootstrapping for statistical inference. As the share of rents captured by the workers ( $\phi$ ) lies within the [0, 1]-range, we use a logit transformation to model the bargaining power of workers.

Table 3 presents the average effect of the regressors in the three regression models. Similar to the probit models specified above, we consider two specifications. The baseline specification (specification 1) permits testing the hypothesis that the effect of serving the foreign market through exporting (FDI) is the same for MNEs and non-MNEs (exporters and non-exporters) whereas specification 2 does not include the interacted regressor. The parameters of interest are  $\alpha_1$ ,  $\alpha_2$  and  $\alpha_3$ .

The estimates of the baseline specification indicate that the coefficient on the interaction term  $(EXP \times MNE)$  is not significantly different from zero in the three regression models. Therefore, we rely on the estimates of specification 2 and focus on our regressors of interest. Conditional on being

<sup>&</sup>lt;sup>19</sup>One could argue that observable firm characteristics might correlate with unobserved firm characteristics such as managerial ability or workplace environment, which would favor applying a fixed effects estimator. However, this would render the interpretation of the effect of e.g. being an exporter difficult. This is because when firm fixed effects are included, identification originates from changes in export status, implying that the benchmark would also comprise continuing exporters.

characterized by PMS=IC, we observe a significantly positive correlation between export status and product market power ( $\hat{\mu}$ ). Conditional on being characterized by LMS=EB, we find a significantly positive correlation between export status and labor market power consolidated on the labor supply side ( $\hat{\phi}$ ). This might be interpreted as exporters being willing to share a larger part of increased rents with their workers. Conditional on being characterized by LMS=MO, we detect a significantly positive correlation between export status and the wage elasticity of a firm's labor supply curve, implying that exporters are less able to exploit wage-setting power. Based on hypothesis testing, we clearly reject the equality of the estimated coefficients on export status and MNE status in each of the three regression models. In fact, the opposite picture appears for firms that engage in FDI. More precisely, we observe a significantly negative correlation between MNE status and either product market power or workers' bargaining power. The former finding could be explained by the fact that MNEs might also engage in service activities where price competition is fierce whereas the latter again confirms that (the threat of) relocating plants from home to foreign countries might involve organizational changes that reduce workers' bargaining power.

#### < Table 3 about here>

Table A.10 in Appendix presents cross-industry variation in the average effects of regressors of interest (EXP, MNE and EXP $\times$ MNE) in the baseline specification for each regression model (Eqs. (39)-(41)). Our main findings are the following. First, engagement in international activities matters for determining the degree of imperfections in product and labor markets within quite a few industries. This particularly holds for the correlations between export status and the respective product and labor market imperfection parameters. Overall, the statistically significant correlations between either export or MNE status and estimated parameters of imperfections in product and labor markets are qualitatively similar to the ones where all industries are pooled (Table 3). Second, the effect of engagement in international activities on the degree of product and labor market imperfections clearly differs between firms that serve the foreign market either through exporting or through FDI in the majority of industries. This differential effect is most pronounced in the regression model of the firm's wage-setting power  $(\widehat{\varepsilon}_w^N)$ . Third, the hypothesis that the effect of being an exporter is the same for MNEs and non-MNEs, or the effect of being an MNE is the same for exporters and non-exporters is rejected at conventional significance levels in 20% of the industries for the regression model of a firm's markup. This is only true in 14% of the industries for the regression model of workers' bargaining power but holds in 56% of the industries for the regression model of a firm's wage-setting power. For example, we find a large, significantly negative coefficient of the interacted regressor in the latter regression model in industries manufacturing food products and beverages, chemicals, iron and steel, and other manufacturing (industries  $J = \{1, 6, 9, 15\}$ ). As such, in these industries, we observe a large significantly positive correlation between being an exporter that does not engage in FDI and a firm' labor supply elasticity while the correlation is much lower or even negative for exporters that engage in FDI. Likewise, the correlation between a non-exporting

MNE and a firm' labor supply elasticity is zero or positive whereas a large significantly negative correlation is observed for exporting MNEs in these industries.

The exclusive focus on mean effects may be misleading. Given that we acknowledge that firms are heterogeneous, we expect, e.g., differences in markups or workers' bargaining power between exporters and non-exporters to vary across firms. In order to study the impact of regressors, in particular engagement in international activities, on different quantiles of the outcome distribution where the vector of outcomes  $\mathbf{y} = \{\ln \hat{\mu}, \ln\left(\frac{\hat{\phi}}{1-\hat{\phi}}\right), \ln(\hat{\varepsilon}_w^N)\}$ , we use quantile regression techniques to model the conditional outcome distribution at various quantiles  $\tau$  ( $0 < \tau < 1$ ), conditional on the regressors  $\mathbf{x}$ . As such, we provide a complete picture of the relationship between  $\mathbf{y}$  and  $\mathbf{x}$ . The use of quantile regression techniques entails two other major advantages. First, while the optimal properties of standard regression estimators are not robust to modest departures from normality, quantile regression results are characteristically robust to outliers and heavy-tailed distributions of  $\mathbf{y}$ . Second, estimation and inference are distribution-free: quantile regression avoids the restrictive assumption that the error terms are *iid* at all points of the conditional distribution.

The quantile regression model, first introduced in Koenker and Bassett's (1978) seminal contribution, that we estimate is written as:

$$y_{it+1} = \mathbf{x}_{it}\beta_{\tau} + \zeta_{\tau it} \quad \text{with} \quad Q_{\tau}\left(y_{it+1}|x_{it}\right) = \mathbf{x}_{it}\beta_{\tau} \tag{42}$$

where  $y_{it+1}$  is the one-year lead of the dependent variable,  $\mathbf{x}_{it}$  a vector of regressors,  $\beta_{\tau}$  the vector of parameters to be estimated and  $\zeta_{\tau it}$  the error term.  $Q_{\tau}(y_{it+1}|x_{it})$  denotes the  $\tau^{th}$  conditional quantile of  $y_{it+1}$  given  $x_{it}$ . The quantile regression estimator of  $\beta_{\tau}$  solves the following minimization problem:

$$\beta_{\tau} = \arg\min_{\beta} \left\{ \sum_{it \in \{it: y_{it+1} \ge \mathbf{x}_{it}\beta\}} \tau |y_{it+1} - \mathbf{x}_{it}\beta| + \sum_{it \in \{it: y_{it+1} < \mathbf{x}_{it}\beta\}} (1 - \tau) |y_{it+1} - \mathbf{x}_{it}\beta| \right\}$$
(43)

The minimization problem (Eq. (43)) is formulated as a linear function of parameters and can be solved efficiently by linear programming methods. We use the simplex method which yields a solution in a finite number of iterations and is suitable for moderate data size (Koenker and D'Orey, 1987). Inference is based on bootstrapped standard errors from individual resampling which avoids direct estimation of the variance-covariance matrix. As one increases  $\tau$  continuously from 0 to 1, one traces the entire conditional distribution of y, conditional on x (Buchinsky, 1994).

In our study, the parameter estimate for the  $k^{th}$  exogenous variable, let's say export status (EXP<sub>it</sub>), on e.g. a firm's markup  $(\ln \hat{\mu}_{it+1})$  is interpreted as the marginal change in markup due to being an exporter conditional on being on the  $\tau^{th}$  quantile of the distribution. This is also called the  $\tau^{th}$  quantile return to exporting. We are particularly interested in how these returns change along the distribution. Similar to the OLS regression models, we consider three outcomes:  $\mathbf{y} = \{\ln \hat{\mu}, \ln\left(\frac{\hat{\phi}}{1-\hat{\phi}}\right), \ln(\hat{\varepsilon}_w^N)\}$  and, depending on the outcome, we use the same set of regressors  $\mathbf{x}$  as in specifications (39)-(41).

Table 4 reports the pooled simultaneous-quantile regression (QR) results of our baseline specification for the  $10^{th}$ ,  $25^{th}$ ,  $50^{th}$ ,  $75^{th}$  and  $90^{th}$  percentiles of the respective outcome distributions.<sup>20</sup> Graph 1 displays the estimated coefficients for our variables of interest (EXP, MNE and EXP×MNE) across all quantiles, together with the 95% confidence intervals. For comparison, the OLS estimates and their 95% confidence intervals are presented as dashed horizontal lines. Clearly, OLS estimates, making inferences about 'the average enterprise', mask important aspects of the relationship between our main variables and the three outcomes.

The left part of Table 4 and Graph 1(a) reveal a heterogeneous pattern for the effect of export/MNE status upon markups at different quantiles. Being an exporter yields significantly positive returns along the conditional markup distribution. The estimates point to increasing returns to exporting up to the  $80^{th}$  percentile, after which the positive returns start to decrease slightly. Hence, it seems that exporting firms that are already successful in charging relatively high markups are able to exploit their advantage of serving a foreign market, except at the very top of the distribution. Contrary to exporter returns, MNE returns are significantly negative along the full conditional markup distribution. The negative MNE returns are rather stable up to the  $80^{th}$  percentile, after which they increase steeply in absolute value. The coefficient of the interacted regressor is estimated to be significantly negative between the  $30^{th}$  and  $75^{th}$  percentiles, and increases in absolute value as we move up the distribution. This implies that the positive exporter returns are largest for exporters that do not engage in FDI whereas exporter returns are close to zero for exporters that engage in FDI. Likewise, negative returns for non-exporting MNEs are smaller in absolute value than negative returns for exporting MNEs.

The middle part of Table 4 and Graph 1(b) highlight non-linearities in the returns to being an exporter/MNE along the conditional workers' bargaining power distribution. Returns to exporting are significantly positive along the full conditional distribution. Firms yield increasing exporter returns up to the  $30^{th}$  percentile, stable returns between the  $30^{th}$  and  $75^{th}$  percentiles and again increasing returns from the  $75^{th}$  percentile onwards. Firms with high workers' bargaining power seem to benefit most from being an exporter. In contrast, MNE returns are significantly negative along the full conditional distribution and at an increasing rate between the  $20^{th}$  and  $40^{th}$  percentiles and at the top of the distribution. The effect of being an exporter (MNE) does not seem to depend on MNE (export) status, except at the  $30^{th}$  percentile.

 $<sup>\</sup>overline{ ^{20} \text{We estimated pooled simultaneous-quantile regressions for } \tau \in \{0.05, 0.10, 0.20, 0.25, 0.30, 0.40, 0.50, 0.60, 0.70, 0.75, 0.80, 0.90, 0.95\}.$ Table 4 shows results for selected quantiles.

The right part of Table 4 and Graph 1(c) show heterogeneous effects of export/MNE status upon labor supply elasticity estimates at different quantiles. More precisely, we observe non-linearities in the returns to exporting. The estimated coefficient for export status is significantly positive along the full distribution. The returns to exporting first increase up to the  $30^{th}$  percentile, decrease between the  $30^{th}$  and  $70^{th}$  percentiles and increase again from the  $70^{th}$  percentile onwards. In contrast, the estimated coefficient for MNE status is significantly negative from the  $10^{th}$  percentile onwards and increases in absolute value, reaching the most negative return at the upper quantiles. Hence, firms in the upper part of the distribution seem to benefit most from being an MNE in terms of wage-setting power. Similar to the quantile regression results for the regression model of workers' bargaining power, the coefficient of the interacted regressor is not significantly different from zero across quantiles.

#### < Table 4 and Graph 1 about here>

While Graph 1 already suggests a heterogeneous pattern for the effect of export/MNE status upon our market imperfection parameters, Table 4 presents additional tests to validate these inferences. For each of our main regressors, we run F-tests to test the equality of coefficients (i) across all quantiles, (ii) at the median and the tails, and (iii) at the tails. Differential effects across the conditional markup distribution are most pronounced as the three F-tests reject the equality of the coefficients of export and MNE status and the interacted regressor at the 1% significance level, except for differential impacts of export status at both tails of the conditional markup distribution. In contrast, the impact of our regressors upon the workers' bargaining power does not differ that much across quantiles, which is corroborated by the F-tests. Finally, the three F-tests suggest statistically significant differential effects of export status on a firm' labor supply elasticity, which confirms the observed non-linearities in the returns to exporting along the conditional labor supply elasticity distribution.

Table A.11 in Appendix reports cross-industry variation in how returns to being an exporter/MNE change along the conditional distribution of our three outcomes  $\left(\ln \hat{\mu}, \ln \left(\frac{\hat{\phi}}{1-\hat{\phi}}\right), \ln(\hat{\varepsilon}_w^N)\right)$ . We observe heterogeneous returns within an industry and also discern cross-industry differences. The within-industry variation explains why we do not detect statistically significant *average* returns in *all* industries (see Table A.10).<sup>21</sup> We selected industries for which we found statistically significant estimated coefficients at different points of the conditional outcome distributions for our variables of interest (EXP and MNE) and displayed these coefficient estimates in Graph A.1-A.3 in Appendix, where each graph focuses on one of the outcomes  $\left(\ln \hat{\mu}, \ln \left(\frac{\hat{\phi}}{1-\hat{\phi}}\right), \ln(\hat{\varepsilon}_w^N)\right)$ . We restrict the discussion to these selected industries.

 $<sup>^{21}</sup>$  For example, the statistically insignificant average impact of being an exporter on markups in textiles and wearing apparel (industry 2) can be explained by the fact that firms up to the  $60^{th}$  percentile experience significantly positive returns whereas firms in the upper tail yield insignificantly negative returns.

Focusing on the impact of export status at different points of the conditional markup distribution, we observe a positive exporter premium along most quantiles in industries manufacturing food products and beverages, textiles and wearing apparel, publishing and printing, chemicals, petroleum and coal products, non-metallic mineral products, fabricated metal products, transport equipment, and other manufacturing (industries  $J = \{1, 2, 5, 6, 7, 8, 11, 14, 15\}$ ). In contrast, firms in machinery (industry 12) experience negative exporter returns along the full distribution and most negative ones in the tails. Focusing on the impact of MNE, we find negative MNE returns along most quantiles for firms in industries manufacturing textiles and wearing apparel, non-metallic mineral products, non-ferrous metals, machinery, and electrical machinery (industries  $J = \{2, 8, 10, 12, 13\}$ ).

We classify industries in two groups in order to provide a more complete picture of export/MNE returns along the conditional workers' bargaining power distribution. The first group includes industries manufacturing petroleum and coal products, fabricated metal products, and transport equipment (industries  $J = \{7, 11, 14\}$ ). Firms in these industries yield positive exporter returns along most quantiles. The second group includes industries  $J = \{1, 2, 13\}$ . Firms in food products and beverages (industry 1) experience a negative impact of export status in upper quantiles, whereas negative exporter returns are found for firms in the upper quantiles in textiles and wearing apparel (industry 2), and electrical machinery (industry 13). Firms in pulp, paper and paper products (industry 4), electrical machinery (industry 13), and transport equipment (industry 14) experience negative MNE returns along most quantiles whereas we find positive MNE returns, and most pronounced so in the tails, for firms in fabricated metal products (industry 11).

Focusing on exporter returns along the conditional distribution of a firm's labor supply elasticity, we observe positive returns along the full conditional distribution for firms in industries  $J = \{1, 6, 9, 12\}$ . MNEs seem to have more wage-setting power along most quantiles in non-metallic mineral products (industry 8) and machinery (industry 12) whereas the opposite holds for firms in iron and steel (industry 9).

#### 7 Conclusion

Do the type and degree of labor market imperfections vary across firms that differ in terms of internationalization? In spite of the growing importance of labor market imperfections in recent international trade theory, this question has not been answered so far. Microeconometric studies in the field have predominantly provided evidence of the well-established productivity premium of firms with international activities relative to firms serving only domestic markets and have recently focused on the underlying sources of this productivity advantage.

This paper examines the links between the internationalization type of firms and market imperfections using Japanese firm-level data over the period 1994-2012. Our contribution to the empirical international trade literature and the econometric literature on identifying market imperfections is twofold. First, we develop a framework for modelling heterogeneity across firms in terms of (i) product market power (price-cost markups), (ii) labor market imperfections (workers' bargaining power during worker-firm negotiations or a firm's degree of wage-setting power) and (iii) revenue productivity. Second, we apply this framework in order to examine whether the type and the degree of market power in product and labor markets depends on a firm's international behavior, while accounting for differences in revenue productivity. We consider two main forms of internationalization: exports and foreign direct investment. As such, we are able to improve our understanding of the wage determination process of firms that engage differently in international activities.

We observe clear differences in behavior between firms that serve the foreign market either through exporting or through FDI. In particular, we find that being an exporter increases the likelihood of being characterized by imperfect competition in the product market. Exporting firms are more likely to share rents based on the bargaining power of workers, but less likely to share rents based on the elasticity of the labor supply curve facing an individual employer. As such, workers' bargaining power rather than search frictions seems to be important in generating wage dispersion across exporting firms.

Engagement in international activities not only matters for determining the type of imperfections in product and labor markets but also for determining the order of magnitude of these imperfections. In particular, we find that being an exporter introduces allocative inefficiencies in product as well as labor markets as export status appears to be positively correlated with both product market power (markups) and market power consolidated on the labor supply side (workers' bargaining power). Interestingly, export status is positively correlated with the wage elasticity of a firm's labor supply curve. This implies that exporting firms where search frictions are inducing wages to vary with revenue, are less able to exploit wage-setting power. Differential impacts are found for MNEs. Firms with foreign subsidiaries seem to reduce price distortions in product and labor markets. A negative correlation is observed between MNE status and markups, and between MNE status and workers' bargaining power.

Finally, acknowledging that firms are heterogeneous, our quantile regression results confirm that differences in product and labor market imperfections between exporters and non-exporters, and between MNEs and non-MNEs vary across firms. In addition, we observe heterogeneous returns to being an exporter/MNE within an industry and also discern cross-industry differences.

#### References

- [1] Abraham F, Konings J, Vanormelingen S. 2009. The effect of globalization on union bargaining and price-cost margins of firms. *Review of World Economics* **145(1)**: 13-36.
- [2] Ackerberg D, Caves K, Frazer G. 2015. Identification properties of recent production function estimators. *Econometrica* 83(6): 2411-2451.
- [3] Amiti M, Davis D. 2011. Trade, firms and wages: Theory and evidence. Review of Economic Studies 79(1): 1-36.
- [4] Ahsan RN, Mitra D. 2014. Trade liberalization and labor's slice of the pie. Journal of Development Economics 108(1): 1-16.
- [5] Bernard, AB, Jensen JB, Redding SJ, Schott PK. 2007. Firms in international trade. Journal of Economic Perspectives 21(3): 105-130.
- [6] Bernard AB, Jensen JB, Redding SJ, Schott PK. 2012. The empirics of firm heterogeneity and international trade. Annual Review of Economics 4(1): 283-313.
- [7] Boulhol H, Dobbelaere S, Maioli S. 2011. Imports as product and labour market discipline. British Journal of Industrial Relations 49(2): 331-361.
- [8] Brock E, Dobbelaere S. 2006. Has international trade affected workers' bargaining power? *Review of World Economics* 142(6): 233-266.
- Buchinsky M. 1994. Changes in the U.S. wage structure 1963-1987: Application of quantile regression. *Econometrica* 62(2): 405-458.
- [10] Coşar K, Guner N, Tybout J, 2016. Firm dynamics, job turnover, and wage distribution in an open economy. American Economic Review 106(3): 625-663.
- [11] Davis DR, Harrigan J. 2011. Good jobs, bad jobs and trade liberalization. Journal of International Economics 84(1): 26-36.
- [12] Davidson C, Heyman F, Matusz S, Sjöholm F, Zhu SC. 2014. Globalization and imperfect labor market sorting. *Journal of International Economics* 94(2): 177-194.
- [13] Davidson C, Matusz S, Shevchenko A. 2008. Globalization and firm-level adjustment with imperfect labor markets. *Journal of International Economics* 75(2): 295-309.
- [14] De Loecker J, Warzynski F. 2012. Markups and firm-level export status. American Economic Review 102(6): 2437-2471.
- [15] De Loecker J, Goldberg PK, Khandelwal AK, Pavcnik N. 2016. Prices, markups and trade reforms. *Econometrica* 84(2): 445-510.

- [16] Dobbelaere S, Mairesse J. 2013. Panel data estimates of the production function and product and labor market imperfections. *Journal of Applied Econometrics* **28(1)**: 1-46.
- [17] Dumont M, Rayp G, Willemé P. 2006. Does internationalization affect union bargaining power? An empirical study for five EU countries. Oxford Economic Papers 58(1): 77-102.
- [18] Egger H, Kreickemeier U. 2009. Firm heterogeneity and the labor market effects of trade liberalization. *International Economic Review* 50(1): 187-216.
- [19] Fajgelbaum P. 2013. Labor market frictions, firm growth and international trade. NBER Working Paper 19492. National Bureau of Economic Research.
- [20] Felbermayr G, Prat J, Schmerer HJ. 2011. Globalization and labor market outcomes: wage bargaining, search frictions, and firm heterogeneity. *Journal of Economic Theory* 146(1): 39-73.
- [21] Felbermayr G, Hauptman A, Schmerer HJ. 2014. International trade and collective bargaining outcomes: Evidence from German employer-employee data. *Scandinavian Journal of Economics* 116(3): 820-837.
- [22] Forlani E, Martin R, Mion G, and Muuls M. 2016. Unraveling firms: Demand, productivity and markups heterogeneity. CEPR Discussion Paper 11058. Centre for Economic Policy Research.
- [23] Fukao K, Hamagata S, Inui T, Ito K, Kwon HU, Makino T, Miyagawa T, Nakanishi Y, Tokui J. 2007. Estimation procedures and TFP analysis of the JIP database 2006 provisional version. RIETI Discussion Paper Series 07-E-003. Research Institute of Economy, Trade and Industry (RIETI).
- [24] Gandhi A, Navarro S, Rivers D. 2017. On the identification of gross output production functions. Unpublished.
- [25] Hall RE. 1988. The relationship between price and marginal cost in US industry. Journal of Political Economy 96(5): 921-947.
- [26] Hallak JC, Sivadasan J. 2009. Exporting behavior under quality constraints. NBER Working Paper 14928. National Bureau of Economic Research.
- [27] Harrison A, McLaren J, McMillan M. 2011. Recent perspectives on trade and inequality. Annual Review of Economics 3: 261-289.
- [28] Helpman E. 2006. Trade, FDI and the organization of firms. Journal of Economic Literature 44(3): 589-630.
- [29] Helpman E., Itskhoki O, Redding S. 2010. Inequality and unemployment in a global economy. Econometrica 78(4): 1239-1283.

- [30] Hopenhayn H. 1992. Entry, exit, and firm dynamics in long run equilibrium. *Econometrica* 60(5): 1127-1150.
- [31] Kato A. 2014. Does export yield productivity and markup premiums? Evidence from the Japanese manufacturing industry. RIETI Discussion Paper Series 14-E-037. Research Institute of Economy, Trade and Industry (RIETI).
- [32] Koenker RW, Bassett GJ. 1978. Regression quantiles. *Econometrica* 46(1): 33-50.
- [33] Koenker RW, d'Orey V. 1987. Algorithm AS 229: Computing regression quantiles. *Applied Statististics* **36(3)**: 383-393.
- [34] Konings J, Marcolin L, van Beveren I. 2016. International rent sharing and takeovers. International Journal of Manpower 37(2): 268-302.
- [35] Krugman P. 1980. Scale economies, product differentiation and the pattern of trade. American Economic Review 70(5):950-59.
- [36] Kugler M, Verhoogen E 2012. Plant size and product quality. *Review of Economic Studies* 79(1): 307-39.
- [37] Lu Y, Sugita Y, Lianming Zhu L. 2017. Wage markdowns and FDI liberalization. Unpublished.
- [38] Malchow-Møller N, Markusen J, Schkerning B. 2013. Foreign firms, domestic wages. Scandinavian Journal of Economics 115(2): 292-325.
- [39] Manning A. 2003. Monopsony in motion: Imperfect competition in labor markets. Princeton University Press: Princeton.
- [40] Mayer T, Ottaviano GIP. 2008. The happy few: New facts on the internationalisation of European firms. *Intereconomics* 43(3): 135-148.
- [41] McDonald IM, Solow RM. 1981. Wage bargaining and employment. American Economic Review 71(5): 896-908.
- [42] Melitz M. 2003. The impact of trade on intra-industry reallocations and aggregate industry productivity. *Econometrica* 71(6): 1695-725.
- [43] Melitz M, Levinsohn JA. 2006. Productivity in a differentiated products market equilibrium. Unpublished.
- [44] Melitz MJ, Ottaviano GIP. 2008. Market size, trade and productivity. Review of Economic Studies 75(1): 295-316.
- [45] Montagna C, Nocco A. 2013. Unionization, international integration and selection. Canadian Journal of Economics 46(1): 23-45.

- [46] Nickell SJ, Andrews M. 1983. Unions, real wages and employment in Britain 1951-79. Oxford Economic Papers 35(supplement): 183-205.
- [47] Rodrik D. 1997. Has globalization gone too far? Washington, DC.: Institute for International Economics.
- [48] Senses MZ. 2010. The effects of offshoring on the elasticity of labor demand. Journal of International Economics 81(1): 89-98.
- [49] Slaughter M. 2001. International trade and labor-demand elasticities. Journal of International Economics 54(1): 27-56.
- [50] Temouri Y, Driffield NL, Higon DA. 2008. Analysis of productivity differentials among foreign and domestic firms: Evidence from Germany. *Review of World Economics* **144(1)**: 32-54.
- [51] World Trade Organization. 2008. World trade report: Trade in a globalizing world. Geneva: World Trade Organization.
- [52] Wooldridge JM. 2002. Econometric analysis of cross sections and panel data. MIT Press: Cambridge, MA.

Jotal         Mean         Sol.         Q1         Q2         Q3         N           Real fram output growth $\Delta q_{14}$ -0.007         -0.001         0.003         64,481           Labor growth rate $\Delta m_{14}$ -0.012         0.111         -0.008         0.0028         64,481           Material growth rate $\Delta m_{14}$ -0.012         0.113         -0.015         0.028         64,481           Labor share in nominal sales $s_{N(4)}$ 0.522         0.168         0.477         0.004         0.025         64,481           Average wage $W_a$ 5.224         1.000         4.061         5.031         6.133         3.01         64,481           Average wage $W_a$ 5.224         1.000         4.061         5.031         6.133         3.01         64,481           Average wage $W_a$ 0.226         0.418         0         0         1         64,481           Average wage $W_a$ 0.226         0.418         0         0         1         64,481           Mace of mon-production workers         0.348         0.331         0.067         0.103         6.4481           Share of non-production workers         0.348         0.033         0.066         0.079 <th></th> <th>2.6</th> <th></th> <th></th> <th></th> <th></th> <th>37</th>		2.6					37
Ideal from output growth $\Delta q_{tt}$ -0.001       0.003       -0.001       0.003       64,81         Material growth rate $\Delta k_{tt}$ -0.012       0.141       -0.085       -0.009       0.062       64,481         Capital growth rate $\Delta k_{tt}$ -0.012       0.141       -0.085       0.036       0.185       0.226       64,481         Material share in nominal soles $s_{Mit}$ 0.197       0.086       0.136       0.185       0.216       64,481         Average wage $W_{it}$ 5.224       1.690       4.061       5.031       6.183       64,481         Exporter dummy EXP <sub>H</sub> 0.310       0.463       0       0       1       64,481         Exporter dummy IMP <sub>d</sub> 0.220       0.414       0       1       64,481         Exporter dummy IMP <sub>d</sub> 0.220       0.418       0       0       0.64,481         Exporter straio       0.032       0.097       0.000       0.006       64,481         Exporter straio       0.032       0.097       0.000       0.006       64,481         Exporter straio       0.032       0.097       0.000       0.000       0.066       44,481         Share of non-productivin       32580       23.209 <td></td> <td>Mean</td> <td>Sd.</td> <td><math>Q_1</math></td> <td><math>Q_2</math></td> <td><math>Q_3</math></td> <td>N</td>		Mean	Sd.	$Q_1$	$Q_2$	$Q_3$	N
Labor growth rate $\Delta m_{tl}$ = -0.010 0.081 = -0.048 = -0.008 0.028 6-4.881 Capital growth rate $\Delta m_{tl}$ = -0.043 0.132 0.105 = -0.088 = -0.024 64,481 Labor share in nominal sales $s_{Mt}$ 0.197 0.086 0.136 0.185 0.246 64,481 1 = $s_{Mt} = s_{Mt}$ 0.221 0.135 0.119 0.194 0.255 64,481 Average wage $W_{tl}$ 5.224 1.690 4.061 5.031 6.138 64,481 Average wage $W_{tl}$ 5.224 1.690 4.061 5.031 6.1481 Age = 45 18 34 46 56 64,481 Exporter dummy LNP <sub>tl</sub> 0.210 0.443 0 0 1 1 64,481 MNE dummy MNE <sub>tl</sub> 0.220 0.444 0 0 0 1 1 64,481 MNE dummy MNE <sub>tl</sub> 0.220 0.444 0 0 0 64,481 Exporter dummy MNE <sub>tl</sub> 0.220 0.448 0 0 0 64,481 Capital intensity 0.220 0.444 0 0 0 64,481 Labor productivity 32.80 2.3320 18.742 2.6405 30.049 64,481 Labor productivity 32.580 2.3320 18.742 2.6405 30.049 64,481 Labor growth rate $\Delta m_{tl}$ 0.001 0.136 0.066 0.006 0.074 19.998 Labor growth rate $\Delta m_{tl}$ 0.001 0.136 0.066 0.070 19.998 Labor growth rate $\Delta m_{tl}$ 0.001 0.133 0.016 0.028 19.998 Labor share in nominal sales $s_{Mt}$ 0.196 0.613 0.706 19.998 Labor share in nominal sales $s_{Mt}$ 0.196 0.081 0.137 0.184 0.244 19.998 Labor share in nominal sales $s_{Mt}$ 0.196 0.031 0.137 0.184 0.244 19.998 Labor share in nominal sales $s_{Mt}$ 0.197 0.028 0.017 19.998 Labor share in nominal sales $s_{Mt}$ 0.197 0.028 0.017 19.998 Labor share in nominal sales $s_{Mt}$ 0.197 0.028 0.017 19.998 Labor share in nominal sales $s_{Mt}$ 0.197 0.028 0.014 0.133 0.106 0.1998 Auterial share in nominal sales $s_{Mt}$ 0.197 0.014 0.134 0.1394 19.998 Labor share in nominal sales $s_{Mt}$ 0.016 0.015 0.110 0.0044 0.134 19.998 Labor share in nominal sales $s_{Mt}$ 0.016 0.015 0.110 0.0043 0.134 19.998 Labor share in nominal sales $s_{Mt}$ 0.027 0.01 0.107 0.014 0.037 19.998 Labor share in nominal sales	Real firm output growth $\Delta q_{it}$	-0.004	0.130	-0.070	-0.001	0.063	64,481
$\begin{aligned} \begin{array}{llllllllllllllllllllllllllllllllllll$	Labor growth rate $\Delta n_{it}$	-0.010	0.081	-0.045	-0.008	0.028	64,481
Capital growth rate $\Delta \kappa_{it}$ = 0.013 0.132 0.108 0.038 0.0058 0.405 04.481 Material share in nominal sales $s_{Mit}$ 0.582 0.168 0.477 0.601 0.706 64.481 $1 - s_{Mi} - s_{Mi}$ 0.221 0.135 0.119 0.194 0.295 64.481 Average wage $W_a$ 5.224 1.600 4.001 5.031 6.183 04.481 Number of workers $N_{it}$ 433 1.849 85 143 301 64.481 Exporter dummy EXP_{it} 0.310 0.463 0 0 1 64.481 Importer dummy ME <sub>it</sub> 0.226 0.418 0 0 0 1 64.481 Exporter dummy ME <sub>it</sub> 0.226 0.418 0 0 0 0 64.481 Capital intensity 1.617 0.853 1.131 1.657 2.164 64.481 Share of non-production workers 0.348 0.239 0.167 0.299 0.479 64.481 Labor productivity 32.580 2.3320 1.677 0.299 0.479 64.481 Labor productivity 32.580 2.3320 1.677 0.299 0.479 64.481 Labor productivity 32.580 0.332 0.006 0.006 0.006 0.474 19.998 Material growth rate $\Delta m_{it}$ 0.001 0.136 0.006 0.006 0.074 19.998 Material growth rate $\Delta m_{it}$ 0.007 0.077 0.039 0.000 0.000 64.481 Labor growth rate $\Delta m_{it}$ 0.001 0.136 0.006 0.006 0.074 19.998 Material growth rate $\Delta m_{it}$ 0.001 0.136 0.006 0.006 0.074 19.998 Material growth rate $\Delta m_{it}$ 0.001 0.138 0.137 0.168 0.281 9.998 Material growth rate $\Delta m_{it}$ 0.001 0.138 0.137 0.168 0.281 9.998 Material growth rate $\Delta m_{it}$ 0.001 0.138 0.137 0.168 0.284 19.998 Material growth rate $\Delta m_{it}$ 0.014 0.133 0.106 0.018 0.262 0.1998 Material growth rate $\Delta m_{it}$ 0.014 0.133 0.106 0.018 0.137 0.181 0.244 19.998 Material growth rate $\Delta m_{it}$ 0.012 0.157 0.496 0.613 0.706 19.998 Material growth rate $\Delta m_{it}$ 0.010 0.107 0.177 0.429 0.131 0.299 19.998 Material share in nominal sales $s_{Mit}$ 0.522 0.157 0.496 0.613 0.706 19.998 Material growth rate $\Delta m_{it}$ 0.522 0.157 0.496 0.613 0.706 19.998 Material growth rate $\Delta m_{it}$ 0.552 0.497 0 1 1 1 9.998 Material growth rate $\Delta m_{it}$ 0.105 0.151 0.000 0.003 1 1 1 9.998 Material share in nominal sales $s_{Mit}$ 0.107 0.11 0.004 0.43 0.134 19.998 Material share in nominal sales $s_{Mit}$ 0.107 0.11 0.1004 0.043 0.344 4.433 Material growth rate $\Delta m_{it}$ 0.007	Material growth rate $\Delta m_{it}$	-0.012	0.141	-0.085	-0.009	0.062	64,481
Labor share in nominal sales $s_{Mit}$ (0.197) (0.086) (0.183) (0.185) (0.240) (64,481) $1 - s_{Nit} - s_{Mit}$ (0.221) (0.133) (0.119) (0.194) (0.225) (64,481) $1 - s_{Nit} - s_{Mit}$ (0.221) (0.133) (0.119) (0.194) (0.225) (64,481) Average wage $W_{it}$ (5.224) (1.690) (4.661) (5.031) (6.183) (64,481) Age (4.61) (0.000) (0.01) (0.463) (0.00) (0.01) (1.64,481) Importer dummy IMP_{it} (0.270) (0.444) (0.00) (1.64,481) MNE dummy MNE <sub>it</sub> (0.226) (0.418) (0.00) (0.000) (0.000) (4.481) (Capital intensity) (1.617) (0.853) (1.131) (1.667) (2.106) (64,481) Capital intensity (1.617) (0.853) (1.131) (1.667) (2.106) (64,481) Share of non-production workers (0.348) (0.232) (0.167) (0.299) (0.479) (64,481) Share of non-production workers (0.348) (0.232) (0.167) (0.299) (0.479) (64,481) Share of non-production workers (0.348) (0.232) (0.167) (0.299) (0.479) (64,481) Share of non-production workers (0.348) (0.232) (0.167) (0.299) (0.479) (64,481) Share of non-production workers (0.348) (0.232) (0.167) (0.299) (0.479) (64,481) Share of non-production workers (0.348) (0.37) (0.006) (0.074) (1.9989) Labor productivity (32,2580) (2.3320) (18,742) (2.610) (0.006) (0.074) (1.9989) Labor growth rate $\Delta n_{it}$ (0.001) (0.136) (0.006) (0.006) (0.074) (1.9989) Labor growth rate $\Delta n_{it}$ (0.001) (0.136) (0.006) (0.006) (0.77) (1.9989) Material growth rate $\Delta n_{it}$ (0.014) (0.133) (0.184) (0.244) (1.9989) Material share in nominal sales $s_{Nit}$ (0.522) (0.157) (0.466) (0.153) (0.100) (0.13) (1.9989) Material share in nominal sales $s_{Nit}$ (0.522) (0.157) (0.466) (0.153) (0.100) (1.103) (1.9998) Material share in nominal sales $s_{Nit}$ (0.522) (0.157) (0.466) (0.153) (0.44) (1.9989) Material share in nominal sales $s_{Nit}$ (0.522) (0.167) (0.460) (0.13) (1.9998) Material share in nominal sales $s_{Nit}$ (0.522) (0.477) (0.11) (1.11) (1.9998) Material share in nominal sales $s_{Nit}$ (0.522) (0.471) (0.11) (0.131) (1.9998) Material share in nominal sales $s_{Nit}$ (0.477) (0.471) (0.044) (0.438) (0.44	Capital growth rate $\Delta k_{it}$	-0.043	0.132	-0.105	-0.088	-0.052	64,481
$\begin{aligned} \begin{array}{llllllllllllllllllllllllllllllllllll$	Labor share in nominal sales $s_{Nit}$	0.197	0.080	0.130	0.185	0.240	04,481
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Material share in nominal sales $s_{Mit}$	0.982	0.108	0.477	0.004	0.700	04,481 64 481
Average wage $W_{it}$ 5.2241.0004.0015.0310.18304,81Age451834465664,481Exporter dummy EXP_{it}0.3100.46300164,181Importer dummy LNP_{it}0.2200.44400064,481Export-sales ratio0.0320.0070.0000.00064,481Capital intensity1.6170.8531.1311.6572.16464,481Capital intensity1.6170.8531.1311.6572.16464,481Share of non-production workers0.3180.2390.1670.2990.17964,481ExportersMeanSt. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ 0.0010.136-0.0660.0060.07419,998Labor growth rate $\Delta n_{it}$ -0.0070.177-0.039-0.0050.02819,998Material growth rate $\Delta h_{it}$ -0.00410.133-0.166-0.088-0.01719,998Labor short into $\Delta n_{it}$ -0.1660.0180.1370.1840.24419,998Material share in nominal sales $s_{Nit}$ 0.5920.1570.4960.6130.70619,998Labor short in $\Delta n_{it}$ 0.0100.00011119,998Material share in nominal sales $s_{Nit}$ 0.5120.4790111Number of workers $N_{it}$ 8733,148126	$1 - S_{Nit} - S_{Mit}$	5.004	0.155	4.061	5 021	0.290	04,401 64 491
	Average wage $W_{it}$	J.224 422	1.090	4.001	143	201	64,401 64,481
AgeTBTBOFTDOFTDOF </td <td>A go</td> <td>455</td> <td>1,049</td> <td>34</td> <td>140</td> <td>56</td> <td>64 481</td>	A go	455	1,049	34	140	56	64 481
Laboric0.4440.0010.445Importer dummy IMP $_{tl}$ 0.2700.44400164,481MNE dummy MNE $_{tl}$ 0.2260.418000.00664,481Export-sales ratio0.0320.0970.0000.00664,481Capital intensity1.6170.8531.1311.6572.16464,481Share of non-production workers0.3480.2391.874226.40539.04964,481ExportersMeanSd.Q1Q2Q3NReal firm output growth $\Delta q_{tt}$ 0.0010.013-0.0660.0060.07419.998Labor growth rate $\Delta m_{tt}$ -0.0070.017-0.038-0.0050.02819.998Material growth rate $\Delta m_{tt}$ -0.0010.133-0.106-0.088-0.01719.998Labor growth rate $\Delta m_{tt}$ -0.0100.0130.1840.21419.998Material share in nominal sales $s_{Mit}$ 0.2120.1280.1180.1850.28019.998Labor share in nominal sales $s_{Mit}$ 0.2120.1280.1180.1850.26019.998Labor share in nominal sales $s_{Mit}$ 0.2120.1280.1180.1850.26019.998Labor share in nominal sales $s_{Mit}$ 0.2120.1280.1180.1350.26019.998Labor share in nominal sales $s_{Mit}$ 0.2120.1280.11119.998Muther of workers $N_{tt}$ 5.	Exportor dummy EXP.	40	0.463	04	40	1	64,401
	Importer dummy IMP	0.310	0.403	0	0	1	64,401 64,481
	$\frac{1}{10000000000000000000000000000000000$	0.270	0.444	0	0	1	64,401 64 481
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Export sales ratio	0.220	0.410	0 000	0 000	0 006	64 481
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Horfindahl Hirschman Indox HHI	0.032	0.097	0.000	0.000	0.000	64,401 64,481
Capital intensity1.010.0301.1010.0032.1040.47964,481Labor productivity32.58023.32018.74226.40539.04964,481SRate0.0140.073-0.0210.0160.05264,481ExportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ 0.0010.0160.0060.0050.02819.998Labor growth rate $\Delta m_{it}$ -0.0070.077-0.039-0.0050.02819.998Capital growth rate $\Delta m_{it}$ -0.0410.133-0.106-0.088-0.07119.998Capital growth rate $\Delta k_{it}$ -0.0410.133-0.166-0.088-0.07119.998Material share in nominal sales $s_{Mit}$ 0.1210.1280.1180.1350.28019.998Average wage $W_{it}$ 5.9151.6674.7265.7746.94019.998Average wage $W_{it}$ 5.9151.6674.7265.7746.94019.998Age491839516019.998MxE dummy MNE_{it}0.5520.49701119.998Importer dummy IMP_{it}0.6450.47901119.998Share of no-production workers0.3980.2280.2180.3040.51419.998Share of no-production workers0.3980.2280.2180.3040.54719.998Share of no-productivity36.273<	$C_{apital}$ intensity	1.617	0.045	1 1 2 1	1.657	0.105	64 481
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Share of non-production workers	0.348	0.855	0.167	0.200	2.104 0.470	64,401 64,481
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Labor productivity	32 580	0.209 23 320	18 749	0.299 26.405	30.040	64 481
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	SR.	0.014	0.073	0.021	20.405	0.059	64,401 64,481
ExportedIntealJoin $Q_1$ $Q_2$ $Q_2$ $Q_3$ $P_1$ Real firm output growth $\Delta q_{it}$ 0.0010.136-0.0660.0060.07419.998Labor growth rate $\Delta m_{it}$ -0.0090.145-0.084-0.0050.02819.998Capital growth rate $\Delta m_{it}$ -0.0190.133-0.106-0.088-0.04719.998Labor share in nominal sales $s_{Mit}$ 0.5920.1570.4660.6130.70619.9981 - $s_{Nit} - s_{Mit}$ 0.2120.1280.1180.1850.28019.998Average wage $W_{it}$ 5.9151.6674.7265.7746.94019.998Number of workers $N_{it}$ 8733.14812625362619.998Age491839516019.998Importer dummy EXP_{it}1.0000.0001119.998Exporter dummy IMP_{it}0.6450.47901119.998Herfindahl-Hirschman Index HHI Jt0.0720.0430.0280.07419.998Share of non-production workers0.3980.2280.2180.3640.54719.998Iabor growth rate $\Delta m_{it}$ 0.0110.0830.0240.01319.998Share of non-production workers0.3980.2280.2180.3640.54719.998Iabor growth rate $\Delta m_{it}$ -0.0070.127-0.0200.0190.05719.998Share of non-production workers <td>Exportors</td> <td>Moan</td> <td>0.075 Sd</td> <td>-0.021</td> <td>0.010</td> <td>0.052</td> <td>04,401 N</td>	Exportors	Moan	0.075 Sd	-0.021	0.010	0.052	04,401 N
Team Hill output growth $\Delta q_{ti}$ 0.0010.0130.0000.00050.000419,998Labor growth rate $\Delta n_{ti}$ -0.0090.014-0.039-0.0050.02819,998Capital growth rate $\Delta k_{it}$ -0.0410.133-0.106-0.088-0.04719,998Labor share in nominal sales $s_{Nit}$ 0.1960.0810.1370.1840.24419,998Material share in nominal sales $s_{Nit}$ 0.2120.1280.1370.1840.24419,998Average wage $W_{it}$ 5.9151.6674.7265.7746.94019,998Number of workers $N_{it}$ 8733,14812625362619,998Average wage $W_{it}$ 1.0000.0001119,998Exporter dummy EXP_{it}1.0000.0001119,998MNE dummy MNE_{it}0.6450.479011Bexport-sales ratio0.1050.1510.0100.0430.134Capital intensity1.7330.7981.2471.7662.259Share of non-production workers0.3980.2280.2180.3640.547Labor growth rate $\Delta n_{it}$ -0.0110.0830.0180.0480.057Share of non-productivity36.27322.79521.82430.41744.390Labor growth rate $\Delta n_{it}$ -0.0140.0380.048-0.05644.483Labor growth rate $\Delta n_{it}$ -0.0140.0380.048-0.05644.483 <td>Pool from output growth A g</td> <td>0.001</td> <td>0.126</td> <td>0.066</td> <td>0.006</td> <td>0.074</td> <td>10.008</td>	Pool from output growth A g	0.001	0.126	0.066	0.006	0.074	10.008
Labor grown rate Δ $h_{it}$ -0.007-0.007-0.0087-0.005-0.002819,998Material growth rate $\Delta k_{it}$ -0.0410.133-0.106-0.088-0.04719,998Labor share in nominal sales $s_{Nit}$ 0.1960.0810.1370.1840.24419,998Material share in nominal sales $s_{Nit}$ 0.5920.1570.4960.6130.70619,998Average wage $W_{it}$ 5.9151.6674.7265.7746.94019,998Average wage $W_{it}$ 5.9151.6674.7265.7746.94019,998Age491839516019,998Exporter dummy EXP <sub>it</sub> 1.0000.00011119,998MNE dummy MNE <sub>it</sub> 0.5520.49701119,998Capital intensity1.7330.7981.2471.7662.25919,998Capital intensity1.7330.7981.2471.7662.25919,998Babor productivity36.27322.79521.82430.41744.30019,998Labor growth rate $\Delta n_{it}$ -0.0070.127-0.071-0.0040.05844,483Labor growth rate $\Delta n_{it}$ -0.0110.083-0.048-0.0090.02844,483Labor growth rate $\Delta n_{it}$ -0.0140.131-0.104-0.088-0.05644,483Labor growth rate $\Delta n_{it}$ -0.0140.131-0.048-0.05644,483Labor growth rate $\Delta$	Lehon growth rate $\Delta r$	0.001	0.130	-0.000	0.000	0.074	19,990
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Labor growth rate $\Delta n_{it}$	-0.007	0.077	-0.039	-0.005	0.028	19,990
Capital growth rate $\Delta k_{it}$ -0.011-0.013-0.003-0.003-0.01410.33Labor share in nominal sales $s_{Nit}$ 0.1900.0810.1370.1840.24419,998Material share in nominal sales $s_{Mit}$ 0.5920.1570.4960.6130.70619,998Average wage $W_{it}$ 5.9151.6674.7265.7746.94019,998Average wage $W_{it}$ 5.9151.6674.7265.7746.94019,998Age491839516019,998Exporter dummy EXP_{it}1.0000.00011119,998MNE dummy MNE <sub>it</sub> 0.6450.47901119,998Export-sales ratio0.1050.1510.0100.0430.13419,998Capital intensity1.7330.7981.2271.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998SR <sub>it</sub> 0.0160.075-0.0200.0190.05719,998SR <sub>it</sub> 0.0160.075-0.0200.0190.05719,998SR <sub>it</sub> 0.0160.075-0.0200.0190.05444,483Labor productivity36.27322.79521.82430,41744.39019,998SR <sub>it</sub> 0.0160.075-0.0200.0190.05719,998SR <sub>it</sub> 0.0160.075-0.0200.0190.05944,483 <tr< td=""><td>Material growth rate <math>\Delta m_{it}</math></td><td>-0.009</td><td>0.140 0.122</td><td>-0.084</td><td>-0.005</td><td>0.070</td><td>19,990</td></tr<>	Material growth rate $\Delta m_{it}$	-0.009	0.140 0.122	-0.084	-0.005	0.070	19,990
Labor share in nominal sales $s_{Nit}$ 0.1500.1510.1510.1240.1240.1593Material share in nominal sales $s_{Nit}$ 0.5920.1570.4960.6130.70619,998 $1 - s_{Nit} - s_{Mit}$ 5.9151.6674.7265.7746.94019,998Number of workers $N_{it}$ 8733,14812625362619,998Age491839516019,998Exporter dummy EXP_{it}1.0000.00011119,998Importer dummy MNE <sub>it</sub> 0.5520.49701119,998Export-sales ratio0.1050.1510.0100.0430.13419,998Capital intensity1.7330.7981.2471.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998SR <sub>it</sub> 0.0160.075-0.0020.0190.05719,998Share of non-productivity36.27322.79521.82430.41744.39019,998SR <sub>it</sub> 0.0160.075-0.0200.0190.05719,998Share of non-productivity36.27322.79521.82430.41744.39019,998Share of non-productivity36.27322.79521.82430.41744.39019,998Share of nom-productivity36.27322.79521.82430.41744.39019,998Share of nom-productivity36.27322.	Labor share in pominal sales $a_{it}$	-0.041	0.133	-0.100	-0.088	-0.047	10,008
Material state in nominal sales $S_{Mit}$ 0.3520.3130.3330.0130.01019,9981 - $s_{Nit} - s_{Mit}$ 0.2120.2120.1280.1180.1530.28019,998Average wage $W_{it}$ 5.9151.6674.7265.7746.94019,998Age491839516019,998Age491839516019,998Exporter dummy EXP_{it}1.0000.00011119,998Importer dummy MNE <sub>it</sub> 0.5520.49701119,998Export-sales ratio0.1050.1510.0100.0430.13419,998Capital intensity1.7330.7981.2471.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998Labor productivity36.27322.79521.82430.41744.39019,998SR_{it}0.0160.075-0.0200.0190.05844,483Labor growth rate $\Delta n_{it}$ -0.0070.127-0.0040.05844,483Material growth rate $\Delta n_{it}$ -0.0110.083-0.048-0.0090.02844,483Labor share in nominal sales $s_{Nit}$ 0.1780.1720.4680.5990.70644,483Labor share in nominal sales $s_{Nit}$ 0.2366117712021844,483Murber of workers $N_{it}$ 236611771	Labor share in nominal sales $s_{Nit}$	0.190	0.031	0.137	0.104	0.244	10,008
1 - $S_{Nit} - S_{Nit}$ 0.1120.1130.1130.1200.20013,393Average wage $W_{it}$ 5.9151.6674.7265.7746.94019,998Number of workers $N_{it}$ 8733,14812625362619,998Age491839516019,998Exporter dummy EXP_{it}1.0000.00011119,998MNE dummy MNE_{it}0.5520.49701119,998Export-sales ratio0.1050.1510.0100.0430.13419,998Capital intensity1.7330.7981.2471.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998SR_{it}0.0160.075-0.0200.0190.05719,998Non-exportersMeanSd.Q1Q2Q3NReal firm output growth $\Delta q_{it}$ -0.0110.083-0.048-0.0090.02844,483Labor growth rate $\Delta n_{it}$ -0.0140.139-0.086-0.0110.05944,483Labor share in nominal sales $s_{Nit}$ 0.1980.0280.1350.1850.24844,483Material growth rate $\Delta k_{it}$ -0.0190.0260.0180.03044,483Labor share in nominal sales $s_{Nit}$ 0.1980.0280.1550.1720.4680.5990.70644,483Labor share in nominal sales $s_{Nit}$ 0.225 </td <td>Material share in nominal sales <math>s_{Mit}</math></td> <td>0.392</td> <td>0.137</td> <td>0.490</td> <td>0.013</td> <td>0.700</td> <td>19,998</td>	Material share in nominal sales $s_{Mit}$	0.392	0.137	0.490	0.013	0.700	19,998
Number of wage $w_{it}$ $3.13$ $1.001$ $4.120$ $5.142$ $5.340$ $15,330$ Number of wage $N_{it}$ $873$ $3,148$ $126$ $253$ $626$ $19,998$ Age $49$ $18$ $39$ $51$ $60$ $19,998$ Exporter dummy EXP <sub>it</sub> $1.000$ $0.000$ $11$ $11$ $19,998$ Importer dummy MNE <sub>it</sub> $0.645$ $0.479$ $0$ $11$ $11$ $19,998$ Export-sales ratio $0.105$ $0.151$ $0.010$ $0.043$ $0.134$ $19,998$ Capital intensity $1.733$ $0.798$ $1.247$ $1.766$ $2.259$ $19,998$ Share of non-production workers $0.398$ $0.228$ $0.218$ $0.364$ $0.547$ $19,998$ Labor productivity $36.273$ $22.795$ $21.824$ $30.417$ $44.300$ $19,998$ SR <sub>it</sub> $0.016$ $0.075$ $-0.020$ $0.019$ $0.057$ $19,998$ Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ $-0.007$ $0.127$ $-0.071$ $-0.004$ $0.058$ $44,483$ Labor growth rate $\Delta m_{it}$ $-0.014$ $0.139$ $-0.086$ $-0.011$ $0.058$ $44,483$ Labor share in nominal sales $s_{Nit}$ $0.198$ $0.0166$ $0.135$ $0.185$ $0.248$ $44,483$ Material share in nominal sales $s_{Nit}$ $0.198$ $0.172$ $0.468$ $0.599$ $0.706$ $44,483$ Arear ge wage $W_{it}$ <td< td=""><td><math>1 - s_{Nit} - s_{Mit}</math></td><td>0.212 5.015</td><td>1.667</td><td>0.110 4 796</td><td>5.774</td><td>6.040</td><td>19,998</td></td<>	$1 - s_{Nit} - s_{Mit}$	0.212 5.015	1.667	0.110 4 796	5.774	6.040	19,998
Age $373$ $3743$ $120$ $223$ $620$ $12,333$ Age491839516019,998Exporter dummy EXP $_{it}$ 1.0000.00011119,998Importer dummy IMP $_{it}$ 0.6450.47901119,998Export-sales ratio0.1050.1510.0100.0430.13419,998Capital intensity1.7330.7981.2471.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998Labor productivity36.27322.79521.82430.41744.39019,998S $R_{it}$ 0.0160.075-0.0200.0190.05719,998Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ -0.0070.127-0.071-0.0040.05844,483Labor growth rate $\Delta m_{it}$ -0.0140.133-0.048-0.0090.02844,483Labor share in nominal sales $s_{Nit}$ 0.1980.0880.1350.1850.24844,483Labor share in nominal sales $s_{Nit}$ 0.1980.000000044,483Labor share in nominal sales $s_{Nit}$ 0.1980.0163.8444.7325.77344,483Material share in nominal sales $s_{Nit}$ 0.2250.1380.1200.1980.30344,483Mure of workers $N_{it}$ 236	Number of workers $N_{it}$	873	3 1/8	4.720	953	626	10,008
Age491006.00111119,998Exporter dummy IMP_{it}0.6450.47901119,998MNE dummy MNE_{it}0.5520.49701119,998Export-sales ratio0.1050.1510.0100.0430.13419,998Capital intensity1.7330.7981.2471.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998Labor productivity36.27322.79521.82430.41744.39019,998SR_{it}0.0160.075-0.0200.0190.05719,998Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ -0.0110.083-0.048-0.0090.02844,483Labor growth rate $\Delta n_{it}$ -0.0140.131-0.104-0.088-0.0110.05944,483Labor growth rate $\Delta n_{it}$ -0.0140.131-0.104-0.088-0.05644,483Labor share in nominal sales $s_{Nit}$ 0.5780.1720.4680.5990.70644,483Labor share in nominal sales $s_{Nit}$ 0.2250.1380.1200.1980.30344,483Material share in nominal sales $s_{Nit}$ 0.2366117712021844,483Mumber of workers $N_{it}$ 2366117712021844,483Importer dummy	A go	40	18	30	200 51	60	10,008
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Exportor dummy EXP.	1 000	0.000	1	1	1	10.008
Importer dummy MNE $it$ 0.0150.0171119,998MNE dummy MNE $it$ 0.5520.49701119,998Export-sales ratio0.1050.0150.0100.0430.13419,998Herfindahl-Hirschman Index HHI $J_t$ 0.0720.0430.0280.0740.10319,998Capital intensity1.7330.7981.2471.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998Labor productivity36.27322.79521.82430.41744.39019,998S $R_{it}$ 0.0160.075-0.0200.0190.05719,998Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ -0.0070.127-0.071-0.0040.05844,483Labor growth rate $\Delta m_{it}$ -0.0140.139-0.086-0.0110.05944,483Capital growth rate $\Delta k_{it}$ -0.0440.131-0.104-0.088-0.05644,483Labor share in nominal sales $s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Inporter dummy $IMP_{it}$ 0.0010.000000044,483Inporter dummy $MNE_{it}$ <	Importer dummy IMP	0.645	0.000 0.479	0	1	1	19,998
Introdummy Nitrelit0.0020.1010.0110.0430.13419,998Export-sales ratio0.1050.1510.0100.0430.13419,998Capital intensity1.7330.7981.2471.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998Labor productivity36.27322.79521.82430.41744.39019,998SRit0.0160.075-0.0200.0190.05719,998Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ -0.0070.127-0.071-0.0040.05844,483Labor growth rate $\Delta n_{it}$ -0.0110.083-0.048-0.0090.02844,483Material growth rate $\Delta k_{it}$ -0.0140.139-0.086-0.0110.05944,483Labor share in nominal sales $s_{Nit}$ 0.1980.1720.4680.5990.70644,4831 - $s_{Nit} - s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ 2366117712021844,483Material share in nominal sales $s_{Mit}$ 0.30100044,483Murge wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ <td>MNE dummy MNE</td> <td>0.519</td> <td>0.497</td> <td>0</td> <td>1</td> <td>1</td> <td>19 998</td>	MNE dummy MNE	0.519	0.497	0	1	1	19 998
Laple bases0.1030.1030.1040.0740.0740.0740.0740.10319,998Capital intensity1.7330.7981.2471.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998Labor productivity36.27322.79521.82430.41744.39019,998 $SR_{it}$ 0.0160.075-0.0200.0190.05719,998Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ -0.0070.127-0.071-0.0040.05844,483Labor growth rate $\Delta n_{it}$ -0.0110.083-0.048-0.0090.02844,483Capital growth rate $\Delta k_{it}$ -0.0140.131-0.104-0.088-0.05644,483Labor share in nominal sales $s_{Nit}$ 0.5780.1720.4680.5990.70644,483I - $s_{Nit} - s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ 2366117712021844,483MNE dummy $MNE_{it}$ 0.0000.00000044,483Mose of workers $N_{it}$ 2366117712021844,483Murber of workers $N_{it}$ 0.0790.27000044,483Murber o	Export-sales ratio	0.002	0.151	0.010	0.043	0 134	19 998
Capital intensity1.7330.7981.2471.7662.25919,998Share of non-production workers0.3980.2280.2180.3640.54719,998Labor productivity36.27322.79521.82430.41744.39019,998 $SR_{it}$ 0.0160.075-0.0200.0190.05719,998Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ -0.0070.127-0.071-0.0040.05844,483Labor growth rate $\Delta n_{it}$ -0.0110.083-0.048-0.0090.02844,483Capital growth rate $\Delta m_{it}$ -0.0140.139-0.086-0.0110.05944,483Labor share in nominal sales $s_{Nit}$ 0.1980.0880.1350.1850.24844,483Material share in nominal sales $s_{Nit}$ 0.2250.1380.1200.1980.30344,483 $1 - s_{Nit} - s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Muber of workers $N_{it}$ 2366117712021844,483MyMy0.0000.00000044,483Muber of workers $N_{it}$ 0.3010.00000044,483Muber of workers $N_{it}$ 0.2666117712021844,483MyMy0.000	Herfindahl-Hirschman Index HHI 14	0.100	0.043	0.028	0.074	0.101	19 998
Share of non-production workers0.3980.2280.2180.3640.54719,998Labor productivity36.27322.79521.82430.41744.39019,998 $SR_{it}$ 0.0160.075-0.0200.0190.05719,998Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ -0.0070.127-0.071-0.0040.05844,483Labor growth rate $\Delta n_{it}$ -0.0110.083-0.048-0.0090.02844,483Material growth rate $\Delta m_{it}$ -0.0140.139-0.086-0.0110.05944,483Capital growth rate $\Delta k_{it}$ -0.0440.131-0.104-0.088-0.05644,483Labor share in nominal sales $s_{Nit}$ 0.1980.0880.1350.1850.24844,483Material share in nominal sales $s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ 2366117712021844,483Exporter dummy $EXP_{it}$ 0.0000.00000044,483Importer dummy $MNE_{it}$ 0.0790.27000044,483MNE dummy $MNE_{it}$ 0.0690.0430.0310.0630.10444,483Share of non-production workers0.3250.2400.1460.2700.44244,483 </td <td>Capital intensity</td> <td>1.733</td> <td>0.798</td> <td>1.247</td> <td>1 766</td> <td>2.259</td> <td>19 998</td>	Capital intensity	1.733	0.798	1.247	1 766	2.259	19 998
Labor production without $36.273$ $22.795$ $21.824$ $30.417$ $44.390$ $19,998$ $SR_{it}$ $0.016$ $0.075$ $-0.020$ $0.019$ $0.057$ $19,998$ Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ $N$ Real firm output growth $\Delta q_{it}$ $-0.007$ $0.127$ $-0.071$ $-0.004$ $0.058$ $44,483$ Labor growth rate $\Delta n_{it}$ $-0.011$ $0.083$ $-0.048$ $-0.009$ $0.028$ $44,483$ Material growth rate $\Delta m_{it}$ $-0.014$ $0.139$ $-0.086$ $-0.011$ $0.059$ $44,483$ Capital growth rate $\Delta k_{it}$ $-0.044$ $0.131$ $-0.104$ $-0.088$ $-0.056$ $44,483$ Labor share in nominal sales $s_{Nit}$ $0.198$ $0.088$ $0.135$ $0.185$ $0.248$ $44,483$ Material share in nominal sales $s_{Mit}$ $0.578$ $0.172$ $0.468$ $0.599$ $0.706$ $44,483$ $1 - s_{Nit} - s_{Mit}$ $0.225$ $0.138$ $0.120$ $0.198$ $0.303$ $44,483$ Average wage $W_{it}$ $4.914$ $1.606$ $3.844$ $4.732$ $5.773$ $44,483$ Number of workers $N_{it}$ $236$ $611$ $77$ $120$ $218$ $44,483$ Age $43$ $17$ $32$ $44$ $54$ $44,483$ Mumber of workers $N_{it}$ $0.000$ $0.000$ $0$ $0$ $0$ $44,483$ Importer dummy $EXP_{it}$ $0.000$ $0.000$ $0.000$ $0.000$ $0.000$ <td>Share of non-production workers</td> <td>0.398</td> <td>0.228</td> <td>0.218</td> <td>0.364</td> <td>0.547</td> <td>19 998</td>	Share of non-production workers	0.398	0.228	0.218	0.364	0.547	19 998
SRit0.0160.075-0.0200.0190.05719,998Non-exportersMeanSd. $Q_1$ $Q_2$ $Q_3$ NReal firm output growth $\Delta q_{it}$ -0.0070.127-0.071-0.0040.05844,483Labor growth rate $\Delta n_{it}$ -0.0110.083-0.048-0.0090.02844,483Material growth rate $\Delta m_{it}$ -0.0140.139-0.086-0.0110.05944,483Capital growth rate $\Delta k_{it}$ -0.0440.131-0.104-0.088-0.05644,483Labor share in nominal sales $s_{Nit}$ 0.1980.0880.1350.1850.24844,483Material share in nominal sales $s_{Mit}$ 0.5780.1720.4680.5990.70644,4831 - $s_{Nit} - s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Age431732445444,483Exporter dummy $EXP_{it}$ 0.0000.00000044,483Importer dummy $MNE_{it}$ 0.0790.27000044,483Export-sales ratio0.0000.0000.0000.00044,483Kaport-sales ratio0.0000.0000.0000.00044,483Kaport-sales ratio0.0000.0430.0310.6630.10444,483Capital intensity1.5650.8721.0811.61	Labor productivity	36 273	22.795	21 824	30 417	44 390	19 998
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	SB:	0.016	0.075	-0.020	0.019	0.057	19 998
Non experienceInealEditeq1eq2eq3InealReal firm output growth $\Delta q_{it}$ -0.0070.127-0.071-0.0040.05844,483Labor growth rate $\Delta n_{it}$ -0.0110.083-0.048-0.0090.02844,483Material growth rate $\Delta m_{it}$ -0.0140.139-0.086-0.0110.05944,483Capital growth rate $\Delta k_{it}$ -0.0440.131-0.104-0.088-0.05644,483Labor share in nominal sales $s_{Nit}$ 0.1980.0880.1350.1850.24844,483Material share in nominal sales $s_{Mit}$ 0.5780.1720.4680.5990.70644,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ 2366117712021844,483Exporter dummy $EXP_{it}$ 0.0000.00000044,483Importer dummy $IMP_{it}$ 0.1010.30100044,483MNE dummy $MNE_{it}$ 0.0790.27000044,483Kaport-sales ratio0.0000.0000.0000.00044,483Share of non-production workers0.3250.2400.1460.2700.442	Non-exporters	Mean	Sd	0.020 Q1	 	 	N
Itean init output growth $\Delta q_{it}$ $-0.001$ $0.121$ $-0.001$ $-0.004$ $-0.004$ $-0.004$ $-0.004$ Labor growth rate $\Delta n_{it}$ $-0.011$ $0.083$ $-0.048$ $-0.009$ $0.028$ $44,483$ Material growth rate $\Delta m_{it}$ $-0.014$ $0.139$ $-0.086$ $-0.011$ $0.059$ $44,483$ Capital growth rate $\Delta k_{it}$ $-0.044$ $0.131$ $-0.104$ $-0.088$ $-0.056$ $44,483$ Labor share in nominal sales $s_{Nit}$ $0.198$ $0.088$ $0.135$ $0.185$ $0.248$ $44,483$ Material share in nominal sales $s_{Mit}$ $0.578$ $0.172$ $0.468$ $0.599$ $0.706$ $44,483$ $1 - s_{Nit} - s_{Mit}$ $0.225$ $0.138$ $0.120$ $0.198$ $0.303$ $44,483$ Average wage $W_{it}$ $4.914$ $1.606$ $3.844$ $4.732$ $5.773$ $44,483$ Number of workers $N_{it}$ $236$ $611$ $77$ $120$ $218$ $44,483$ Age $43$ $17$ $32$ $44$ $54$ $44,483$ Importer dummy $IMP_{it}$ $0.101$ $0.301$ $0$ $0$ $44,483$ MNE dummy $MNE_{it}$ $0.079$ $0.270$ $0$ $0$ $0$ $44,483$ Export-sales ratio $0.000$ $0.000$ $0.000$ $0.000$ $44,483$ Capital intensity $1.565$ $0.872$ $1.081$ $1.611$ $2.115$ $44,483$ Share of non-production workers $0.325$ $0.240$ $0.146$ $0.270$ $0.442$	$\frac{1}{1} \frac{1}{1} \frac{1}$	-0.007	0.127	-0.071	-0.004	0.058	11 183
Labor growth rate $\Delta m_{it}$ -0.0110.000-0.040-0.0000.02044,400Material growth rate $\Delta m_{it}$ -0.0140.139-0.086-0.0110.05944,483Capital growth rate $\Delta k_{it}$ -0.0440.131-0.104-0.088-0.05644,483Labor share in nominal sales $s_{Nit}$ 0.1980.0880.1350.1850.24844,483Material share in nominal sales $s_{Mit}$ 0.5780.1720.4680.5990.70644,4831 - $s_{Nit} - s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ 2366117712021844,483Exporter dummy $EXP_{it}$ 0.0000.00000044,483Importer dummy $IMP_{it}$ 0.1010.30100044,483Kaper statio0.0000.0000.0000.00044,483MAE dummy $MNE_{it}$ 0.1010.3010044,483Kaport-sales ratio0.0000.0000.0000.00044,483Kaport-sales ratio0.0000.0000.0000.00044,483Capital intensity1.5650.8721.0811.6112.115Share of non-production workers0.3250.2400.1460.2700.44244,483	Labor growth rate $\Delta n_{it}$	-0.001	0.121	-0.071	-0.004	0.000	14 483
Autorial growth rate $\Delta k_{it}$ 0.0110.1010.1000.0000.0110.0001,100Capital growth rate $\Delta k_{it}$ -0.0440.131-0.104-0.088-0.05644,483Labor share in nominal sales $s_{Nit}$ 0.1980.0880.1350.1850.24844,483Material share in nominal sales $s_{Mit}$ 0.5780.1720.4680.5990.70644,483 $1 - s_{Nit} - s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ 2366117712021844,483Age431732445444,483Importer dummy $EXP_{it}$ 0.0000.00000044,483MNE dummy $MNE_{it}$ 0.1010.30100044,483Export-sales ratio0.0000.0000.0000.00044,483Herfindahl-Hirschman Index $HHI_{jt}$ 0.0690.0430.0310.0630.10444,483Share of non-production workers0.3250.2400.1460.2700.44244,483	Material growth rate $\Delta m_{ii}$	-0.014	0.139	-0.086	-0.011	0.020	44 483
Labor share in nominal sales $s_{Nit}$ 0.1980.0880.1350.1850.24844,483Material share in nominal sales $s_{Mit}$ 0.5780.1720.4680.5990.70644,483 $1 - s_{Nit} - s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ 2366117712021844,483Age431732445444,483Exporter dummy $EXP_{it}$ 0.0000.0000044,483Importer dummy $IMP_{it}$ 0.1010.3010044,483MNE dummy $MNE_{it}$ 0.0000.0000.0000.00044,483Export-sales ratio0.0000.0000.0000.00044,483Herfindahl-Hirschman Index $HHI_{jt}$ 0.6690.0430.0310.0630.104Share of non-production workers0.3250.2400.1460.2700.44244,483	Capital growth rate $\Delta k_{\perp}$	-0.044	0.131	-0.104	-0.088	-0.056	44.483
Material share in nominal sales $s_{Mit}$ 0.1000.1000.1100.1101.110Material share in nominal sales $s_{Mit}$ 0.5780.1720.4680.5990.70644,483 $1 - s_{Nit} - s_{Mit}$ 0.2250.1380.1200.1980.30344,483Average wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ 2366117712021844,483Age431732445444,483Exporter dummy $EXP_{it}$ 0.0000.0000044,483Importer dummy $IMP_{it}$ 0.1010.301000MNE dummy $MNE_{it}$ 0.0790.2700044,483Export-sales ratio0.0000.0000.0000.00044,483Herfindahl-Hirschman Index $HHI_{jt}$ 0.0690.0430.0310.0630.10444,483Share of non-production workers0.3250.2400.1460.2700.44244,483	Labor share in nominal sales $s_{Nit}$	0.198	0.088	0.135	0.185	0.248	44.483
1 $0.112$ $0.112$ $0.112$ $0.100$ $0.100$ $0.100$ $0.100$ $1.100$ $1 - s_{Nit} - s_{Mit}$ $0.225$ $0.138$ $0.120$ $0.198$ $0.303$ $44,483$ Average wage $W_{it}$ $4.914$ $1.606$ $3.844$ $4.732$ $5.773$ $44,483$ Number of workers $N_{it}$ $236$ $611$ $77$ $120$ $218$ $44,483$ Age $43$ $17$ $32$ $44$ $54$ $44,483$ Exporter dummy $EXP_{it}$ $0.000$ $0.000$ $0$ $0$ $44,483$ Importer dummy $IMP_{it}$ $0.101$ $0.301$ $0$ $0$ $0$ MNE dummy $MNE_{it}$ $0.079$ $0.270$ $0$ $0$ $0$ Export-sales ratio $0.000$ $0.000$ $0.000$ $0.000$ $44,483$ Herfindahl-Hirschman Index $HHI_{jt}$ $0.669$ $0.043$ $0.031$ $0.063$ $0.104$ Capital intensity $1.565$ $0.872$ $1.081$ $1.611$ $2.115$ $44,483$	Material share in nominal sales $s_{Mit}$	0.578	0.172	0.468	0.599	0.706	44.483
A verage wage $W_{it}$ 4.9141.6063.8444.7325.77344,483Number of workers $N_{it}$ 2366117712021844,483Age431732445444,483Exporter dummy $EXP_{it}$ 0.0000.00000044,483Importer dummy $IMP_{it}$ 0.1010.30100044,483MNE dummy $MNE_{it}$ 0.0790.27000044,483Export-sales ratio0.0000.0000.0000.00044,483Herfindahl-Hirschman Index $HHI_{jt}$ 0.0690.0430.0310.0630.10444,483Share of non-production workers0.3250.2400.1460.2700.44244,483	$1 - s_{Nit} - s_{Mit}$	0.225	0.138	0.120	0.198	0.303	44.483
Number of workers $N_{it}$ 2366117712021844,483Age431732445444,483Exporter dummy $EXP_{it}$ 0.0000.00000044,483Importer dummy $IMP_{it}$ 0.1010.30100044,483MNE dummy $MNE_{it}$ 0.0790.27000044,483Export-sales ratio0.0000.0000.0000.00044,483Herfindahl-Hirschman Index $HHI_{jt}$ 0.0690.0430.0310.0630.10444,483Capital intensity1.5650.8721.0811.6112.11544,483Share of non-production workers0.3250.2400.1460.2700.44244,483	A verage wage W:	4.914	1.606	3.844	4.732	5.773	44.483
Age431732445444,483Exporter dummy $EXP_{it}$ 0.0000.00000044,483Importer dummy $IMP_{it}$ 0.1010.30100044,483MNE dummy $MNE_{it}$ 0.0790.27000044,483Export-sales ratio0.0000.0000.0000.0000.00044,483Herfindahl-Hirschman Index $HHI_{jt}$ 0.0690.0430.0310.0630.10444,483Capital intensity1.5650.8721.0811.6112.11544,483Share of non-production workers0.3250.2400.1460.2700.44244,483	Number of workers $N_{i*}$	236	611	77	120	218	44.483
Exporter dummy $EXP_{it}$ 0.0000.0000044,483Importer dummy $IMP_{it}$ 0.1010.30100044,483MNE dummy $MNE_{it}$ 0.0790.27000044,483Export-sales ratio0.0000.0000.0000.0000.00044,483Herfindahl-Hirschman Index $HHI_{jt}$ 0.0690.0430.0310.0630.10444,483Capital intensity1.5650.8721.0811.6112.11544,483Share of non-production workers0.3250.2400.1460.2700.44244,483	Age	43	17	32	44	54	44.483
Importer dummy $IMP_{it}$ 0.1010.30100044,483MNE dummy $MNE_{it}$ 0.0790.27000044,483Export-sales ratio0.0000.0000.0000.0000.00044,483Herfindahl-Hirschman Index $HHI_{jt}$ 0.0690.0430.0310.0630.10444,483Capital intensity1.5650.8721.0811.6112.11544,483Share of non-production workers0.3250.2400.1460.2700.44244,483	Exporter dummy $EXP_{i}$	0.000	0.000	0	0	0	44.483
MNE dummy $MNE_{it}$ 0.0790.27000044,483Export-sales ratio0.0000.0000.0000.0000.00044,483Herfindahl-Hirschman Index $HHI_{jt}$ 0.0690.0430.0310.0630.10444,483Capital intensity1.5650.8721.0811.6112.11544,483Share of non-production workers0.3250.2400.1460.2700.44244,483	Importer dummy IMP:	0.101	0.301	Ő	0 0	0 0	44.483
Export-sales ratio $0.000$ $0.000$ $0.000$ $0.000$ $0.000$ $0.000$ $44,483$ Herfindahl-Hirschman Index $HHI_{jt}$ $0.069$ $0.043$ $0.031$ $0.063$ $0.104$ $44,483$ Capital intensity $1.565$ $0.872$ $1.081$ $1.611$ $2.115$ $44,483$ Share of non-production workers $0.325$ $0.240$ $0.146$ $0.270$ $0.442$ $44,483$	$\frac{1}{1} MNE \text{ dummy } MNE_{it}$	0.079	0.270	0	0	0	44.483
Herfindahl-Hirschman Index $HHI_{jt}$ 0.0690.0430.0310.0630.10444,483Capital intensity1.5650.8721.0811.6112.11544,483Share of non-production workers0.3250.2400.1460.2700.44244,483	Export-sales ratio	0.000	0.000	0 000	0 000	0 000	44 483
Capital intensity         1.565         0.872         1.081         1.611         2.115         44,483           Share of non-production workers         0.325         0.240         0.146         0.270         0.442         44,483	Herfindahl-Hirschman Index HHI	0.069	0.043	0.031	0.063	0.104	44.483
Share of non-production workers         0.325         0.240         0.146         0.270         0.442         44,483	Capital intensity	1.565	0.872	1.081	1.611	2.115	44.483
	Share of non-production workers	0.325	0.240	0.146	0.270	0.449	44.483
Labor productivity 30 920 23 363 17 586 24 729 36 246 44 483	Labor productivity	30.920	23.363	17.586	24.729	36.246	44.483
$SR_{it}$ 0.013 0.072 -0.021 0.015 0.050 44.483	CD	0.013	0.079	0.091	0.015	0.050	11 192

 Table 1: Descriptive statistics: Total estimation sample, and by export and MNE status, 1994-2012

=

MNEs	Mean	Sd.	$Q_1$	$Q_2$	$Q_3$	N
Real firm output growth $\Delta q_{it}$	0.001	0.133	-0.062	0.007	0.071	$14,\!557$
Labor growth rate $\Delta n_{it}$	-0.006	0.079	-0.038	-0.005	0.028	$14,\!557$
Material growth rate $\Delta m_{it}$	-0.008	0.141	-0.079	-0.003	0.068	$14,\!557$
Capital growth rate $\Delta k_{it}$	-0.043	0.127	-0.105	-0.087	-0.048	$14,\!557$
Labor share in nominal sales $s_{Nit}$	0.180	0.074	0.127	0.170	0.222	$14,\!557$
Material share in nominal sales $s_{Mit}$	0.614	0.151	0.525	0.636	0.723	$14,\!557$
$1 - s_{Nit} - s_{Mit}$	0.206	0.124	0.116	0.178	0.269	$14,\!557$
Average wage $W_{it}$	6.019	1.791	4.734	5.882	7.128	$14,\!557$
Number of workers $N_{it}$	1,208	3,738	165	375	904	$14,\!557$
Age	52	18	42	52	62	$14,\!557$
Exporter dummy $\text{EXP}_{it}$	0.758	0.428	1	1	1	$14,\!557$
Importer dummy $IMP_{it}$	0.660	0.474	0	1	1	$14,\!557$
MNE dummy $MNE_{it}$	1.000	0.000	1	1	1	$14,\!557$
Export-sales ratio	0.097	0.151	0.000	0.032	0.127	$14,\!557$
Herfindahl-Hirschman Index $\mathrm{HHI}_{Jt}$	0.074	0.045	0.028	0.074	0.109	$14,\!557$
Capital intensity	1.817	0.770	1.316	1.824	2.310	$14,\!557$
Share of non-production workers	0.397	0.232	0.215	0.364	0.545	$14,\!557$
Labor productivity	39.496	24.312	23.797	33.239	48.827	$14,\!557$
Non-MNEs	Mean	Sd.	$Q_1$	$Q_2$	$Q_3$	N
Real firm output growth $\Delta q_{it}$	-0.006	0.129	-0.071	-0.004	0.060	49,924
Labor growth rate $\Delta n_{it}$	-0.010	0.082	-0.047	-0.009	0.028	$49,\!924$
Material growth rate $\Delta m_{it}$	-0.013	0.141	-0.087	-0.011	0.060	49,924
Capital growth rate $\Delta k_{it}$	-0.043	0.133	-0.105	-0.089	-0.055	$49,\!924$
Labor share in nominal sales $s_{Nit}$	0.202	0.088	0.139	0.190	0.254	49,924
Material share in nominal sales $s_{Mit}$	0.573	0.171	0.463	0.593	0.699	49,924
$1 - s_{Nit} - s_{Mit}$	0.225	0.138	0.121	0.199	0.303	49,924
Average wage $W_{it}$	4.992	1.586	3.925	4.831	5.877	49,924
Number of workers $N_{it}$	207	337	78	121	219	49,924
Age	43	17	32	44	54	49,924
Exporter dummy $\text{EXP}_{it}$	0.179	0.384	0	0	0	49,924
Importer dummy $IMP_{it}$	0.156	0.363	0	0	0	49,924
MNE dummy $MNE_{it}$	0.000	0.000	0	0	0	49,924
Export-sales ratio	0.014	0.064	0.000	0.000	0.000	49,924
Herfindahl-Hirschman Index $\mathrm{HHI}_{Jt}$	0.069	0.043	0.031	0.065	0.103	49,924
Capital intensity	1.559	0.867	1.077	1.611	2.113	49,924
Share of non-production workers	0.333	0.239	0.154	0.281	0.456	49,924
Labor productivity	30.564	22.628	17.747	24.684	35.976	49,924
$SR_{it}$	0.014	0.073	-0.021	0.016	0.052	49,924

Table 1 - Continued: Descriptive statistics: Total estimation sample, and by export and MNE status, 1994-2012

Note:  $SR_{it} = \Delta q_{it} - s_{Nit}\Delta n_{it} - s_{Mit}\Delta m_{it} - (1 - s_{Nit} - s_{Mit})\Delta k_{it}.$ 

	SI	pecification 1 (baseli	ne)		Specification 2	
	$\Pr(PMS=IC \mathbf{x})^a$	$\Pr(\text{LMS}=\text{EB} \mathbf{x})^b$	$\Pr(LMS=MO \mathbf{x})^b$	$\Pr(PMS=IC \mathbf{x})^a$	$\Pr(\text{LMS}=\text{EB} \mathbf{x})^b$	$\Pr(LMS=MO \mathbf{x})^b$
	dF/dx	dF/dx	dF/dx	dF/dx	dF/dx	dF/dx
Exporter dummy (EXP)	0.011	0.042***	-0.028***	0.014**	0.037***	-0.024***
	(0.007)	(0.013)	(0.007)	(0.006)	(0.011)	(0.005)
MNE dummy (MNE)	-0.030***	-0.047***	0.013	-0.023***	-0.058***	0.020***
	(0.010)	(0.018)	(0.010)	(0.005)	(0.012)	(0.008)
EXP×MNE	0.011	-0.021	0.013			
	(0.012)	(0.021)	(0.012)			
TFP	-0.020***	-0.017***	$0.017^{***}$	-0.020***	-0.017***	$0.017^{***}$
	(0.003)	(0.004)	(0.003)	(0.003)	(0.004)	(0.002)
HHI	0.131	0.420**	-0.286**	0.129	$0.424^{**}$	-0.289**
	(0.138)	(0.193)	(0.142)	(0.138)	(0.193)	(0.142)
Size	-0.034***	-0.048***	$0.021^{***}$	-0.034***	-0.048***	0.021***
	(0.003)	(0.005)	(0.003)	(0.003)	(0.005)	(0.003)
Age	0.0002	0.0017***	-0.0007***	0.0001	$0.0016^{***}$	-0.0007***
	(0.0001)	(0.0003)	(0.0002)	(0.0001)	(0.0002)	(0.0001)
Share of non-production workers	-0.053***	-0.076***	$0.030^{**}$	-0.053***	-0.076***	0.030**
	(0.012)	(0.017)	(0.012)	(0.012)	(0.017)	(0.011)
Time dummies	yes	yes	yes	yes	yes	yes
Industry dummies	yes	yes	yes	yes	yes	yes
Log likelihood	-16,575.3	-45,	580.7	-16,576.8	-45,583.8	
Pseudo $R^2$	0.539			0.539		
Ν	64,481	64.	481	64,481	64,	481

Table 2: Probit models on the likelihood of being characterized by PMS=IC and LMS={EB,MO}

Notes: Significance level of \*\*\*1%, \*\*5%, \*10%. Bootstrapped standard errors (50 replications).

<sup>*a*</sup>: Marginal effects of univariate probit model. <sup>*b*</sup>: Marginal effects of bivariate probit model.

	Spe	cification 1 (base	eline)		Specification 2	
Dependent variable	$\ln \widehat{\mu}_{it+1}$	$\ln\left(\frac{\widehat{\phi}_{it+1}}{1-\widehat{\phi}_{it+1}}\right)$	$\ln(\widehat{arepsilon}_w^N)_{it+1}$	$\ln \widehat{\mu}_{it+1}$	$\ln\left(\frac{\widehat{\phi}_{it+1}}{1-\widehat{\phi}_{it+1}}\right)$	$\ln(\widehat{\varepsilon}_w^N)_{it+1}$
Exporter dummy (EXP)	0.030***	0.046	0.209***	0.027***	0.080*	0.201***
	(0.010)	(0.050)	(0.036)	(0.008)	(0.044)	(0.032)
MNE dummy (MNE)	-0.034***	-0.108**	-0.054	-0.040***	-0.124**	-0.073***
	(0.011)	(0.055)	(0.049)	(0.009)	(0.054)	(0.028)
EXP×MNE	-0.010	0.026	-0.032			
	(0.013)	(0.060)	(0.059)			
TFP	0.001	-0.049***	-0.301***	0.001	-0.052***	-0.302***
	(0.005)	(0.018)	(0.016)	(0.005)	(0.016)	(0.016)
HHI	0.433*	-0.633	6.957***	0.434*	-0.618	6.965***
	(0.244)	(0.553)	(0.753)	(0.244)	(0.727)	(0.741)
Size	-0.023***	-0.045	-0.343***	-0.024***	-0.059	-0.343***
	(0.004)	(0.041)	(0.017)	(0.004)	(0.044)	(0.017)
Age	0.0004**	-0.0006	0.004***	0.0004**	-0.0006	0.004***
	(0.0002)	(0.002)	(0.001)	(0.0002)	(0.002)	(0.001)
Share of non-production workers	-0.090***	0.274***	-0.166***	-0.090***	$0.218^{***}$	-0.166***
	(0.016)	(0.079)	(0.053)	(0.016)	(0.073)	(0.052)
Inverse Mills ratio	0.101***	0.065	-0.925***	0.102***	0.216	-0.927***
	(0.032)	(0.360)	(0.166)	(0.032)	(0.375)	(0.163)
Time dummies	yes	yes	yes	yes	yes	yes
Industry dummies	yes	yes	yes	yes	yes	yes
Adjusted $R^2$	0.146	0.268	0.493	0.146	0.268	0.493
N	42,302	23,243	16,207	42,302	23,243	16,207

 Table 3: Mean regression results (OLS)

Notes: Significance level of \*\*\*1%, \*\*5%, \*10%. Standard errors are heteroskedasticity consistent and clustered by enterprises.

Regressor	Quantile	$\ln \widehat{\mu}_{it+1}$	$\ln\left(\frac{\hat{\phi}_{it+1}}{1-\hat{\phi}_{it+1}}\right)$	$\ln(\widehat{\varepsilon}_w^N)_{it+1}$
EXP	q10	0.013***	0.050*	0.161***
		(0.002)	(0.030)	(0.025)
	q25	0.021***	$0.086^{***}$	$0.215^{***}$
		(0.003)	(0.027)	(0.020)
	q50	0.031***	$0.081^{***}$	$0.169^{***}$
		(0.005)	(0.022)	(0.019)
	q75	$0.044^{***}$	$0.074^{***}$	$0.152^{***}$
		(0.007)	(0.028)	(0.023)
	q90	0.041***	$0.108^{***}$	$0.178^{***}$
		(0.010)	(0.039)	(0.031)
F-test: Equality across all quantiles		0.000	0.961	0.000
F-test: Equality across $q5$ , $q50$ and $q95$		0.000	0.541	0.000
F-test: Equality across q5 and q95		0.473	0.382	0.001
MNE	q10	-0.007***	-0.069*	-0.049
		(0.003)	(0.040)	(0.038)
	q25	-0.008**	-0.085***	-0.075**
		(0.003)	(0.033)	(0.033)
	q50	-0.009*	-0.119***	-0.081**
		(0.005)	(0.029)	(0.034)
	q75	-0.019**	-0.109***	-0.115***
		(0.007)	(0.034)	(0.039)
	q90	-0.070***	-0.145***	-0.117*
		(0.012)	(0.038)	(0.066)
F-test: Equality across all quantiles		0.000	0.444	0.599
F-test: Equality across $q5$ , $q50$ and $q95$		0.000	0.116	0.783
F-test: Equality across $q5$ and $q95$		0.000	0.039	0.536
EXP×MNE	q10	-0.004	0.054	0.035
		(0.003)	(0.044)	(0.040)
	q25	-0.012***	0.038	-0.001
		(0.004)	(0.037)	(0.034)
	q50	-0.026***	0.010	-0.001
		(0.006)	(0.038)	(0.033)
	q75	-0.048***	-0.026	0.023
		(0.009)	(0.043)	(0.042)
	q90	-0.013	-0.051	0.074
		(0.014)	(0.058)	(0.079)
F-test: Equality across all quantiles		0.000	0.499	0.237
F-test: Equality across $q5$ , $q50$ and $q95$		0.001	0.932	0.189
F-test: Equality across q5 and q95		0.042	0.714	0.780

 Table 4: Quantile regression (QR) results - Baseline specification

Notes: Significance level of \*\*\*1%, \*\*5%, \*10%. Bootstrapped standard errors (50 replications). Results are based on pooled simultaneous-quantile regressions for

 $\tau \in \{0.05, 0.10, 0.20, 0.25, 0.30, 0.40, 0.50, 0.60, 0.70, 0.75, 0.80, 0.90, 0.95\}.$ 

Results for other regressors and other quantiles are available upon request.

F-tests for equality of regression coefficients across specified quantiles, p-values are reported.

$$\text{Graph 1: Average and quantile impact of export/MNE status on } \mathbf{y} = \{\ln \hat{\mu}_{it+1}, \ln \left(\frac{\hat{\theta}_{it+1}}{1-\hat{\Phi}_{it+1}}\right), \ln(\hat{e}_w^N)_{it+1}\}$$

$$(a) \text{ Dep. var.: } \ln \hat{\mu}_{it+1}$$

$$(b) \text{ Dep. var.: } \ln \left(\frac{\hat{\theta}_{it+1}}{1-\hat{\Phi}_{it+1}}\right)$$

$$(c) \text{ Dep. var.: } \ln \left(\frac{\hat{\theta}_{it+1}}{1-\hat{\Phi}_{it+1}}\right)$$

(c) Dep. var.:  $\ln(\widehat{\varepsilon}_w^N)_{it+1}$ 



Notes: Solid lines represent coefficient estimates and 95% confidence intervals of quantile regressions. For comparison, dashed lines mark coefficient estimates and 95% confidence intervals of OLS regressions.

### Statistical appendix

# of participations				
	# obs.	%	# firms	%
2	1,074	1.7	1,046	14.0
3	1,817	2.8	886	11.9
4	1,757	2.7	566	7.6
5	2,297	3.6	551	7.4
6	2,043	3.2	384	5.1
7	2,117	3.3	333	4.5
8	2,138	3.3	294	3.9
9	2,418	3.7	283	3.8
10	1,824	2.8	178	2.4
11	2,310	3.6	212	2.8
12	2,168	3.4	186	2.5
13	1,949	3.0	155	2.1
14	2,198	3.4	165	2.2
15	2,153	3.3	150	2.0
16	2,332	3.6	154	2.1
17	2,679	4.2	167	2.2
18	4,369	6.8	257	3.4
19	26,838	41.6	$1,\!491$	20.0
Total	64,481	100.0	7,458	100.0

 Table A.1: Panel structure: Number of participations

	# obs.	%	# firms	%	% of	% of
					exporters	MNEs
Total	64,481	100.0	7,458	100.0	24.7	15.4
Food products and beverages	8,644	13.4	957	12.8	10.1	7.5
Textiles and wearing apparel	3,345	5.2	498	6.7	13.0	13.8
Wood, wooden products, and furniture	1,014	1.6	176	2.4	5.8	9.8
Pulp, paper and paper products	2,362	3.7	251	3.4	7.2	4.0
Publishing and printing	4,030	6.2	438	5.9	6.3	6.3
Chemicals	5,869	9.1	557	7.5	50.4	26.5
Petroleum and coal products	4,065	6.3	458	6.1	23.3	19.3
Non-metallic mineral products	3,101	4.8	377	5.1	18.3	11.1
Iron and steel	2,213	3.4	240	3.2	23.1	15.6
Non-ferrous metals	1,659	2.6	183	2.5	31.5	18.5
Fabricated metal products	5,164	8.0	609	8.2	18.0	10.7
Machinery	7,005	10.9	797	10.7	43.8	20.3
Electrical machinery	6,054	9.4	820	11.0	29.0	16.5
Transport equipment	6,411	9.9	670	9.0	29.0	22.4
Other manufacturing	3,545	5.5	427	5.7	41.6	21.2
Exporters	19,998	31.0	1,859	24.9	100.0	45.5
Non-exporters	44,483	69.0	$5,\!599$	75.1	0.0	5.7
MNEs	14,557	22.6	1,163	15.6	72.7	100.0
Non-MNEs	49,924	77.4	$6,\!295$	84.4	16.1	0.0

 Table A.2: Repartition by industry, export and MNE status

	Expo	rters	MN	IEs
	(1)	(2)	(1)	(2)
$\ln(\text{employment}_{it})$	0.780***	-	1.113***	-
$\ln(\text{value added per worker}_{it})$	$0.273^{***}$	$0.129^{***}$	$0.278^{***}$	$0.122^{***}$
$\ln(\mathrm{TFP}_{it}) = \widehat{\omega}_{it}$	$0.139^{***}$	0.020***	$0.111^{***}$	$0.029^{***}$
$\ln(\text{wage}_{it})$	$0.196^{***}$	$0.085^{***}$	$0.192^{***}$	$0.066^{***}$
$\ln(\text{capital per worker}_{it})$	$0.168^{***}$	$0.069^{***}$	$0.257^{***}$	$0.097^{***}$
$\ln(\text{share of non-production workers}_{it})$	$0.306^{***}$	$0.300^{***}$	$0.250^{***}$	$0.255^{***}$
$\ln(\mathrm{markup}_{it}) = \ln \widehat{\mu}_{it}$	-0.099***	-0.011	-0.121***	-0.059***
Additional covariates				
Industry fixed effects	No	Yes	No	Yes
$\ln(\text{employment})_{it}$	No	Yes	No	Yes

Table A.3: Exporter and MNE premia

Notes: \*\*\*: Significance level of 1%. Each row summarizes the average percent difference

for a particular characteristic between either exporters and non-exporters, or MNEs and non-MNEs . All results in column (1) are from bivariate OLS regressions of a firm characteristic on a dummy variable indicating either a firm's export status or a firm's MNE status. Colum (2) includes industry fixed effects and  $\ln(\text{employment}_{it})$  as additional controls.

$\ln(\mathrm{TFP}_{it}) = \widehat{\omega}_{it}$	Expo	orters	MNEs		
	(1)	(2)	(1)	(2)	
Food products and beverages	-0.045***	0.006***	-0.081***	0.004**	
Textiles and wearing apparel	0.003	0.021***	-0.002	$0.016^{***}$	
Wood, wooden products, and furniture	0.030*	0.003	0.047**	0.000	
Pulp, paper and paper products	0.067***	$0.032^{*}$	$0.142^{***}$	$0.036^{*}$	
Publishing and printing	-0.122***	-0.027**	-0.131***	-0.004	
Chemicals	0.000	0.011	0.005	0.027***	
Petroleum and coal products	0.094***	$0.011^{**}$	0.101***	$0.017^{***}$	
Non-metallic mineral products	-0.072***	0.001	-0.075***	$0.019^{***}$	
Iron and steel	0.102**	-0.020	$0.185^{***}$	0.023	
Non-ferrous metals	0.129***	0.012	$0.144^{***}$	-0.012	
Fabricated metal products	-0.021***	$0.007^{***}$	-0.026***	$0.013^{***}$	
Machinery	0.004	$0.027^{***}$	0.002	$0.045^{***}$	
Electrical machinery	0.207***	$0.051^{***}$	$0.311^{***}$	$0.095^{***}$	
Transport equipment	0.057***	$0.005^{**}$	$0.073^{***}$	$0.013^{***}$	
Other manufacturing	0.011***	$0.009^{*}$	$0.015^{***}$	$0.013^{*}$	
$\ln(\mathrm{markup}_{it}) = \ln \widehat{\mu}_{it}$	Expo	orters	MN	NEs	
$\ln(\mathrm{markup}_{it}) = \ln \widehat{\mu}_{it}$	(1)	orters (2)	(1) MN	NEs (2)	
$\ln(\mathrm{markup}_{it}) = \ln \widehat{\mu}_{it}$ Food products and beverages	Expo (1) 0.008	0.034	(1) -0.056***	NEs (2) -0.019	
$\ln(\text{markup}_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel	Expo (1) 0.008 -0.045	(2) 0.034 -0.049	(1) -0.056*** -0.083**	VEs (2) -0.019 -0.096**	
$\ln(\text{markup}_{it}) = \ln \widehat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture	Expo (1) 0.008 -0.045 -0.053	(2)           0.034           -0.049           -0.001	(1) -0.056*** -0.083** -0.205***	VEs (2) -0.019 -0.096** -0.135*	
$\ln(\text{markup}_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products	Expe (1) 0.008 -0.045 -0.053 0.002	(2)           0.034           -0.049           -0.001           0.000	MN (1) -0.056*** -0.083** -0.205*** -0.102***	NEs (2) -0.019 -0.096** -0.135* -0.113***	
$\ln(\text{markup}_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing	Expe (1) 0.008 -0.045 -0.053 0.002 -0.109**	(2)           0.034           -0.049           -0.001           0.000           -0.078	MN (1) -0.056*** -0.083** -0.205*** -0.102*** -0.114***	NEs (2) -0.019 -0.096** -0.135* -0.113*** -0.074	
$\ln(\text{markup}_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing Chemicals	Expe (1) 0.008 -0.045 -0.053 0.002 -0.109** 0.015	(2)           0.034           -0.049           -0.001           0.000           -0.078           0.024	MN (1) -0.056*** -0.083** -0.205*** -0.102*** -0.114*** -0.035*	NEs (2) -0.019 -0.096** -0.135* -0.135* -0.074 -0.074 -0.040*	
$\ln(\max \sup_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing Chemicals Petroleum and coal products	Expe (1) 0.008 -0.045 -0.053 0.002 -0.109** 0.015 -0.029	(2)           0.034           -0.049           -0.001           0.000           -0.078           0.024           -0.012	MN (1) -0.056*** -0.083** -0.205*** -0.102*** -0.114*** -0.035* -0.081***	NEs           (2)           -0.019           -0.096**           -0.135*           -0.113***           -0.074           -0.040*           -0.070***	
$\ln(\text{markup}_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing Chemicals Petroleum and coal products Non-metallic mineral products	Expe (1) 0.008 -0.045 -0.053 0.002 -0.109** 0.015 -0.029 0.015	(2)           0.034           -0.049           -0.001           0.000           -0.078           0.024           -0.012           0.051	MN (1) -0.056*** -0.205*** -0.102*** -0.114*** -0.035* -0.081*** -0.085**	(2)           -0.019           -0.135*           -0.113***           -0.074           -0.040*           -0.070***           -0.062	
$\ln(\max \sup_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing Chemicals Petroleum and coal products Non-metallic mineral products Iron and steel	Expe (1) 0.008 -0.045 -0.053 0.002 -0.109** 0.015 -0.029 0.015 -0.021	(2)           0.034           -0.049           -0.001           0.000           -0.078           0.024           -0.012           0.051           -0.024	MN (1) -0.056*** -0.083** -0.205*** -0.102*** -0.114*** -0.035* -0.081*** -0.085** -0.105**	VEs           (2)           -0.019           -0.135*           -0.113***           -0.074           -0.040*           -0.070***           -0.062           -0.164***	
$\ln(\max \sup_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing Chemicals Petroleum and coal products Non-metallic mineral products Iron and steel Non-ferrous metals	Expe (1) 0.008 -0.045 -0.053 0.002 -0.109** 0.015 -0.029 0.015 -0.021 -0.136***	(2)           0.034           -0.049           -0.001           0.000           -0.078           0.024           -0.012           0.051           -0.024           -0.129***	MN (1) -0.056*** -0.083** -0.205*** -0.102*** -0.114*** -0.035* -0.081*** -0.085** -0.105** -0.129***	VEs           (2)           -0.019           -0.135*           -0.135*           -0.074           -0.040*           -0.062           -0.164****           -0.119**	
$\ln(\max \sup_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing Chemicals Petroleum and coal products Non-metallic mineral products Iron and steel Non-ferrous metals Fabricated metal products	Expo (1) 0.008 -0.045 -0.053 0.002 -0.109** 0.015 -0.029 0.015 -0.021 -0.136*** -0.043	(2)           0.034           -0.049           -0.001           0.000           -0.078           0.024           -0.012           0.051           -0.024           -0.024           -0.024           -0.008	MN (1) -0.056*** -0.083** -0.205*** -0.102*** -0.035* -0.081*** -0.085** -0.105** -0.129*** -0.095**	VEs           (2)           -0.019           -0.135*           -0.135*           -0.074           -0.040*           -0.062           -0.164****           -0.119**           -0.056*	
$\ln(\max \sup_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing Chemicals Petroleum and coal products Non-metallic mineral products Iron and steel Non-ferrous metals Fabricated metal products Machinery	Expe (1) 0.008 -0.045 -0.053 0.002 -0.109** 0.015 -0.029 0.015 -0.021 -0.136*** -0.043 -0.070***	(2)           0.034           -0.049           -0.001           0.000           -0.078           0.024           -0.012           0.051           -0.024           -0.024           -0.008           -0.010	MN (1) -0.056*** -0.205*** -0.102*** -0.114*** -0.035* -0.081*** -0.085** -0.105** -0.129*** -0.095*** -0.112***	(2)           -0.019           -0.135*           -0.135*           -0.074           -0.040*           -0.062           -0.164***           -0.056*           -0.019	
$\ln(\max \sup_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing Chemicals Petroleum and coal products Non-metallic mineral products Iron and steel Non-ferrous metals Fabricated metal products Machinery Electrical machinery	Expe (1) 0.008 -0.045 -0.053 0.002 -0.109** 0.015 -0.029 0.015 -0.021 -0.136*** -0.043 -0.070*** -0.095***	(2)           0.034           -0.049           -0.001           0.000           -0.078           0.024           -0.012           0.051           -0.024           -0.129***           -0.008           -0.010	MIN -0.056*** -0.083** -0.205*** -0.102*** -0.114*** -0.035* -0.081*** -0.085** -0.105** -0.105** -0.129*** -0.112*** -0.166***	VEs           (2)           -0.019           -0.135*           -0.135*           -0.074           -0.074           -0.062           -0.164***           -0.119**           -0.056*           -0.019	
$\ln(\max \sup_{it}) = \ln \hat{\mu}_{it}$ Food products and beverages Textiles and wearing apparel Wood, wooden products, and furniture Pulp, paper and paper products Publishing and printing Chemicals Petroleum and coal products Non-metallic mineral products Iron and steel Non-ferrous metals Fabricated metal products Machinery Electrical machinery Transport equipment	Expe (1) 0.008 -0.045 -0.053 0.002 -0.109** 0.015 -0.029 0.015 -0.021 -0.136*** -0.043 -0.070*** -0.095*** -0.046***	(2)           0.034           -0.049           -0.001           0.000           -0.078           0.024           -0.012           0.051           -0.024           -0.129***           -0.008           -0.010           -0.058**           -0.002	MIN (1) -0.056*** -0.083** -0.205*** -0.102*** -0.114*** -0.035* -0.081*** -0.085** -0.085** -0.105** -0.129*** -0.112*** -0.166*** -0.073***	VEs           (2)           -0.019           -0.135*           -0.135*           -0.074           -0.074           -0.062           -0.119**           -0.056*           -0.019           -0.121***           -0.029	

Table A.4: Exporter and MNE premia by industry

Notes: Significance level of \*\*\*1%, \*\*5%, \*10%. Each row summarizes the average percent difference for TFP (top) or markups (bottom) between either exporters and non-exporters, or MNEs and non-MNEs in a particular industry.
All results in column (1) are from bivariate OLS regressions of each firm characteristic on a dummy variable indicating either a firm's export status or a firm's MNE status. Colum (2) includes ln(employment<sub>it</sub>)

as an additional control.

	# obs.	$\#~{\rm firms}$	PC-PR	IC-PR	PC-EB	IC-EB	PC-MO	IC-MO
Total	64,481	$7,\!458$	12.0	17.9	1.6	42.0	11.0	15.4
Exporters	19,998	1,859	11.4	19.7	1.4	45.0	8.1	14.4
Non-exporters	44,483	5,599	12.8	13.5	2.0	36.1	18.2	17.5
MNEs	14,557	1,163	11.7	18.2	1.6	44.1	9.5	14.9
Non-MNEs	49,924	$6,\!295$	13.7	16.8	1.6	34.1	16.9	17.0

Table A.5: Percentage of firms in each regime: Total and by export/MNE status

Notes:  $R \in \Re = \{PC-PR, IC-PR, PC-EB, IC-EB, PC-MO, IC-MO\}$ , with PMS  $\in \{PC, IC\}$  and LMS  $\in \{PR, EB, MO\}$ . PC refers to perfect competition, IC to imperfection competition, PR to perfect competition/right-to-manage bargaining, EB to efficient bargaining and MO to monopsony.

	# obs.	$\#~{\rm firms}$	PC-PR	IC-PR	PC-EB	IC-EB	PC-MO	IC-MO
Total	6,4481	7,458	12.0	17.9	1.6	42.0	11.0	15.4
Food products and beverages	8,644	957	0.0	25.0	0.0	45.8	0.8	28.4
Textiles and wearing apparel	$3,\!345$	498	0.8	28.3	0.0	57.2	1.2	12.5
Wood, wooden products and furniture	$1,\!014$	176	10.2	13.6	0.0	76.1	0.0	0.0
Pulp, paper and paper products	2,362	251	37.5	17.9	8.0	36.7	0.0	0.0
Publishing and printing	4,030	438	2.7	37.9	0.0	59.1	0.0	0.2
Chemicals	$5,\!869$	557	0.0	4.5	0.0	19.6	2.9	73.1
Petroleum and coal products	4,065	458	16.2	8.3	2.0	72.1	1.5	0.0
Non-metallic mineral products	$3,\!101$	377	4.2	36.9	0.0	28.6	8.5	21.8
Iron and steel	2,213	240	12.5	10.4	0.0	2.9	72.9	1.3
Non-ferrous metals	$1,\!659$	183	45.9	30.1	0.0	22.4	1.6	0.0
Fabricated metal products	$5,\!164$	609	2.3	45.5	0.0	40.4	5.3	6.6
Machinery	$7,\!005$	797	0.0	1.4	0.0	3.1	67.9	27.6
Electrical machinery	$6,\!054$	820	66.3	3.4	11.3	18.9	0.0	0.0
Transport equipment	$6,\!411$	670	0.1	10.4	0.0	82.7	0.0	6.7
Other manufacturing	$3,\!545$	427	0.2	12.9	0.0	82.7	0.5	3.7

Table A.6: Percentage of firms in each regime by industry

Notes:  $R \in \Re = \{PC-PR, IC-PR, PC-EB, IC-EB, PC-MO, IC-MO\}$ , with PMS  $\in \{PC, IC\}$  and LMS  $\in \{PR, EB, MO\}$ . PC refers to perfect competition, IC to imperfection competition,

PR to perfect competition/right-to-manage bargaining, EB to efficient bargaining and MO to monopsony.

Total	TFP	EXP	MNE	$EXP \times MNE$	IMP	IC	EB	MO	Size	HHI	Age	NonP
TFP	1											
Exporter dummy (EXP)	0.047**	1										
MNE dummy (MNE)	0.059**	$0.523^{**}$	1									
EXP×MNE	0.071**	$0.678^{**}$	$0.842^{**}$	1								
Importer dummy (IMP)	0.052**	$0.567^{**}$	$0.475^{**}$	$0.506^{**}$	1							
IC dummy (IC)	-0.031**	-0.100**	-0.075**	-0.078**	-0.093**	1						
EB dummy (EB)	-0.034**	-0.073**	-0.074**	-0.071**	-0.080**	$0.381^{**}$	1					
MO dummy (MO)	0.078**	$0.126^{**}$	$0.085^{**}$	$0.088^{**}$	$0.093^{**}$	$-0.245^{**}$	$-0.531^{**}$	1				
Herfindahl-Hirschman Index (HHI)	0.033**	0.029**	$0.047^{**}$	$0.037^{**}$	-0.0007	-0.239**	0.007	-0.016**	1			
Number of workers (Size)	-0.083**	$0.358^{**}$	$0.461^{**}$	$0.446^{**}$	0.297**	-0.039**	-0.068**	$0.056^{**}$	0.025**	1		
Age	0.066**	$0.173^{**}$	$0.208^{**}$	0.205**	$0.149^{**}$	0.0001	-0.020**	$0.066^{**}$	$0.017^{**}$	$0.266^{**}$	1	
Share of non-production workers (NonP)	0.007	0.141**	0.111**	0.116**	$0.165^{**}$	0.016**	-0.070**	0.098**	-0.070**	0.097**	$0.136^{**}$	1

 Table A.7: Correlation matrix of variables in regression models

Note: \*\*: Significance level of 5%.

Regressor		EXP			MNE		EXP×MNE			
	$\Pr(PMS=IC \mathbf{x})^a$	$\Pr(\text{LMS}=\text{EB} \mathbf{x})^b$	$\Pr(LMS=MO \mathbf{x})^b$	$\Pr(PMS=IC \mathbf{x})^a$	$\Pr(\text{LMS}=\text{EB} \mathbf{x})^b$	$\Pr(LMS=MO \mathbf{x})^b$	$\Pr(PMS=IC \mathbf{x})^a$	$\Pr(\text{LMS}=\text{EB} \mathbf{x})^b$	$\Pr(LMS=MO \mathbf{x})^b$	
Industry	dF/dx	dF/dx	dF/dx	dF/dx	dF/dx	dF/dx	dF/dx	dF/dx	dF/dx	
1	0.009	0.165***	-0.160***	0.000	-0.018	-0.047	-0.014	-0.202**	0.052	
	(0.008)	(0.050)	(0.045)	(0.007)	(0.060)	(0.050)	(0.011)	(0.088)	(0.078)	
2	0.188***	0.020	-0.118***	-0.008	-0.168***	0.002	-0.179***	0.144	0.131**	
	(0.038)	(0.062)	(0.039)	(0.009)	(0.060)	(0.035)	(0.040)	(0.093)	(0.062)	
3	0.643***	0.056		-0.051	-0.043		-0.607***	0.041		
	(0.131)	(0.166)		(0.058)	(0.085)		(0.139)	(0.193)		
4	-0.029	-0.115		0.034	0.012		0.013	0.040		
	(0.099)	(0.093)		(0.118)	(0.092)		(0.215)	(0.187)		
5	0.057***	0.070		0.050**	0.217		-0.232**	-0.812**		
	(0.017)	(0.200)		(0.020)	(0.215)		(0.110)	(0.382)		
6	0.007	0.086***	-0.120***	-0.015	-0.021	0.048	-0.007	-0.044	0.057	
	(0.011)	(0.031)	(0.040)	(0.017)	(0.061)	(0.073)	(0.018)	(0.067)	(0.080)	
7	0.120***	0.202***		-0.054	-0.035		-0.056	-0.159**		
	(0.042)	(0.044)		(0.052)	(0.056)		(0.063)	(0.068)		
8	0.357	0.048	-0.022	-0.565**	-0.060	0.092	0.373	0.043	-0.212**	
	(0.291)	(0.047)	(0.055)	(0.278)	(0.075)	(0.066)	(0.490)	(0.085)	(0.101)	
9	0.032		0.083	0.025		0.079	-0.090		-0.072	
	(0.038)		(0.062)	(0.041)		(0.066)	(0.055)		(0.089)	
10	-0.023	0.264		-0.174***	-1.027***		0.011	0.296		
	(0.051)	(0.247)		(0.059)	(0.322)		(0.088)	(0.392)		
11	0.062	0.061	-0.070*	0.012	-0.002	-0.024	0.021	-0.049	0.077	
	(0.219)	(0.044)	(0.042)	(0.235)	(0.057)	(0.045)	(0.304)	(0.071)	(0.063)	
12	-0.006	-0.009	0.013	-0.134***	-0.010	0.040	0.190***	0.012	-0.045	
	(0.026)	(0.008)	(0.011)	(0.051)	(0.027)	(0.040)	(0.055)	(0.026)	(0.036)	
13	0.005	-0.049		-0.034	-0.421***		-0.012	0.132		
	(0.024)	(0.096)		(0.045)	(0.161)		(0.050)	(0.188)		
14	0.002*	0.062	-0.024	0.005**	0.053	-0.026	-0.003**	-0.030	-0.006	
	(0.001)	(0.039)	(0.020)	(0.002)	(0.034)	(0.020)	(0.002)	(0.050)	(0.029)	
15	0.031***	0.049	-0.008	0.033	0.025	-0.081*	-0.024	-0.008	$0.076^{*}$	
	(0.011)	(0.041)	(0.019)	(0.024)	(0.073)	(0.042)	(0.026)	(0.080)	(0.044)	

Table A.8: Probit models on the likelihood of being characterized by PMS=IC and LMS={EB,MO}, by industry - Baseline specification

Notes: Significance level of \*\*\*1%, \*\*5%, \*10%. Bootstrapped standard errors (50 replications). <sup>a</sup>: Marginal effects of univariate probit model. <sup>b</sup>: Marginal effects of bivariate probit model, except for industries  $J = \{3,4,5,7,9,10,13\}$ . In industry 9, too few firms are characterized by LMS=EB, hence, marginal effects of the univariate probit model  $Pr(LMS=MO|x) = \Phi(x\beta)$  are reported, while either none of the firms in industries  $J = \{3,4,5,7,9,10,13\}$  or too few firms in industries  $J = \{5,7,10\}$  are characterized by LMS=MO. Hence, for industries  $J = \{3,4,5,7,10,13\}$ , marginal effects of the univariate probit model  $Pr(LMS=EB|x) = \Phi(x\beta)$  are reported.

$\mathbf{R} = \mathbf{P}\mathbf{C}\text{-}\mathbf{P}\mathbf{R}$	$\mu$	$\psi$	ω			R = IC-PR	$\mu$	$\psi$	ω		
All firms	0.948	0.389	-0.256			All firms	1.256	0.048	0.003		
Exporters	0.902	0.388	0.036			Exporters	1.235	0.033	-0.120		
Non-exporters	0.965	0.390	-0.380			Non-exporters	1.262	0.053	0.039		
MNEs	0.914	0.345	0.205			MNEs	1.214	0.033	-0.141		
Non-MNEs	0.955	0.401	-0.385			Non-MNEs	1.266	0.051	0.033		
R = PC-EB	$\mu$	$\psi$	ω	$\gamma$	$\phi$	R = IC-EB	$\mu$	$\psi$	ω	$\gamma$	$\phi$
All firms	1.020	0.758	-0.047	1.243	0.555	All firms	1.518	0.713	-0.073	0.408	0.290
Exporters	0.989	0.738	0.077	1.207	0.543	Exporters	1.419	0.599	0.054	0.393	0.282
Non-exporters	1.030	0.767	-0.108	1.246	0.562	Non-exporters	1.554	0.769	-0.117	0.414	0.293
MNEs	0.985	0.730	0.183	1.149	0.546	MNEs	1.355	0.540	0.137	0.363	0.266
Non-MNEs	1.029	0.763	-0.111	1.276	0.556	Non-MNEs	1.560	0.769	-0.122	0.420	0.296
R = PC-MO	$\mu$	$\psi$	ω	β	$\varepsilon_w^L$	R = IC-MO	$\mu$	$\psi$	ω	$\beta$	$\varepsilon_w^L$
All firms	0.863	-1.713	0.159	0.332	0.542	All firms	1.175	-0.749	0.089	0.620	1.633
Exporters	0.858	-1.747	0.186	0.325	0.508	Exporters	1.214	-0.856	0.030	0.597	1.479
Non-exporters	0.869	-1.668	0.134	0.340	0.579	Non-exporters	1.156	-0.692	0.116	0.633	1.723
MNEs	0.834	-1.979	0.216	0.296	0.463	MNEs	1.201	-0.964	0.034	0.569	1.322
Non-MNEs	0.877	-1.595	0.129	0.354	0.593	Non-MNEs	1.167	-0.696	0.105	0.635	1.737
All regimes	$\mu$	$\psi$	ω								
All firms	1.230	0.248	-0.031								
Exporters	1.181	0.102	0.048								
Non-exporters	1.254	0.299	-0.064								
MNEs	1.166	0.091	0.104								
Non-MNEs	1.252	0.288	-0.067								

**Table A.9:** Three-dimensional firm heterogeneity: Markups  $(\mu)$ , labor market imperfections (workers' bargaining power  $\phi$  or the wage elasticity of a firm's labor supply curve  $\varepsilon_w^N$ ), and productivity  $(\omega)$ 

Note: Median values of the relevant parameter estimates are reported.

	Dependent variable		$\ln \widehat{\mu}_{it+1}$			$\ln\left(\frac{\hat{\phi}_{it+1}}{1-\hat{\phi}_{it+1}}\right)$		$\ln(\widehat{arepsilon}^N_w)_{it+1}$			
Industry	Name	EXP	MNE	EXP×MNE	EXP	MNE	EXP×MNE	EXP	MNE	EXP×MNE	
1	Food products and beverages	0.057**	-0.011	-0.030	-0.321	0.111	0.280	1.775***	0.335	-0.820***	
		(0.027)	(0.017)	(0.035)	(0.584)	(0.138)	(0.773)	(0.664)	(0.220)	(0.278)	
2	Textiles and wearing apparel	0.021	-0.090**	0.116	-0.269*	-0.284	0.127	-0.389	-0.072	$0.609^{*}$	
		(0.040)	(0.041)	(0.075)	(0.156)	(0.235)	(0.250)	(0.265)	(0.121)	(0.316)	
3	Wood, wooden products, and furniture	0.040	-0.055	-0.144	-0.160	0.018	0.348				
		(0.178)	(0.092)	(0.233)	(0.177)	(0.158)	(0.248)				
4	Pulp, paper and paper products	0.191	-0.191	-0.125	0.117	-0.290	0.411				
		(0.161)	(0.124)	(0.128)	(0.302)	(0.251)	(0.291)				
5	Publishing and printing	0.051	-0.044	-0.118	-0.330	0.449	0.062				
		(0.074)	(0.054)	(0.090)	(0.487)	(0.550)	(0.389)				
6	Chemicals	0.052**	0.015	-0.072*	1.249*	-0.961	-0.992**	$1.106^{***}$	-0.069	-0.466***	
		(0.021)	(0.036)	(0.037)	(0.748)	(0.768)	(0.475)	(0.226)	(0.138)	(0.179)	
7	Petroleum and coal products	0.136***	-0.016	-0.141**	0.344*	-0.143	-0.138				
		(0.049)	(0.039)	(0.056)	(0.188)	(0.138)	(0.180)				
8	Non-metallic mineral products	0.101	-0.081	0.034	0.132	-0.211	0.012	0.051	-0.272**	0.414	
		(0.062)	(0.080)	(0.083)	(0.123)	(0.233)	(0.248)	(0.098)	(0.128)	(0.254)	
9	Iron and steel	0.370**	0.168	-0.406				0.182**	$0.159^{***}$	-0.329***	
		(0.178)	(0.218)	(0.432)				(0.081)	(0.059)	(0.086)	
10	Non-ferrous metals	-0.120**	-0.373***	$0.245^{***}$	0.156	0.168	0.021				
		(0.051)	(0.113)	(0.086)	(0.201)	(0.715)	(0.358)				
11	Fabricated metal products	0.031	-0.035	0.004	0.368***	0.157	-0.317	0.130	0.063	0.013	
		(0.029)	(0.036)	(0.047)	(0.117)	(0.175)	(0.197)	(1.239)	(0.600)	(1.388)	
12	Machinery	-0.051**	-0.233	0.338	0.049	$0.541^{***}$	-0.272	0.125***	-0.130*	0.122	
		(0.023)	(0.186)	(0.215)	(0.186)	(0.177)	(0.265)	(0.041)	(0.069)	(0.084)	
13	Electrical machinery	0.008	-0.166**	0.005	-0.243*	-1.672***	0.572**				
		(0.040)	(0.077)	(0.089)	(0.127)	(0.582)	(0.270)				
14	Transport equipment	0.017	-0.016	0.006	0.393***	-0.562***	0.033	1.479	-0.393	-0.265	
		(0.018)	(0.021)	(0.025)	(0.135)	(0.177)	(0.147)	(1.883)	(0.325)	(0.742)	
15	Other manufacturing	0.039	-0.034	-0.023	0.062	0.031	-0.288	2.402*	20.719*	-19.940*	
		(0.028)	(0.035)	(0.041)	(0.168)	(0.200)	(0.233)	(1.255)	(10.641)	(10.231)	

Table A.10: Mean regression results (OLS), by industry - Baseline specification

Notes: Significance level of \*\*\*1%, \*\*5%, \*10%. Standard errors are heteroskedasticity consistent and clustered by enterprises. In industry 9, too few firms are characterized by LMS=EB, hence, no average effects are reported for the regression model of the workers' bargaining power ( $\hat{\phi}$ ), while either none of the firms in industries  $J = \{3,4,13\}$  or too few firms in industries  $J = \{5,7,10\}$  are characterized by LMS=MO. Hence, for industries  $J = \{3,4,5,7,10,13\}$ , no average effects are reported for the regression model of the firm's labor supply elasticity ( $\hat{\varepsilon}_w^N$ ).

Dep. var:			FYP			MNE					EXP×MNE						
$\ln \widehat{\mu}_{it+1}$			ĽAI														
Industry	q10	q25	q50	q75	q90	q10	q25	q50	q75	q90	q10	q25	q50	q75	q90		
1	0.045***	0.038***	0.064***	0.067***	0.062	0.007	0.010	0.008	-0.009	-0.051***	-0.026**	-0.032**	-0.048**	-0.077***	-0.013		
	(0.005)	(0.009)	(0.017)	(0.020)	(0.038)	(0.010)	(0.006)	(0.008)	(0.018)	(0.017)	(0.011)	(0.013)	(0.023)	(0.026)	(0.051)		
2	0.039***	0.060***	$0.031^{**}$	0.025	0.042	-0.023***	-0.037***	-0.073***	-0.122***	-0.137**	-0.034**	-0.003	$0.058^{***}$	0.181***	$0.218^{*}$		
	(0.012)	(0.016)	(0.015)	(0.035)	(0.095)	(0.006)	(0.009)	(0.017)	(0.026)	(0.058)	(0.015)	(0.022)	(0.020)	(0.054)	(0.129)		
3	-0.050*	-0.059	-0.106	0.199	$0.389^{**}$	0.038	0.027	-0.077	-0.101	-0.291**	0.001	0.008	0.040	-0.317*	-0.264		
	(0.027)	(0.042)	(0.096)	(0.180)	(0.157)	(0.027)	(0.038)	(0.053)	(0.097)	(0.148)	(0.040)	(0.046)	(0.090)	(0.179)	(0.223)		
4	0.054	0.064	0.132	$0.255^{*}$	0.232	-0.023	-0.058	-0.097	0.005	-0.309*	-0.055	0.004	-0.102	-0.435**	-0.065		
	(0.041)	(0.045)	(0.101)	(0.150)	(0.197)	(0.033)	(0.048)	(0.082)	(0.159)	(0.174)	(0.051)	(0.043)	(0.110)	(0.180)	(0.203)		
5	0.026**	$0.045^{**}$	0.003	$0.126^{**}$	0.136	0.000	-0.016	-0.045*	-0.067	-0.070	-0.062***	-0.088***	-0.076	-0.143*	-0.300**		
	(0.012)	(0.018)	(0.030)	(0.054)	(0.097)	(0.015)	(0.015)	(0.027)	(0.049)	(0.051)	(0.022)	(0.025)	(0.049)	(0.083)	(0.122)		
6	0.032***	0.045***	0.055***	$0.054^{***}$	0.073***	0.002	0.016	0.039***	0.016	0.033	-0.017	-0.045***	-0.091***	-0.233***	-0.135***		
	(0.007)	(0.006)	(0.008)	(0.016)	(0.023)	(0.009)	(0.011)	(0.012)	(0.013)	(0.025)	(0.011)	(0.012)	(0.014)	(0.040)	(0.031)		
7	0.023**	0.030**	0.103***	$0.258^{***}$	0.371***	-0.014	-0.012	-0.010	-0.019	-0.094**	-0.011	-0.024	-0.102***	-0.096***	-0.243***		
	(0.009)	(0.014)	(0.019)	(0.031)	(0.046)	(0.010)	(0.018)	(0.023)	(0.026)	(0.045)	(0.010)	(0.021)	(0.029)	(0.019)	(0.054)		
8	0.063***	0.075***	0.133***	0.125***	0.074	-0.054***	-0.018	-0.042	-0.132*	-0.167***	0.051*	0.013	0.003	0.046	0.029		
	(0.015)	(0.020)	(0.022)	(0.035)	(0.068)	(0.020)	(0.047)	(0.036)	(0.071)	(0.058)	(0.027)	(0.053)	(0.041)	(0.082)	(0.074)		
9	0.046	0.074	0.107	0.186	0.608*	0.078	0.003	0.053	-0.017	0.225	-0.142	0.053	0.246	0.285	0.011		
	(0.062)	(0.064)	(0.104)	(0.216)	(0.337)	(0.076)	(0.103)	(0.100)	(0.200)	(0.281)	(0.172)	(0.205)	(0.346)	(0.546)	(0.791)		
10	-0.002	0.007	-0.023	-0.170***	-0.319***	-0.103***	-0.128***	-0.298***	-0.570***	-0.720***	0.023	0.038	0.113**	$0.358^{***}$	0.690***		
	(0.016)	(0.019)	(0.024)	(0.038)	(0.053)	(0.032)	(0.045)	(0.073)	(0.127)	(0.178)	(0.027)	(0.036)	(0.049)	(0.078)	(0.169)		
11	0.006	0.027***	$0.031^{***}$	$0.058^{***}$	0.044	-0.021**	-0.019	-0.003	0.012	-0.060	0.011	-0.018	-0.053**	-0.058**	0.071		
	(0.008)	(0.006)	(0.011)	(0.019)	(0.035)	(0.008)	(0.014)	(0.020)	(0.018)	(0.039)	(0.012)	(0.016)	(0.022)	(0.029)	(0.048)		
12	-0.010**	-0.029***	-0.033**	-0.031	-0.100**	-0.026	-0.121*	-0.205***	-0.385***	-0.346	0.035	$0.163^{*}$	$0.295^{***}$	$0.502^{***}$	$0.517^{*}$		
	(0.005)	(0.009)	(0.014)	(0.029)	(0.044)	(0.039)	(0.071)	(0.076)	(0.146)	(0.274)	(0.053)	(0.091)	(0.096)	(0.178)	(0.305)		
13	-0.021	-0.012	-0.001	0.054	-0.027	-0.062***	-0.136***	-0.197***	-0.196	-0.197	-0.008	-0.008	0.086	-0.035	-0.072		
	(0.014)	(0.018)	(0.042)	(0.063)	(0.056)	(0.022)	(0.045)	(0.076)	(0.144)	(0.214)	(0.039)	(0.040)	(0.085)	(0.160)	(0.243)		
14	0.017***	0.008	0.022***	$0.041^{***}$	0.017	-0.003	-0.011	-0.006	-0.007	-0.018	-0.005	0.001	-0.023**	-0.019	0.034		
	(0.005)	(0.005)	(0.006)	(0.012)	(0.017)	(0.005)	(0.007)	(0.007)	(0.012)	(0.018)	(0.007)	(0.009)	(0.009)	(0.019)	(0.031)		
15	0.004	0.007	0.028**	$0.098^{***}$	$0.064^{**}$	-0.024	-0.013	-0.038*	-0.013	-0.055	$0.042^{*}$	0.019	-0.003	-0.110***	-0.085*		
	(0.008)	(0.009)	(0.013)	(0.023)	(0.028)	(0.023)	(0.012)	(0.022)	(0.035)	(0.046)	(0.023)	(0.013)	(0.024)	(0.041)	(0.044)		

 Table A.11: Quantile regression (QR) results, by industry - Baseline specification

Dep. var:															
$\ln\left(\frac{\widehat{\phi}_{it+1}}{1-\widehat{\phi}_{it+1}}\right)$			EXP					MNE					EXP×MN	Ε	
Industry	q10	q25	q50	q75	q90	q10	q25	q50	q75	q90	q10	q25	q50	q75	q90
1	-1.015**	-0.878***	-0.711*	0.108	0.241	0.210*	0.211***	0.113	0.089	0.056	1.123*	0.925**	0.863*	-0.243	-0.432
	(0.451)	(0.301)	(0.422)	(0.339)	(0.605)	(0.112)	(0.068)	(0.086)	(0.094)	(0.099)	(0.600)	(0.397)	(0.521)	(0.452)	(0.760)
2	0.030	-0.238***	-0.385***	-0.236**	0.125	-0.108	-0.161	-0.165	-0.781***	-0.259	0.152	$0.316^{**}$	0.136	0.365	-0.572
	(0.162)	(0.073)	(0.102)	(0.114)	(0.323)	(0.142)	(0.127)	(0.140)	(0.203)	(0.181)	(0.228)	(0.147)	(0.130)	(0.243)	(0.416)
3	-0.415**	-0.217*	-0.204	-0.003	-0.133	0.202*	0.057	0.079	0.052	-0.013	0.470	0.380	0.272	0.188	0.175
	(0.203)	(0.131)	(0.182)	(0.193)	(0.203)	(0.113)	(0.113)	(0.143)	(0.103)	(0.111)	(0.300)	(0.237)	(0.252)	(0.206)	(0.258)
4	0.494**	0.209	0.020	-0.130	-0.092	-0.265**	-0.303*	-0.123	-0.291**	-0.392	0.489**	$0.612^{***}$	0.319	$0.319^{*}$	0.510
	(0.207)	(0.208)	(0.211)	(0.169)	(0.310)	(0.131)	(0.161)	(0.223)	(0.135)	(0.288)	(0.192)	(0.177)	(0.222)	(0.184)	(0.401)
5	-0.086	-0.363***	-0.125	-0.270	-0.974**	0.167	$0.281^{*}$	0.078	0.419	$1.298^{**}$	-0.131	0.114	-0.166	0.404	$0.879^{**}$
	(0.246)	(0.128)	(0.371)	(0.412)	(0.468)	(0.229)	(0.151)	(0.398)	(0.486)	(0.526)	(0.227)	(0.112)	(0.253)	(0.289)	(0.406)
6	1.124	0.302	$1.440^{**}$	$1.146^{***}$	0.721	-0.814	0.125	-1.132	-1.040**	-0.542	-0.700	-0.411	-1.203***	-0.904***	-0.777**
	(0.803)	(0.969)	(0.699)	(0.385)	(0.525)	(0.839)	(1.051)	(0.730)	(0.417)	(0.551)	(0.581)	(0.570)	(0.458)	(0.242)	(0.340)
7	0.405***	0.223	0.210	$0.536^{***}$	0.802***	-0.152	-0.132	-0.012	-0.148**	-0.097	0.094	-0.033	-0.053	-0.285*	-0.655***
	(0.110)	(0.137)	(0.143)	(0.157)	(0.184)	(0.109)	(0.084)	(0.077)	(0.066)	(0.144)	(0.284)	(0.117)	(0.136)	(0.168)	(0.204)
8	0.223*	0.080	0.000	0.144	0.070	-0.061	-0.240	-0.253	-0.017	-0.416	-0.170	0.250	0.220	-0.331	-0.022
	(0.115)	(0.082)	(0.108)	(0.120)	(0.144)	(0.252)	(0.219)	(0.284)	(0.335)	(0.284)	(0.146)	(0.229)	(0.276)	(0.308)	(0.286)
10	0.336*	0.451**	0.237	0.252	-0.082	0.659	0.161	0.146	-0.520	-0.538	-0.198	-0.104	0.005	0.165	$0.788^{**}$
	(0.194)	(0.189)	(0.181)	(0.158)	(0.203)	(0.745)	(0.587)	(0.657)	(0.751)	(0.852)	(0.385)	(0.289)	(0.310)	(0.422)	(0.382)
11	0.334***	$0.312^{***}$	$0.431^{***}$	$0.381^{***}$	0.242***	0.271**	$0.207^{**}$	0.098	0.257	$0.372^{*}$	-0.284*	-0.229	-0.364**	-0.484**	-0.298
	(0.069)	(0.075)	(0.071)	(0.069)	(0.086)	(0.113)	(0.083)	(0.125)	(0.182)	(0.195)	(0.169)	(0.145)	(0.150)	(0.226)	(0.195)
12	-0.299	0.028	0.264	0.241	0.042	1.045**	$0.838^{*}$	0.222	0.220	0.188	-0.396	-0.454	0.142	-0.040	-0.007
	(0.513)	(0.376)	(0.237)	(0.303)	(0.327)	(0.468)	(0.488)	(0.513)	(0.530)	(0.487)	(0.816)	(0.705)	(0.588)	(0.700)	(0.681)
13	0.051	-0.153**	-0.211***	-0.379***	-0.333**	0.075	-1.182**	-1.466***	-1.901***	-0.770	-0.194	$0.375^{*}$	$0.527^{***}$	0.821***	0.453
	(0.091)	(0.070)	(0.080)	(0.112)	(0.157)	(0.661)	(0.488)	(0.400)	(0.582)	(0.878)	(0.315)	(0.205)	(0.178)	(0.261)	(0.398)
14	0.162**	0.096	0.310***	$0.462^{***}$	$0.541^{***}$	-0.296***	-0.246**	-0.333***	-0.562***	-0.738***	0.209**	0.269***	0.015	-0.067	-0.095
	(0.082)	(0.102)	(0.078)	(0.106)	(0.087)	(0.081)	(0.098)	(0.090)	(0.104)	(0.075)	(0.098)	(0.093)	(0.082)	(0.084)	(0.094)
15	0.152	0.121	0.064	0.065	0.169	0.189	0.125	0.009	-0.055	-0.121	-0.346**	-0.308*	-0.309***	-0.431***	-0.364
	(0.094)	(0.078)	(0.075)	(0.088)	(0.176)	(0.125)	(0.183)	(0.104)	(0.101)	(0.217)	(0.138)	(0.183)	(0.111)	(0.126)	(0.240)

Table A.11 - Continued: Quantile regression (QR) results, by industry - Baseline specification

Dep. var: $\ln(\widehat{\varepsilon}_w^N)_{it+1}$			EXP			MNE					EXP×MNE				
Industry	q10	q25	q50	q75	q90	q10	q25	q50	q75	q90	q10	q25	q50	q75	q90
1	1.333***	1.896***	1.377***	1.806***	2.488***	0.100	0.141	0.181	0.521***	0.816***	-0.564***	-0.769***	-0.852***	-0.886***	-1.175***
	(0.411)	(0.510)	(0.350)	(0.403)	(0.578)	(0.134)	(0.168)	(0.118)	(0.137)	(0.146)	(0.175)	(0.230)	(0.179)	(0.233)	(0.197)
2	-0.571*	-0.354	-0.177	-0.440	-0.315	-0.171	0.107	-0.035	-0.169	-0.245*	1.090**	0.553	0.276	$0.678^{*}$	$0.687^{*}$
	(0.329)	(0.286)	(0.284)	(0.364)	(0.331)	(0.202)	(0.190)	(0.124)	(0.124)	(0.136)	(0.427)	(0.425)	(0.373)	(0.408)	(0.415)
6	0.371**	0.907***	1.335***	$1.378^{***}$	1.352***	-0.026	-0.156**	0.123	-0.147	-0.050	-0.011	-0.274***	-0.750***	-0.620***	-0.618***
	(0.155)	(0.129)	(0.124)	(0.274)	(0.289)	(0.093)	(0.073)	(0.103)	(0.095)	(0.133)	(0.131)	(0.090)	(0.131)	(0.150)	(0.201)
8	0.188*	$0.201^{**}$	0.089	-0.050	-0.082	-0.092	-0.213*	-0.350***	-0.368***	-0.326**	0.051	0.299	$0.381^{*}$	0.487**	$0.631^{**}$
	(0.096)	(0.082)	(0.064)	(0.056)	(0.079)	(0.165)	(0.123)	(0.111)	(0.119)	(0.147)	(0.372)	(0.300)	(0.227)	(0.245)	(0.259)
9	0.085*	$0.095^{*}$	$0.235^{***}$	0.260***	$0.218^{***}$	0.137**	$0.143^{***}$	0.233***	$0.176^{***}$	$0.097^{**}$	-0.171**	-0.222***	-0.439***	-0.448***	-0.377***
	(0.049)	(0.050)	(0.063)	(0.044)	(0.065)	(0.053)	(0.051)	(0.051)	(0.042)	(0.045)	(0.070)	(0.059)	(0.069)	(0.068)	(0.086)
11	0.609	0.068	$1.617^{*}$	0.731	0.484	0.448	0.026	$0.831^{*}$	0.310	0.135	-0.599	0.132	-1.724*	-0.640	-0.380
	(0.824)	(0.839)	(0.900)	(1.012)	(0.998)	(0.390)	(0.388)	(0.438)	(0.473)	(0.482)	(0.914)	(0.918)	(1.015)	(1.121)	(1.107)
12	0.147***	$0.157^{***}$	0.147***	$0.094^{***}$	0.056	-0.127**	-0.076	-0.095**	-0.121***	-0.142	0.123**	0.065	0.056	$0.111^{**}$	0.140
	(0.023)	(0.022)	(0.019)	(0.033)	(0.049)	(0.053)	(0.073)	(0.043)	(0.047)	(0.089)	(0.057)	(0.081)	(0.051)	(0.048)	(0.095)
14	-3.566	4.815	0.620	2.062	$5.374^{**}$	0.206	-0.972	-0.323	-0.351	-0.682*	1.746	-1.503	0.230	-0.645	-2.115**
	(5.094)	(4.200)	(2.345)	(2.467)	(2.529)	(0.820)	(0.647)	(0.410)	(0.350)	(0.377)	(1.824)	(1.632)	(0.951)	(1.057)	(1.043)
15	1.814	2.948	2.060	1.647	1.859	14.368	24.913	17.356	13.341	14.239	-14.200	-24.145	-16.685	-12.831	-13.634
_	(2.461)	(2.266)	(1.898)	(1.861)	(2.314)	(21.031)	(19.355)	(16.314)	(16.147)	(20.022)	(20.241)	(18.635)	(15.712)	(15.597)	(19.313)

Table A.11 - Continued: Quantile regression (QR) results, by industry - Baseline specification

Notes: Significance level of \*\*\*1%, \*\*5%, \*10%. Bootstrapped standard errors (50 replications).

Results are based on pooled simultaneous-quantile regressions for  $\tau \in \{0.05, 0.10, 0.20, 0.25, 0.30, 0.40, 0.50, 0.60, 0.70, 0.75, 0.80, 0.90, 0.95\}$ .

Results for other regressors and other quantiles are available upon request. See Table A.10 for the selection of industries.

**Graph A.1:** Average and quantile impact of export/MNE status on  $\ln \hat{\mu}_{it+1}$  in selected industries



**Graph A.1 - Continued:** Average and quantile impact of export/MNE status on  $\ln \hat{\mu}_{it+1}$  in selected industries



Notes: In industries manufacturing food products and beverages, textiles and wearing apparel, publishing and printing, chemicals, petroleum and coal products, non-metallic mineral products, fabricated metal products, transport equipment, and other manufacturing (industries  $J = \{1, 2, 5, 6, 7, 8, 11, 14, 15\}$ ), we observe positive exporter returns along most quantiles, whereas we observe negative exporter returns in machinery (industry 12). Negative MNE returns are found for firms in industries manufacturing textiles and wearing apparel, non-metallic mineral products, non-ferrous metals, machinery, and electrical machinery (industries  $J = \{2, 8, 10, 12, 13\}$ ).

**Graph A.2:** Average and quantile impact of export/MNE status on  $\ln\left(\frac{\hat{\phi}_{it+1}}{1-\hat{\phi}_{it+1}}\right)$  in selected industries



**Graph A.2 - Continued:** Average and quantile impact of export/MNE status on  $\ln\left(\frac{\hat{\phi}_{it+1}}{1-\hat{\phi}_{it+1}}\right)$  in selected industries



Notes: In industries manufacturing petroleum and coal products, non-ferrous metals, and transport equipment (industries  $J = \{7, 11, 14\}$ ), we observe positive exporter returns whereas negative exporter returns are found in industries manufacturing food products and beverages, textiles and wearing apparel, and electrical machinery (industries  $J = \{1, 2, 13\}$ ). Positive MNE returns are found in fabricated metal products (industry 11) whereas firms in industries manufacturing pulp, paper and paper products, electrical machinery, and transport equipment (industries  $J = \{4, 13, 14\}$ ) experience negative MNE returns.



**Graph A.3:** Average and quantile impact of export/MNE status on  $\ln(\hat{\varepsilon}_w^N)_{it+1}$  in selected industries

Notes: Positive returns along the full conditional distribution are found for firms in industries manufacturing food products and beverages, chemicals, iron and steel, and machinery (industries  $J = \{1, 6, 9, 12\}$ ). Positive MNE returns are observed for firms in iron and steel (industry 9) whereas negative MNE returns are found for firms in industries manufacturing non-metallic mineral products and machinery (industries  $J = \{8, 12\}$ ).

.4 .5 .6 .7 .8 .9 Quantile

-0.60

.1 .2 .3

.8 .9

.7

.4 .5 .6 Quantile

0.0

-0.30

.1 .2 .3 .4 .5 .6 Quantile .7 .8 .9